THE MERITS OF THE IRANIAN PETROLEUM CONTRACT MODEL TO MEET THE DEVELOPMENT NEEDS OF IRAN'S OIL RESOURCES SECTOR

Pegah Pornouri

Principal supervisor: Dr Catherine Renshaw Co-supervisor: John Juriansz

A thesis submitted in fulfilment of the requirements for the degree of Doctor of Philosophy Western Sydney University School of Law Sydney, Australia 2020

ABSTRACT

This research provides an examination of the Iranian Petroleum Contract (IPC) system from inception to the present day. First, the thesis analyses the expansion of petroleum contracts from the initial Concession Agreements to the termination of all obligations following the Islamic Revolution of 1979 and the Nationalisation Movement. Second, the thesis examines the ongoing policy tension between the Iranian oil industry's need to access foreign funding and technology and its aim to avert foreign exploitation of its natural resources. Focus is given to the impact of this tension on Iran's current position regarding foreign investment and the development of the current contractual model. In this context, the thesis compares the Buy Back Contract as a model for access to Iranian petroleum in the period since 1989 with the IPC model, noting that the latter is fundamentally characterised by the provision of funds by foreign investors for the development of oil fields. The thesis argues that the IPC model, where rights are granted to successful foreign investor companies to access their capital, technology, and know-how in exchange for remuneration and the potential reward for successful production, is essentially a compromise approach that addresses the twin national priorities of (1) accessing international funds, expertise and services while (2) constraining the exploitation of natural resources by foreign parties. The thesis provides an extensive analysis of the IPC structure and relative advantages via a comparative analysis with the Buy Back Contract model. The thesis also analyses other major contractual models in the international arena and assesses their suitability for the oil industry in Iran.

The thesis analyses IPCs from the perspectives of both Iran and the foreign oil companies. As such, it provides a platform from which to determine whether all participants may be better served by revising the terms of IPCs or by substituting IPCs altogether with an alternative model. A relevant factor in this regard is the degree to which the petroleum industry in Iran was and may continue to be impacted by pressure from international actors, particularly the United States (US). The thesis considers the extent to which Iran resists international pressure and positions itself towards other countries who are willing to continue to sign Buy Back contracts. The thesis entails a close study of IPC's terms and conditions through a review of several petroleum fields in order to ascertain whether perceived issues with the model emerge in reality.

The thesis concludes that notwithstanding the demonstrable flaws in the IPC model, its revised terms and conditions regarding key elements related to contract length, risks borne by investor companies, and the mechanism for remuneration approach make it the best contract structure to meet the development needs of Iran's oil and gas sectors. Based on this conclusion, the thesis recommends that in the current economic, political and legal climate, the IPC model provides a framework for Iran to initiate reforms to its resources sector while at the same time preserving Iran as an attractive option to foreign investors.

STATEMENT OF AUTHENTICATION

The work presented in this thesis is, to the best of my knowledge and belief, original except as acknowledged in the text. I hereby declare that I have not submitted this material, either in full or in part, for a degree at this or any other institution.

Signature: Pegah Pornou Date: 08/09/2021



ACKNOWLEDGEMENTS

I would like to express my deep thanks and appreciation to my principal supervisor Doctor Catherine Renshaw for her great teaching, encouragement and invaluable ideas. She was always providing great help and support with patience throughout the period of my candidature.

My sincere thanks go to my co-supervisor, John Juriansz for his constructive criticism, invaluable comments and great support. I am also thankful to the Western Sydney University School of Law and all its staff for all the considerate guidance they provided.

Lastly, I am profoundly grateful to my family for their great support and to all those people who assisted me with valued and insightful comments. It is not possible to list all their names, but their cooperation and support are sincerely acknowledged. CONTENTS

ABSTRACT	Ι
STATEMENT OF AUTHENTICATION	III
ACKNOWLEDGEMENTS	IV
LIST OF TABLES	VIII
LIST OF FIGURES	VIII
LIST OF SELECTED ACRONYMS	Х
DEFINITIONS OF KEY TERMS	XI
1 INTRODUCTION	1
1.1 Snap-Shot of the Iranian Petroleum Industry	2
1.2 Foreign Investment in Iran's Petroleum Industry	6
1.2.1 Buy Back Contracts	6
1.2.2 Iranian Petroleum Contracts	8
1.3 Scope of the Thesis	9
1.4 Research Questions	10
1.5 Methodology	12
1.6 Organisation of this Thesis	15
2 IRAN'S PETROLEUM INDUSTRY: INFLUENCE OF PRE-NATIONALISATION (1908-1951)	
CONCESSIONS, NATIONALISATION AND SANCTIONS ON MODERN PRETROLEUM CONTRACTS	18
2.1 Introduction	18
2.2 The Global Upstream Oil and Gas Sector	18
2.2.1 Iran's place within the global upstream oil and gas sector	20
2.3 Pre-1900 Oil Concessions	22
2.3.1 The de Reuter Concession	24
2.3.2 Hotz Concession	25
2.3.3 Imperial Bank of Persia Concession	25
2.4 Twentieth Century Pre-Nationalisation Oil Concessions	26
2.4.1 D'Arcy Concession and foreign involvement	27
2.4.2 D'Arcy Concession dispute and the Armitage-Smith Agreement	29
2.4.3 D'Arcy Concession: Cancellation and the new agreement	30
2.4.4 1933 Concession	31
2.5 World War Two and Politics in Iran	32
2.5.1 Soviet Union influence on oil concessions	33
2.6 Overview of Pre-Nationalisation Era	34
2.6.1 Concession issues and their renegotiation	35
2.6.2 Rejecting the compromise agreement	35
2.7 Nationalisation of the Oil Industry and its Consequences	36

2.7.	1 The 1953 Coup D' Etat and its implications for foreign investment	37
2.7.	2 The 1954 Consortium	38
2.7.	3 The National Iranian Oil Company and the International Consortium	39
2.7.	4 Reforming the petroleum agreements	40
2.8	Contractual and Commercial Changes between 1960 and 1979	41
2.8.	1 Petroleum transactions and the Iranian economic development plans	41
2.8.	2 The 1965 agreements	42
2.9	The Islamic Revolution and its Impact on Petroleum Contracts	42
2.10	Iran-Iraq wars	43
2.11	Conclusion	44
3	Regulatory Frameworks	44
3.1	Introduction	44
3.1.	1 Single article non-voiding the contracts	44
3.1.	2 Bill to establish the Ministry of Petroleum, 1941	46
3.1.	3 Petroleum Law 1947	48
3.2	Additional Regulations following the Islamic Revolution	49
3.2.	1 Annual Budget Law	50
3.2.	2 Annual budget laws and the five-year development plan	51
3.2.	3 Legal support for foreign investment law	52
3.3	Conclusion	52
4	MODERN CONTRACTUAL FORMS: THE EMERGENCE OF BUY BACK TRANSACTIONS	55
4.1	Introduction	55
4.2	International Petroleum Agreements	55
4.3	Addressing the Imbalance	58
4.4	Contractual Systems	61
4.4.	1 Production Sharing Agreements	62
4.4.	2 Revenue sharing contracts	66
4.4.	3 Joint Venture Agreements	66
4.4.	4 Service Contracts	67
4.5	Buy Back Contracts	69
4.5.	1 International Buy Back Contracts	71
4.5.	2 Limitations of Buy Back Contracts	72
4.6	Conclusion	73
5 CONT	IRANIAN PETROLEUM CONTRACT – AS A RESPONSE TO THE PROBLEMS WITH BUY BA	.cks 75
5.1	Energy Sector in Iran	75
5.1.	1 Competition in Iran's oil fields	77

5.2	А	ttracting Foreign Direct Investment into Iran's Energy Sector	79			
5.2.	5.2.1 Foreign direct investment and the Iranian Petroleum Contract					
5.3	5.3 Iranian Petroleum Contract and Iran's Constitution					
5.3.	5.3.1 Redesigning the contracts					
5.3.	5.3.2 Iranian Petroleum Contract fiscal parameters					
5.3.	3	Investment, technology and exploration benefits	92			
5.3.	5.3.4 Foreign interest in Iranian Petroleum Contracts					
5.4	U	nited States Sanctions	99			
5.5	С	omparing Buy Back Contracts to Iranian Petroleum Contracts	102			
5.5.	1	Pros and cons of the Buy Back Contract	107			
5.6	S	ummary of Iranian Petroleum Contracts	113			
5.7	С	ritical Analysis of the Iranian Petroleum Contract	115			
5.7.	1	Perceived weaknesses and ambiguities	115			
5.7.	2	Iranian law	117			
5.7.	3	Joint Venture Agreements and credit risk for international oil companies	118			
5.8	P	ros and cons of the Iranian Petroleum Contract	119			
5.8.	1	Unconstitutional?	120			
5.8.	2	Undermines Iran's sovereignty over its natural resources	121			
5.8.	3	Lack of clarity around role of domestic entity	121			
5.8.	4	More risks for Iran related to international oil company remuneration payments	122			
5.8.	5	Recovery of costs	122			
5.8.	6	Take advantage of foreign technologies and expertise	123			
5.8.	7	Increase optimisation of natural resources	123			
5.8.	8	Favourable dispute resolution mechanism	124			
5.8.	9	International oil company concerns with the Iranian Petroleum Contract	125			
5.9	Ir	nplications of the analysis of the Iranian Petroleum Contracts pros and cons	127			
5.9.	1	Political landscape	129			
5.9.	2	Cost recovery	130			
5.9.	3	Fee structure	131			
5.9.	4	Joint Venture Agreements and management structures	133			
5.9.	5	Time extensions	134			
5.9.	6	Timing	137			
5.9.	7	Lack of access to technology	138			
5.10	5.10 Recommendations 140					
6	6 CONCLUSION 144					
6.1 Petro	S leu	trengths and Weaknesses of the Iran Petroleum Contract in regard to the Internati m Trade (RQ1)	onal 145			

6.1. 81 c	1 The long duration of the IPC and the impact on its efficacy in the context of artion of the Iranian Constitution	cle 148
6.1. stab	2 Establishing stable and predictable contractual relationships: The role of ilisation clauses	150
6.1.	3 Meeting legal, economic, and social objectives	153
6.2 Petrol	Impact of the Buy Back Contract Model on Iran's Participation in the Internationaleum Trade	1 154
6.2.	1 Buy Back Contract model's limited fiscal system	155
6.2.	2 Technology and knowledge transfer	156
6.3 Needs	The Extent to which the Iran Petroleum Contract Addresses the Current and Poten s of the Oil Resources Industry in Iran (RQ3)	itial 157
7	BIBLIOGRAPHY	160
7.1	Books	160
7.2	Journal Articles	163
7.3	Iran	169
7.4	Government Publications (outside of Iran)	170
7.5	Industry Publications	170
7.6	Online News Publications / Newspapers	173
7.7	Working Papers	175
7.8	Conferences	175
7.9	Interviews	176
7.10	Dictionaries	176

LIST OF TABLES

Table 4.1 Non-Concessionary Contracts	57
Table 5.1 IOCs with certification to bid at Iran's oil and natural gas tenders, 2017	97
Table 5.2 Comparison of the Buy Back Contract and IPC models	105
Table 5.3 Pros and cons of Buy Back Contracts for the oil and gas sectors in Iran	107
Table 5.4 IPC Highlights	114
Table 5.5 Pros and cons of the IPC for the oil and gas sectors in Iran	119

LIST OF FIGURES

Figure 1.1 OPEC share of world crude oil reserves 2018	3
Figure 1.2 Monthly Iran crude oil and lease condensate exports in 2018	5
Figure 1.3 Crude and condensate exports from Iran to Asia, Turkey and Europe, November	
2017 to April 2018	6

Figure 5.1 Basic costs structure of IPC

LIST OF SELECTED ACRONYMS

AIOC	Anglo-Iranian Oil Company
APOC	Anglo-Persian Oil Company
BOT	Build, Operate and Transfer
BP	British Petroleum
BPD	Barrels Per Day
BBC	Buy Back Contract
CAPEX	Capital Expenditures
СоМ	Cost of Money
CSI	Cost Saving Index
DCC	Direct Capital Costs
EPC	Engineering, Procurement and Construction contract
EPD	Engineering, Procurement and Drilling contract
EEA	European Economic Area
FIPPA	Foreign Investment Promotion and Protection Act
FDI	Foreign Direct Investment
ICJ	International Court of Justice
IDC	Indirect Costs
IEA	International Energy Agency
IOC	International Oil Company
IPAC	Iran Pan American Corporation
IPC	Iranian Petroleum Contract
JCPOA	Joint Comprehensive Plan of Action
JOA	Joint Operation Agreement
JVA	Joint Venture Agreement
LPG	Liquefied Petroleum Gas
LIBOR	London Interbank Offered Rate
NIOC	National Iran Oil Company
OPEC	Organization of Petroleum Exporting Countries
PSA	Production Sharing Agreement
R&D	Research and Development
ROR	Rate Of Return
UK	United Kingdom
US	United States

DEFINITIONS OF KEY TERMS

Agreement: Agreed understanding by two or more parties about their respective rights and duties concerning past or future performances; a demonstration of mutual agreement by two or more parties.¹

Buy Back Contract: A financing deal whereby the host nation sells a property to a foreign investor on the condition it buys back the production output via long-term sales contract.² *Concession Contract*: Grants rights to juridical persons or International Oil Company to explore and exploit oil and gas in exchange for the payment of costs and taxes related to the operations.³

Contract: An agreement among two or more parties producing obligations and responsibilities that are enforceable by law.⁴

Exploration license: An investor is granted permission to survey and drill shallow wells to access petroleum.

Joint Venture Agreement: A contract form involving a business undertaking by two or more parties to deliver project outcomes.⁵

License: An investor is granted the right to exploit a specified area for a fee and/or royalty payment.⁶

New Concession Contract: Includes the basic legal format of the Concession Contract, but with modifications to its financial terms and features.⁷

Production License: An investor is granted the exclusive right to search for and bore petroleum.

Production Sharing Agreement: A contract between an international company and a host government whereby the company provides investment capital in exchange for control over an oil and gas field and access to revenues from sales.⁸

Reservoir: A subsurface, porous, permeable rock formation containing oil and gas.9

¹ Black's Law Dictionary (4th ed, 1951) 89.

² Willem JH van Groenendaal and Mohammad Mazraati, 'A Critical Review of Iran's Buyback Contracts' (2006) 34(18) *Energy Policy* 3709, 3709.

³ Keith Blinn et al, *Exploration & Exploitation Agreements, Legal, Economic, and Policy Aspects* (Euromoney, 1986) 47.

⁴ Macquarie Dictionary (3rd ed, 1997) 241.

⁵ Black's Law Dictionary (n 1) 973.

⁶ Mohammad Alramahi, Oil and Gas Law in the UK (Bloomsbury, 2012) 5.

⁷ Albert T Chandler, 'Thailand Petroleum Concessions' (Research Discussion Paper Chandler & Thong-ek Law Offices, 14 December 2015) 2.

⁸ Ibid 3.

⁹ Anthony Jennings, Oil and Gas Exploration Contracts (Sweet & Maxwell, 2002) 56.

Royalty payments: Payments made by the licensee in cash or in kind as a percentage of the volume of the petroleum produced.¹⁰

Service Contract: An agreement by the host nation to grant an international company a contractual (not proprietary) right to conduct exploration and production of natural resources while retaining control over the operations.¹¹

Upstream oil and gas industry: Points in production that commence early in the process including exploration, drilling, and extraction.¹²

¹⁰ Daniel Johnston, *International Petroleum Fiscal Systems and Production Sharing Contracts* (Pennwell, 1994)
3.

¹¹ Alramahi (n 6) 6.

¹² Paolo Mecini and Ezio Mesini, 'The Petroleum Upstream Industry: Hydrocarbons Exploration and Production' (*Encyclopaedia of Life Support System*, 2011) 2.

1 INTRODUCTION

The Iran Petroleum Contract (IPC) is currently used as the contractual model for petroleum transactions in Iran. This thesis seeks to shine a light on the suitability of the IPC scheme for meeting Iran's resources management strategy as well as proposing solutions on how the nation might improve the contract scheme to strengthen relations with foreign investors. Recent political and legal developments in petroleum producing countries within the Middle East particularly has seen continued progress towards reinforcing the laws of the host country, in alignment with lex petrolea, 13 in the terms and conditions of the new petroleum contracts.¹⁴ In Iran specifically, the Oil Regulation 1974, the Oil Regulation 1987, the Oil Law Reform Act 2011, and relevant regulations in the Fourth and Fifth Development Plans, were each confirmed as compliant with Iranian laws as an obligatory order.¹⁵ As a starting point, it is important to clarify the definitions of the key terms applied in this thesis referring to natural resources and petroleum in particular. Although the standard Iranian contract and the relevant regulations refers to 'petroleum' rather than 'oil', the focus of this thesis is on concessions applied to the crude oil resources of Iran. Petroleum, meaning 'rock oil', is defined as geologically extracted hydrocarbons embedded mostly in sedimentary basins, oil shale and tar sands, which may be liquid (oil) or gaseous (natural gas) in form.¹⁶ The focus of this thesis is limited to oil contracts. It does not separately consider gas or refined condensates. It is also important to briefly note that although there is little difference in lexical terms between a contract and an agreement, an agreement may or may not be formalised as a contract. Lastly, although Iranian legislation is based on the Islamic (lunar) calendar, for the purpose of this thesis, all legislation is dated according to the Gregorian calendar.

The rationale for the focus of this thesis emerged from the ongoing tension in Iran between the petroleum industry's need to access foreign funding and technology and its aim to avert exploitation of its natural resources by foreign entities. This tension continues to have an impact on Iran's current position regarding foreign investment and the development of the

¹³ Lex Petrolea was introduced into the legal lexicon of the international oil and gas industry in an international arbitration case in 1982. It refers to the customary law comprising the legal rules for the nature and specificities of international petroleum industry disputes. Thomas C Childs, 'Update on Lex Petrolea: The Continuing Development of Customary Law Relating to International Oil and Gas Exploration and Production' (2011) 4(3) *Journal of World Energy Law and Business* 214, 214.

¹⁴ Ahmad Heidari, 'The Governing Law of International Oil Contracts in Iran Legislations' (2015) 36(4) Fen Bilimleri Dergisi (CFD) 1756, 1756.

¹⁵ Ibid.

¹⁶ John E McMurry, Organic Chemistry with Biological Applications (Cengage Learning, 2014) 3.

current contractual model. Indeed, this tension was originally manifest in the Buy Back Contact model used in transactions with foreign oil companies for access to Iranian petroleum since 1989 (this is discussed in detail in Chapter 4). The Buy Back Contract model was fundamentally a short-term risk service contract characterised by the provision of funds by foreign investors for oil exploration and exploitation of petroleum in exchange for a set remuneration return for successful production. As such, it presented essentially as a compromise approach to both access international funds, expertise and services while constraining the exploitation of natural resources by foreign parties. A key factor in this approach is that when the oil field goes into full production, its operation is taken over by the National Iranian Oil Company (NIOC). It is this legal element which differentiates the Buy Back Contact model and the IPC. Other contract models exist in the international arena which are briefly referred to in sections throughout Chapter 4.

A comprehensive analysis of the terms and conditions of the Iranian IPC helps to ascertain whether the perceived issues with the contract model can be the subject of reform to improve outcomes for all stakeholders. As such, this study adopts a qualitative interpretive methodology to examine the current implementation of IPCs in Iran to attract international oil companies (IOCs) to invest in the exploitation of the nation's petroleum reserves. Specifically, industry and research literature such as journal articles, industry publications, government reports, media reports, and the like are analysed to draw conclusions about the merits of the different contract forms in relation to their stated legal, economic and social objectives (see Section 1.5 for further details).

1.1 Snap-Shot of the Iranian Petroleum Industry

The petroleum industry in Iran is more than 100 years old.¹⁷ Indeed, Iran was the first Middle Eastern nation to discover oil, grant a foreign country an oil concession,¹⁸ and to nationalise its oil sector. Iran is a leader in the global petroleum industry, accounting for the second-largest petroleum output within the Organisation of the Petroleum Exporting Countries (OPEC).¹⁹ Figure 1.1 compares the share of world crude oil reserves between

¹⁸ The first Concession was granted to William D'Arcy in 1901. The Concession gave D'Arcy access rights to designated areas over a stated time period to explore for hydrocarbons (which included crude oil and gas) and, if discovered, for their transfer from the host nation (Iran) to the enterprise. The exploration, production, sale rights etc. were granted in exchange for tax and royalty payments on profits.

¹⁷ The British discovered oil in Iran in 1908, with Iran celebrating 100 years of oil production in 2008/09.

¹⁹ OPEC was established at the Baghdad Conference on September 10–14, 1960. The founding states included Iran, Iraq, Kuwait, Saudi Arabia and Venezuela. Eight other countries have subsequently gained OPEC membership. The goal of OPEC is to formulate and implement petroleum policies to deliver stability in oil prices for producers and to establish an efficient and reliable supply of petroleum to the market.

OPEC and non-OPEC nations in 2018. It shows that Iran's petroleum production is the fourth-largest globally, with 155.6 billion barrels produced at the end of 2018, equating to 13.1 percent of the global oil reserves.



OPEC proven crude oil reserves, at end 2018 (billion barrels, OPEC share)

Venezuela	302.81 25.5%	Kuwait	101.50 8.5%	Algeria	12.20 1.0%	Gabon	2.00 0.2%
Saudi Arabia	267.03 22.4%	UAE	97.80 8.2%	Ecuador	8.27 0.7%	Equatorial Guinea	1.10 0.1%
IR Iran	155.60 13.1%	Libya	48.36 4.1%	Angola	8.16 0.7%		
Iraq	145.02 12.2%	Nigeria	36.97 3.1%	Congo	2.98 0.3%		

Figure 1.1 OPEC share of world crude oil reserves 2018²⁰

The petroleum industry in Iran is thus crucial to the national economy and thus has a substantial impact on social development and domestic politics. An estimated 80-90 percent of Iran's annual income comes from petroleum revenues, making it significant to the spending capacity of the government.²¹

Arguably the most significant domestic event to impact the petroleum industry in Iran was Islamic Revolution in 1979. This changed the national petroleum industry irrevocably as it led to the annulment of all national rules and regulations in the industry. The 8-year war with Iraq (1980-1988) shortly after exacerbated this upheaval due to the heavy damage to petroleum industry infrastructures and the considerable time and investment required, particularly from foreign companies, to finance the repairs. As will be seen in Chapter 2 at Section 2.9, these events affected the contract types and contract terms agreed to by Iran with

²⁰ OPEC 'Annual Statistical Bulletin 2019', <<u>https://www.opec.org/opec_web/en/data_graphs/330.htm</u>> (accessed 9 November 2019).

²¹ Energy Information Administration, 'Iran: Country Analysis Brief August 2006' <www.eai.doe.gov> (accessed 4 July 2019).

foreign investor companies. Lack of technical staff needed for petroleum production in Iran was especially disruptive to its petroleum industry as it increased the difficulty in achieving production targets.²² Moreover, lack of government funds to finance the exploration and production of oil and gas added to the difficulties.²³ These domestic issues prompted Iran to open up the petroleum industry to foreign investors through the Buy Back Contract scheme²⁴ with the first contract issued on 6 March 1995. As elaborated upon throughout this thesis, more recent disruptions to the petroleum industry in Iran have included the 2018 trade sanctions imposed on Iran by the US government (specifically, the Trump Administration). The sanctions include a ban on petroleum sales to the West which has led to a significant reduction in the number of direct crude oil buyers in the Iranian petroleum trade.²⁵ As with the sanctions of the 1980s and 1990s, the recent tough sanctions have been imposed against Iran with the primary objective to stop or to reduce the development of the country's nuclear program.²⁶ Some exemptions were initially granted by the Trump Administration to eight countries who were among Iran's biggest petroleum consumers; namely, China, India, Turkey, South Korea, Japan, Taiwan, Greece and Italy.²⁷ Figure 1.2 sets out Iran's petroleum exports to these countries during 2018.

²⁷ Reuters, US is Reportedly Granting 8 Countries Iran Sanctions Waivers

²² Edward Chow, Cyrus Ashayeri and Andrew J Stanley, 'The Future of Iran's Oil and Gas Industry' (USAEE Working Paper No. 18-364, 2018) 5.

²³ van Groenendaal and Mazraati (n 2) 3709.

²⁴ Maryam Shafiei Khah and Ali Amiri, 'Petroleum Contracts in Iran' (2014) 3(3) *European Online Journal of Natural and Social Sciences* 375, 376.

²⁵ Benoit Faucon, 'Amid Tensions, Iran's Crude Buyers Jump Ship' (27 May, 2019) *The Wall Street Journal* <<u>https://www.wsj.com/articles/amid-tensions-irans-crude-buyers-jump-ship-11558964215</u>> (accessed 21 June 2019).

²⁶ Farzam Ardalan, Nejad Ali Almasi, Mansour Atasheneh, 'Effects of Contractor and Employer's Obligations in Buy Back Contracts: Case Study of Oil Exporting Country' (2017) 5(2) *Entrepreneurship and Sustainability Issues, Entrepreneurship and Sustainability Center* 345, 346.

<<u>https://www.cnbc.com/2018/11/02/us-is-reportedly-granting-8-countries-iran-sanctions-waivers.html</u>> (accessed 19 June 2019).



Figure 1.2 Monthly Iran crude oil and lease condensate exports in 2018²⁸

However, these exemptions were ended by the US in April 2019 including in relation to China (Iran's biggest customer), leaving the United Arab Emirates (UAE), Egypt and potentially Russia²⁹ as the main 'customers' in the Iranian petroleum trade. Figure 1.3 presents further details of the crude and condensate³⁰ exports from Iran to Asia, Turkey and Europe.

²⁸ United States Energy Information Administration (2018)

<<u>https://www_flickr.com/photos/eiagov/43699512980</u>> (accessed 21 June 2019).

²⁹ Ibid.

³⁰ A gas condensate is a hydrocarbon liquid stream separated from natural gas comprising higher-molecularweight hydrocarbons which are found in the reservoir as elements of natural gas. They are extracted as liquids in separators or processing plants, in James G Speight, *Handbook of Offshore Oil and Gas Operations* (Elsevier, 2014) 19.



Figure 1.3 Crude and condensate exports from Iran to Asia, Turkey and Europe, November 2017 to April 2018³¹

1.2 Foreign Investment in Iran's Petroleum Industry

The history of foreign investment in Iran's petroleum industry is complex. The development of petroleum contracts in Iran along with the expansion of the petroleum industry over time is marked by key developments. Collectively, they establish the context for the eventual introduction of the Buy Back Contract model until its replacement with the IPC in 2016. Milestone events such as the D'Arcy Concession Agreement in 1901, the first discovery of oil at Masjid-e Soleiman in 1908, and the attempt and failure to revise the 1933 Concession are all of importance. Moreover, both World War One and World War Two had a significant impact on the petroleum industry in Iran and on the 1951 Nationalisation Movement and 1954 Consortium (see Chapter 2, Section 2.6 for further details).

1.2.1 Buy Back Contracts

Broadly speaking, the Buy Back Contract is a concession agreement in which the foreign company foregoes any claim to the profits made following receipt of an agreed remuneration payment. The foreign company also commits to handing over management and operations of the petroleum resources to the NIOC (or a representative) following the finalisation of oil

³¹ 'Bloomberg Tanker Tracking' in Brandon Kochkodin and Grant Smith, 'Iran Ships Most of Its Oil to Asia, Turkey and Europe' *Bloomberg* <<u>https://www.bloomberg.com/news/articles/2018-05-08/iran-ships-most-of-its-oil-to-asia-turkey-and-europe-chart</u>> (accessed 30 June 2019).

extraction and commencement of oil production.³² Van Groenendaal and Mazraati describe the four core elements to an Iranian Buy Back Contracts as follows:³³

- The contract stipulates the yearly capital expenditures during the investment period to establish the total investment by the foreign oil company to be repaid after the stated year. The duties paid on imports by the foreign oil company are reimbursed following the handover of the project to the NIOC.
- The payment of bank charges on the amount invested. The London Inter Bank Offer Rate (LIBOR), in addition to a premium of up to 1%, is the interest rate used for calculating the Buy Back Contract bank charges.
- 3. Remuneration for the resource development activities undertaken by the foreign oil company. The remuneration is paid within an agreed time period, typically between 7 and 12 years. Total remuneration to the foreign oil company may include a reward for the financing of engineering, procurement and project construction services as well as for the transfer of technology.
- A negotiated internal rate of return for the foreign oil company, typically between 12% and 15%. This return rate is based on the agreed investment and production schedule.

When first introduced by the Islamic government in Iran, Buy Back Contracts had the support of the state. Nonetheless, what was particularly problematic was achieving a satisfactory balance between ensuring the contract form generated sufficient revenue for Iran to meet the economic, social, political and resources sector interests of the nation while remaining attractive to foreign entities.³⁴

These issues surrounding the Buy Back Contract arrangements were not limited to the academic and public arenas with disagreements over their structure and use also subjected to 'the intrusion of politics'.³⁵ These disagreements came to a head following the election of President Ahmadinejad in 2009 when the scheme came under heavy criticism from his rivals within the Iranian parliament about the over-involvement of politics in commercial matters. This was in response to what the critics argued was the President tightening 'his grip over the oil sector' in addition to the increase in quasi-state players and domestic contractors with ties

³² van Groenendaal and Mazraati (n 2) 3709.

³³ Ibid 3711–3712.

³⁴ Nima Mersadi Tabari, *Lex Petrolea and International Investment Law: Law and Practice in the Persian Gulf* (Routledge, 2016) 152.

³⁵ William Yong, 'NIOC and the State–Commercialization, Contestation and Consolidation in the Islamic Republic of Iran' (Oxford Institute for Energy Studies, 2013) 11.

to the Islamic Revolutionary Guard Corps (IRGC).³⁶ The constant shifting ground in relation to political support for Buy Back Contract transactions in Iran highlights the importance of the broader political context in Iran to its approach to contracting with foreign entities for the development of its petroleum sector. This is because the political conditions inevitably impacted upon the scheme itself, the stance taken by the media towards the scheme, and the views of foreign entities who entered into partnerships with the Iranian government when signing the contracts.³⁷ In response to changing sentiment towards the Buy Back Contract in Iran, the government ultimately shifted its position to endorse the IPC.

1.2.2 Iranian Petroleum Contracts

The IPC was introduced into Iran in 2016 as a new form of contract. It is essentially a type of service contract whereby a foreign oil company agrees to be paid in sale revenues for extracting an oil or gas resource. The IPC is used in Iran to develop its oil and gas resources. This model is attractive to the government because the risk is carried by the contractors who are required to fund all costs (i.e. technical, exploration, engineering and labour). In exchange, they receive compensation or profit from the discovery and exploitation of a commercially viable resource. The IPC thus emerged as an attempt by the government to reverse the county's reliance on local companies to develop oil and natural gas fields in the face of sanctions imposed by the US. However, as of mid-2018, only two IPCs had been finalised - a July 2017 agreement with French Total and the China National Petroleum Corporation (CNPC) and a 2018 agreement with Russian state-controlled Zarubezhneft.³⁸ Of significance to an understanding of the current IPC are the alternatives advocated by international critics of the model. In particular, the foundational elements and legal limitations of the IPCs, along with the procedural challenges associated with such alternatives, are important for assessing their impact. The theoretical frameworks to explain recent IPC transactions also provide insights into the factors shaping the current IPCs and those driving reform. There has also been extensive consideration in recent years to future developments in the international petroleum contract arena in the context of current sanctions imposed on Iran by the US, as well as the changes surrounding the nature of oil-supply in the Middle East. Arguably for any reform initiative to be successful it must emerge from a

³⁶ Ibid 14.

³⁷ Ibid 4.

³⁸ United States Energy Information Administration, 'Background Reference: Iran' (2019)<<<u>https://www.eia.gov/beta/international/analysis_includes/countries_long/Iran/pdf/iran_bkgd.pdf</u>> (accessed 7 November 2019).

position of mutual agreement and understanding between the Iranian government and the foreign investors. This points to the importance of analysing the respective positions and exploring the potential areas of agreement. The key findings from such an analysis may then be used as a springboard to providing recommendations on how to move forward to address the salient issues around the IPC (see Chapter 5).

Notably, IOCs do not generally enter into IPCs and, as a result, this type of contract remains a relatively under-researched area. This thesis contributes to closing the gap with a comprehensive analysis of the academic and industry literature relating to IPCs. It adopts a multi-disciplinary approach that canvasses the economic, political and legal issues relevant to IPCs. The rationale for analysing IPCs from the economic and political perspectives (in addition to the legal perspective) emerges from the interconnections between the laws and regulations and the broader society and economy. There is significant potential for the constraints placed around Iran's laws on petroleum to transform the nation's political and economic outcomes.

Iran's legal restrictions around foreign investment in its natural resources sector are important in order to understand the current challenges related to petroleum transactions and to predict future trends in development. Indeed, the history of Iran's contract concessions, the Buy Back Contract scheme, the effect of the nationalisation of the resources industries, and now the IPC, all form important elements in the narrative of Iran's effort to control the exploitation of its petroleum resources. They are all a part of the historical development of the contract regime and it is necessary to carefully analyse the elements of the Buy Back Contracts in order to identify which parts have been incorporated into the new IPC and the potential effect.

1.3 Scope of the Thesis

This study aims to critically analyse the current contractual model applied to petroleum transactions in Iran, namely, IPCs. The overarching objective of the analysis is to assess the suitability of IPCs and to propose remedies to the flaws in the contracts that remain aligned with the Constitutional constraints. To propose realistic reforms, it is first necessary to explore and understand the historical factors underpinning the grievances of foreign investors and the Iranian Government in relation to the IPC contract. Chapter Four (see section 4.2) examines International contract schemes and compares them with the IPC in order to understand the status quo and to determine how improvement may be achieved. In order to provide recommendations, this thesis analyses possible legal and political constraints around

future reforms to the IPC, taking into consideration the relevant theoretical and practical context.

Another important objective of this research is to ensure that the scope of the material being studied is sufficiently narrow to support the provision of targeted findings and specific rather than generic research conclusions.³⁹ To support this principle and to optimise the quality of analysis, the scope of this study has been narrowed to petroleum related IPCs; that is, petroleum contracts rather than gas contracts. Gas contracts are discussed at times however only to establish a point of contrast to their oil counterparts.

Although Iran's natural gas deposits, second only in size to Russia's gas deposits, are included in the definition of petroleum, they are not examined for three reasons. First, there are fewer sources available for an analysis of Iran's gas contracts than for IPCs. Second, gas exports from Iran commenced as recently as the 1960s and, as a result, there is not the historical influences and aspects for analysis as is the case for IPCs. Third, and most importantly, gas contracts in Iran have almost identical terms and conditions to those included in petroleum contracts, with only minor technical details comprising the points of difference. Hence, focusing on petroleum contracts (including greater access to data sources) also permits a developing understanding of gas contracts in Iran, with references to the latter providing a useful platform to demonstrate points of similarity and difference.

1.4 Research Questions

To achieve these aims and to respond to the present gap in the research literature, the thesis addresses the following research questions:

- 1. What are the strengths and weaknesses of the Iran Petroleum Contract in regard to the international petroleum trade?
- 2. How did the Buy Back Contract model impact Iran's participation in the international petroleum trade?
- 3. To what extent does the Iran Petroleum Contract best address the current and potential needs of the oil resources industry in Iran?

All three research questions are relevant to Iran's operations within the international petroleum trade, which is generally defined as the trading of crude oil among supplier and

³⁹ David Silverman, *Interpreting Qualitative Data: Methods for Analysing Talk, Text and Interaction* (Sage Publications, 1993) 1.

user nations at designated oil prices. Research question 1 (RQ1) focuses on the IPC within the international petroleum trade framework and is important given that crude oil and petroleum products remain vital for the development of countries all around the world. Indeed, petroleum represents the largest primary commodity of international trade for both volume and value, with the international community dependant to a significant degree on access to affordable oil within the international marketplace.⁴⁰ By focusing on the strengths and weaknesses of the IPC within the context of the international trading market place, insights can be gained in relation to its capacity to achieve its stated objectives (as determined by the Iranian Government) and for informing decision making on how the contract option may be improved more generally.

Regarding research question 2 (RQ2), an effective way to better understand the strengths and weaknesses of the IPC as a contract option for Iran is to compare it to its predecessor; that is, the Buy Back Contract. In this way, RQ2 is important because it introduces the concept of obsolescing bargains into the discussion in this thesis of regarding the motivations and objectives of Iran for transitioning to the IPC. To explain, obsolescing bargains in the petroleum trade are those contract or other forms of interaction models that exist between the IOC and host country (i.e. Iranian Government).⁴¹ Such 'bargains' initially favour the IOC but, over time, the bargaining power shifts more towards the host government as the fixed assets of the IOC in the host country increase over time.⁴² It is anticipated that investigating how the Buy Back Contract model impacted Iran's participation in the international petroleum trade in the context of its pathway towards obsolescence will provide important insights into the political processes and efforts towards the control of Iran's energy resources by the Iranian Government laid the foundation for the introduction of the IPC. Then, research question 3 (RQ3) ensures that explicit focus is given in this research investigation to the extent to which the IPC may be considered as success. As implied in the question, success will be evaluated in broad terms according to the extent to which it emerges that the IPC meets the needs (actual or perceived) of the oil resources industry in Iran. However, given the design complexities of the different petroleum contract models combined with the competing needs of all stakeholders in the contract, the success of the IPC model

⁴⁰ Zhijie Zhang, Huiqing Lan, and Wanli Xing, 'Global Trade Pattern of Crude Oil and Petroleum Products: Analysis Based on Complex Network' (2018) 153(2) *IOP Conference Series: Earth and Environmental Science* 1.

⁴¹ Abdelrehim Neveen and Steven Toms, 'The Obsolescing Bargain Model and Oil: The Anglo-Iranian Oil Company 1933–1951', (2017) 59(4) *Business History* 555, 555.

⁴² Ibid.

must also be evaluated against several other outcomes. The principal outcomes among them for consideration in this thesis include the extent to which the IPC appeals to IOCs (in terms of investment risk and return) and therefore encourages uptake, the degree to which it strikes the right balance regarding compliance with Iran's Constitution and other legal frameworks around control and ownership of domestic natural resources, and how well it achieves its primary objectives to inject revenue, technology, and know-how into Iran's energy sector. It is anticipated that the combination of providing detailed answers to these three questions will provide a string platform for assessing the merits of the IPC model in relation to the specific needs of Iran's oil resources sector.

1.5 Methodology

The research in this thesis is a systematic inquiry into the questions outlined above.⁴³ The overarching purpose of the research is to address these questions and to provide meaningful explanations for the proposed conclusions.⁴⁴ The particular methodological design of this study was adopted to permit conclusions to unfold in a logical, useful, and effective way.⁴⁵ In turn, this study adopted a qualitative interpretive methodology to examine the IPCs currently being implemented in Iran to engage foreign investors in the exploitation of the nation's petroleum reserves.⁴⁶ Barkhuizen and Ellis remind us that 'there is no one way of doing qualitative research'⁴⁷ and this thesis relies on a document analysis approach primarily, with some minor doctrinal research. The adoption of qualitative interpretive methodology is justified for two reasons. First, the interpretivist principles underlying qualitative research establish a robust platform for an examination of the IPC models for their capacity to benefit the energy sectors in Iran. Applying qualitative paradigms to the analysis of primary and secondary documents thus permits an interpretative understanding of the nature of IPCs and their potential benefits using inductive discovery processes.⁴⁸ Second, qualitative research allows for an insider's perspective of the topic to be formulated.⁴⁹

⁴³ John W Creswell, *Qualitative Inquiry and Research Design: Choosing Among Five Approaches* (Sage, 2012)24.

⁴⁴ Ibid.

⁴⁵ Louis Cohen, Lawrence Manion and Keith Morrison, *Research Methods in Education* (Routledge, 7th ed, 2007) 33.

⁴⁶ Ted Benton *Philosophical Foundations of the Three Sociologies (RLE Social Theory)* (Routledge, 2014). 23.

⁴⁷ Gary Barkhuizen and Rod Ellis, *Analysing Learner Language* (Oxford University Press, 2005) 254.

⁴⁸ Benton (n 46) 23.

⁴⁹ Robert E Burns, Introduction to Research Methods (Longman, 4th ed, 2000) 11.

In terms of our understanding of the merits and deficiencies of IPCs in Iran compared to the previous Buy Bank Contract model, qualitative research permits historical, theoretical and analytical comparisons to be made. In turn, the document analysis research method has been applied to facilitate this comparative analysis. The analysis aims to apply at times relevant legal concepts and principles (i.e. cases, statutes and rules) to examine relevant legislation in a bid to explore their political, economic and social implications around the research topic under investigation.⁵⁰ Moreover, a traditional document analysis method has been applied within a two-part process. To clarify, the first part involved identifying and accessing relevant sources of law and the second part involved an interpretation and analysis of the text in terms of the deductive logic and /or inductive reasoning that has been applied within it.⁵¹ To better understand the application and implications of the concepts and legal principles related to the petroleum contracts in Iran, industry and research literature including journal articles, industry publications, government reports, and media reports and the like have been utilised. The primary objective here was to read, analyse and link the industry, government, research and media source information to the relevant body of law to draw conclusions about the merits of the contract forms in relation to their stated legal, economic and social objectives.

As such, qualitative research paradigms are integral to the research methods applied in this study to enable the interpretation and analysis of the context in which the legal principles are being applied.⁵² The qualitative component of this research is appropriately described by Hutchinson and Duncan as an analytical reasoning process in which the outcome is contingent upon the experience of the individual.⁵³ In this thesis the process undertaken was a problem-based research methodology. Hutchinson and Duncan describe this as a seven-step process:⁵⁴

- Gathering relevant facts,
- Identifying the legal issues,
- Analysing the legal issues with the aim to identify relevant law,
- Conducting background research,
- Identifying and accessing primary material (e.g. legislation),

⁵⁰ Terry Hutchinson and Nigel Duncan, 'Defining and Describing What We Do: Doctrinal Legal Research' (2012) 17(1) *Deakin Law Review* 83, 84.

⁵¹ Ibid 85.

⁵² Ibid.

⁵³ Ibid 86.

⁵⁴ Ibid 106.

- Synthesising all issues in context, and
- Drawing a tentative conclusion.

The analysis and interpretation of industry and institutional documents 'have been a staple in qualitative research for many years.'⁵⁵ The standard approach to document analysis was applied in this research; namely, a systematic review (and evaluation at times) of printed and online documents with the aim to elicit meaning, develop an understanding, and acquire empirical knowledge.⁵⁶ While undertaking the interpretation and analysis of the primary and secondary documents, the overarching aim was to ensure a balanced approach was adopted. That is, regard was shown towards the national interests of the Iranian government in relation to the development of the nation's upstream oil and gas sector as well as to the interests of foreign investors. As such, the primary focus of the analysis in this thesis is on the current legal regimes in Iran governing petroleum contracts (i.e. IPCs and Buy Back Contracts) for the development of Iran's upstream oil industry.

However, given the political sensitivities around the formulations of these contract schemes and the extremely limited access to primary commercial documents due to confidentiality concerns, it has been necessary in this thesis to rely on secondary resources of the type identified above. More specifically, data for analysis and discussion in relation to the three research questions was primarily drawn from relevant industry and academic publications. Examples of the former include, but are not limited to, publications from the Energy Information Administration, International Monetary Fund, Iran Business News, the Organisation for Economic Cooperation and Development (OECD), Organisation of the Petroleum Exporting Countries, Tavana Energy, the United States Energy Information Administration, and the World Energy Council. Examples of the latter include, but are not limited to, journal publications such as the *Global Trade and Customs Journal, Journal of World Energy Law & Business, Iranian Studies*, and the *Oil, Gas & Energy Law Journal*. This is in addition to several book publications, Working Papers, Conference Papers and interview transcripts.

The discussion of these publications for their main points and insights is then combined with a discussion of primary sources related to the legal frameworks in Iran relevant to nation's energy sector and the utilisation of its natural resources. Principal among these is the

⁵⁵ Glenn A Bowen, 'Document Analysis as a Qualitative Research Method' (2009) 9(2) *Qualitative Research Journal* 27, 27.

⁵⁶ Juliet Corbin and Anselm Strauss, 'Strategies for Qualitative Data Analysis' in *Basics of Qualitative Research. Techniques and Procedures for Developing Grounded Theory* (Sage, 4th ed, 2014) 85.

Constitution of Iran, however others include, but are not limited to, Iran's Annual Budget Law, and the objectives of the nation's Five-Year Economic, Social and Cultural Development Plans. These primary resources for data collection and discussion are written in Farsi in their original form. Given the Australian context in which this thesis is produced, however, English translations of the documents were utilised and included as referenced material where available. As a person who is proficient at speaking and writing in both Farsi and English, the reliability of the translations of the primary resources was assessed by me prior to their use.

In turn, the comprehensive review of the secondary literature against relevant primary sources, combined with the comparative analysis of the IPC and Buy Back contract models, lays the foundations for the recommendations at the conclusion of this thesis pertaining to the future role of IPCs in the development of Iran's upstream energy sector.

1.6 Organisation of this Thesis

The current position of Iran's petroleum contracts is important given the current context of fluctuating oil markets due to the instabilities or uncertainties surrounding some major petroleum producers around the world. As the world's leading oil producer, Iran arguably has the ability to determine the very nature of the global petroleum market. In this context, this thesis provides a comprehensive examination of Iranian petroleum contracts in relation to the present issues and their prospects in the future. With a focus on IPC transactions, this thesis comprises six chapters, which collectively analyse and report on the historical, social and commercial factors that shape the current contractual relationships undertaken in the petroleum industry in Iran. Chapter 1 has introduced the main points of focus in this thesis along with the rationale for and importance of the research. The research questions underpinning the study were also outlined. This chapter also provided a brief overview of the importance of petroleum contacts and the industry.

Chapter 2 presents an analysis of the historical factors impacting the IPC. The objective for beginning in this way is to lay the foundation for an in-depth analysis of the IPC provided in the subsequent sections. Specifically, Chapter 2 provides a short history of concession contracts in Iran from their introduction to the modern form. It explores the exploitative nature and outcomes of the 1933 Concession, including the unsuccessful attempts by Iran to renegotiate the deal to avert significant disruption to its petroleum industry. The analysis also focuses on the consequences the 1951 Nationalisation of the petroleum industry, including the attempts made to avoid subsequent annulments by altering the contract terms, as per the

1954 Consortium. The subsequent changes to the legal framework underpinning the petroleum contracts and the extent to which they impacted the commercial landscape are explored. The objective in this regard is to better understand the opposing interests and how they may be reconciled given the importance of petroleum industry to Iran's ambitious economic reforms. As such, as review is provided of the failure to address the concerns of the Iranian public and authorities, leading to the 1979 Revolution and the subsequent cancellation of all petroleum contracts. The review also considers the impact of the Iran-Iraq War on Iranian infrastructures and the increasing need for foreign investment and the impact of current sanctions.

Chapter 3 presents a critical analysis of the regulatory framework underpinning the petroleum contract agreements in Iran. It examines the Constitutional provisions relating to petroleum, the standing of the state-owned companies, particularly the NIOC, and the legislative provisions relating to contracts. Focus is given to the 'restrictions' on petroleum contracts in Iran emerging from the Constitution and statutory law. Also examined in this chapter are the Five-Year Development Plans implemented in Iran and the introduction of Foreign Investment Laws and the extent to which they placed constraints around foreign involvement in Iran's petroleum industry along with the IPC scheme.

Chapter 4 then provides a comparison of the contract models utilised in Iran's petroleum sector and the international alternatives. It explores the broader context of petroleum contract by identifying and discussing the alternatives to the contract model applied in the international petroleum industry. The alternatives discussed include, but are not limited to Production Sharing Agreements, Revenue Sharing models and Joint Venture Agreements (JVAs). The examination focuses on their operationalisation and the legal foundations upon which they are based and explores the extent to which they are compatible (legally and operationally) with the current contract system and legislative framework in Iran. The objective underpinning this analysis is to better understand the criticisms made by Iranian and foreign entities in relation to the contract structures and how these criticisms may have contributed to the transition towards the IPC system.

Chapter 5 provides an analysis of the IPC model specifically, using the Buy Back Contract as the model for comparison. The 'pros' and 'cons' of each contract model are identified and discussed. The result of this examination is then used as a springboard to make a recommendation as to the most suitable contract form (from the Buy Back Contract and IPC options) to best meet the interests of the NIOC as a contracting party and the oil and gas sectors in Iran more broadly. Chapter 5 considers whether the IPC is fit for purpose and is the best for Iran in light of the sanction regime. As such, an evaluation is made as to whether there are other contracts or forms that are better suited for Iran moving forward. In addition, the practical implications of implementing some of the controversial provisions are explored to ascertain whether they strengthen or weaken the integrity and appeal of the IPC model. The main objective of the comparative analysis is to provide a springboard into an exploration of the potential for reforms to modernise the contract scheme in Iran to align with current global market conditions. As such, a recommendation is also included in this chapter as to which contract model should be pursued in modern-day Iran to best manage and exploit the nation's vast petroleum reserves.

Chapter 6 concludes this thesis. It includes a summary of the main findings from the analyses performed and provides recommendations for the modernisation of Iran's petroleum contracts. The recommendations consider the concerns of both Iran and foreign entities and the current constraints imposed by the Constitution in Iran and its legal framework.

2 IRAN'S PETROLEUM INDUSTRY: INFLUENCE OF PRE-NATIONALISATION (1908-1951) CONCESSIONS, NATIONALISATION AND SANCTIONS ON MODERN PRETROLEUM CONTRACTS

2.1 Introduction

This chapter examines the history of oil and gas Concessions in Iran. This is an important step towards understanding the purpose and form of the recently adopted IPC and for providing a general contextual background to this study. The examination begins with a broader contextual discussion of the global upstream oil and gas sector and Iran's place within that sector. The focus then shifts back to the introduction of the pre-1900 petroleum Concessions in Iran and traces the developments and changes to the petroleum contracts up to the nationalisation of the petroleum industry in Iran in 1951. The chapter then concludes with a discussion of the changes to contractual and commercial agreements in Iran with foreign investor companies following the nationalisation of the industry. This includes focusing on the fundamental changes to the terms and conditions of petroleum contract models in Iran following the Islamic Revolution in 1979 as well as the impact of the Iran-Iraq wars on the design of petroleum contract.

2.2 The Global Upstream Oil and Gas Sector

The oil and gas industry is comprised of three major sectors: <u>upstream</u>, midstream, and downstream.⁵⁷ The most capital-intensive and important of the three is the upstream sector because this is where the crude oil and natural gas are produced. All activities relating to crude oil and natural gas exploration and extraction occur in the upstream sector which is done prior to transporting the products to refineries to be processed.⁵⁸ Notwithstanding the increase globally in the use of renewables to facilitate electricity generation since the turn of the century (e.g. wind, solar, biofuels, hydro and so forth), several countries maintain a significant level of energy production by burning fossil fuels (e.g. crude oil, coal, natural gas).⁵⁹

⁵⁷ Shafiee, Mahmood, Isaac Animah, Babakalli Alkali and David Baglee, 'Decision Support Methods and Applications in the Upstream Oil and Gas Sector' (2019) 173(1) *Journal of Petroleum Science and Engineering* 1173.

⁵⁸ Ibid.

⁵⁹ Ibid.

Indeed, oil is still the primary source of fuel globally – at around 33% of all energy consumption – followed by coal at around 29% and natural gas at around 24%.⁶⁰ This is largely because upstream oil and gas industries around the world continue to seek new ways to increase volumes of production, achieve cost reductions, enhance safety, improve operational performance, and ensure environmental protections.⁶¹ Such efforts involve complex decision making and the need to consider and address uncertainties and risks in regard to a range of factors including field exploitation, production and maintenance, and life extension and decommissioning.⁶²

Notwithstanding the status of oil as the primary source of fuel globally, Bagheri and Di Minin⁶³ note that IOCs' share of oil and gas reserves has fallen from around 85% in 1970 to less than 10% today. This raises considerable concerns about the business models of IOCs and their sustainability. Added to this is the change in the competitive landscape of the global upstream petroleum industry which has seen IOCs experience significant constraints around access to oil and gas resource bases worldwide.⁶⁴ In addition, research evidence shows knowledge retention initiatives and activities in oil and gas companies globally is inconsistent, with little effort being made to mitigate impacts from the knowledge loss from older employees.⁶⁵ A range of factors have contributed to this outcome including falls in the price of oil, layoffs, political conditions, and the country locations of the companies.⁶⁶ This loss of knowledge is more impactful in the upstream oil and gas sector as it is most affected by falls in oil prices and because knowledge around explorations, drilling, and reservoir management are critical to operations.⁶⁷

⁶⁰ World Energy Council, World Energy Council World Energy Issues Monitor 2017 <www.worldenergy.org> 60.

⁶¹ Shafiee (n 57) 1173.

⁶² Isaac Animah and Mahmood Shafiee, 'Condition Assessment, Remaining Useful Life Prediction and Life Extension Decision Making for Offshore Oil and Gas Assets' (2018) 53(1) *Journal of Loss Prevention in the Process Industries* 20.

⁶³ Seyed Kamran Bagheri and Alberto Di Minin, 'The Changing Competitive Landscape of the Global Upstream Petroleum Industry' (2015) 8(1) *The Journal of World Energy Law & Business* 1.

⁶⁴ Ibid.

⁶⁵ Muhammad Saleem Sumbal, Eric Tsui, Eric See-to and Andrew Barendrecht 'Knowledge Retention and Aging Workforce in the Oil and Gas Industry: A Multi Perspective Study' (2017) 21(1) *Journal of Knowledge Management* 180, 180.

⁶⁶ Ibid.

⁶⁷ Ibid 181.

2.2.1 Iran's place within the global upstream oil and gas sector

Iran remains one of the world's and the Middle East region's main suppliers of energy, gas in particular. In turn, oil and gas resources are important to Iran's energy policy for two main reasons: Iran derives most of its energy from hydrocarbons, and oil and gas provide Iran with a point of connection to the international community along with regional and global energy markets.⁶⁸ Revenues from the sale of oil and gas are used by the government of Iran to deliver economic and social transformations such as industrialisation and modernisation as well as to keep the status quo in domestic power dynamic and relations.⁶⁹ Hence, Iran is eager to maintain its role in the global oil and gas sector due to the politically influenced nature of the Iranian economy combined with the reality that the economic performance of the nation relies on the effectiveness of its oil and gas sectors.⁷⁰

In terms of Iran's place with the global oil and gas sector, the International Energy Agency reports that global primary energy needs are expected to increase by 55% up to 2030. A key implication of this is that oil and gas industries around the world as the main suppliers of energy will be crucial to support societal advancement and development.⁷¹ Growth in the global oil and gas sector is further evidenced in the increasing demand for both oil and gas fields. Regarding the demand for oil specifically, this increased from 894,283,000 barrels per day in 2012 to 951,151,000 barrels per day in 2016.⁷² As a significant contributor to OPEC – with an estimated 158 billion barrels of proved crude oil reserves – Iran remains an appealing target market IOCs including TOTAL, BP and others.⁷³ Moreover, as a global player, Iran is a primary supplier of fossil energy to China and India – the world's two highest populated countries – with China especially continuing to demonstrate economic growth and a willingness for infrastructure development.

Nonetheless, the volatility of the global export energy market always has the potential to impact Iran's domestic economy as well as its role in the global energy market. This is evident in the revenue received by Iran from its oil and natural gas exports over recent years.

⁶⁸ Tamás Dudlák, 'After the Sanctions: Policy Challenges in Transition to a New Political Economy of the Iranian Oil and Gas Sectors' (2018) 121 *Energy Policy* 464, 465.

⁶⁹ Benjamin Smith, *Hard Times in the Lands of Plenty. Oil Politics in Iran and Indonesia.* Cornell University Press, 2007) 23.

⁷⁰ Dudlák (n 68) 464.

⁷¹ Sweis, Rateb, <u>Moarefi, Alireza;</u> <u>Mahmood Hosseini Amiri;</u> <u>Moarefi, Soad;</u> <u>Saleh, Rawan</u>. 'Causes of Delay in Iranian Oil and Gas Projects: A Root Cause Analysis' (2019) 13(3) *International Journal of Energy Sector Management* 630, 630.

⁷² Ibid.

⁷³ Ibid.

Based on figures from the International Monetary Fund,⁷⁴ the revenue received by Iran from its oil and natural gas exports fell from US\$55.4 billion in 2014-2015 to U\$33.6 billion in 2015-2016, representing a decrease of almost 40%. The reasons for this significant decrease included the ongoing slump in export volumes combined with a fall in the price of crude oil leading to lower total export revenue. However, it was estimated that revenue from oil and natural gas exports in Iran rose to US\$57.4 billion in 2016-2017 – mostly from crude oil and condensate exports. This was due to a rise in the volume of crude oil exports after the implementation of the JCPOA.⁷⁵

Within the context of this volatility and the competing driving forces, however, the petroleum sector in Iran has been experiencing a decline due to both external sanctions and the lack of investment in the development of new capacities.⁷⁶ International sanctions imposed on Iran and the disruptions they have brought to its oil and gas sectors have unquestionably had an influence on Iran's domestic affairs as well as its international status.⁷⁷ Given the significant revenue to Iran derived from the sale of its oil and gas, the raw materials form the basis of the national economy.⁷⁸ In addition, the raw materials are strategic and political instruments that shape Iran's domestic and foreign policies and are often used in diplomacy by the policy makers. Furthermore, the development of Iran's strategic sectors; namely, its military and energy sectors are concentrated in the hands of Iran's political leadership. Hence, the importance of the interrelationships between politics and hydrocarbon management in the country cannot be over-stated.⁷⁹

Following the partial lifting of the international sanctions during the early twenty-first century, Iran sought to implement significant changes in its energy sector be in a position to provide a new, favourable environment for foreign investments.⁸⁰ Despite this intention, however, the relatively limited growth in production combined with a significant increase in domestic consumption continue to shape the future potential of Iran's energy sector

- ⁷⁹ Ibid.
- ⁸⁰ Ibid.

⁷⁴ International Monetary Fund, Article IV Consultation, <u>Islamic Republic of Iran</u> (IMF Country Report No. 17/62, February 2017) 4.

⁷⁵ Ibid.

⁷⁶ Bijan Khajehpour. The Future of the Petroleum Sector in Iran, *Global Transitions* (Legatum Institute, 2013), 1.

⁷⁷ Ibid.

⁷⁸ Ibid.

specifically and its place in the global upstream oil sector more generally .⁸¹ Therefore, of particular importance to Iran's role in the global oil and gas sectors are its ability to maintain oil production capacity in the context of international sanctions, the down turn in the global hydrocarbon energy sector and the shifting landscape of domestic energy consumption as the country undertakes subsidy reforms and pursues better energy efficiency.⁸²

There is little doubt that Iran is transitioning away from its reliance on the exporting of crude oil towards a more diversified energy export portfolio including pipeline gas and gas-based industrial products, petroleum products, and electricity.⁸³ From a regional perspective, Iran would consider itself as a leading energy provider and therefore would likely want to work from the position that energy interdependency is a pathway to mutual cooperation. While the recent international sanctions have slowed down the development of Iran's energy sector, they have arguably compelled the nation to become a significant regional and global player in the production and exporting of energy-related products and services.⁸⁴ The pathway towards tension-free relations between Iran the West has many complexities and requires years to achieve, but Iran has the natural and human resources and the geographical positioning to establish itself as a leading player in a regional energy hub. Of course, there is still uncertainty around whether the recent political and economic reform initiated by the Iranian government and the international sanctions and pressure will lay the foundation for a petroleum sector in Iran that is more transparent and accountable. Moreover, disruptions from the sanctions as well as from the domestic mismanagement of the energy sector place pressure on Iran's role in the global oil and gas sector.⁸⁵ However, oil and gas resources remain a significant source of revenue for Iran and key stakeholders will naturally be determined to maintain a strong role for Iran in the global supply of energy.

2.3 Pre-1900 Oil Concessions

Concessions or license agreements are a type of "one-sided contract" first introduced during the late 1800s, becoming more widespread during the early 1900s.⁸⁶ They describe the

⁸¹ Ibid.

⁸² Ibid 2.

⁸³ Ibid 6.

⁸⁴ Bijan Khajehpour Ibid 6.

⁸⁵ Ibid.

⁸⁶ R Global, 'Petroleum Contract Models' <<u>https://irglobal.com/article/petroleum-contract-models-0f76</u>> (accessed 12 December 2019).

agreement entered into by a State and a company for the latter to explore and develop the natural resources of a designated area over an agreed (fixed) period of time. Oil companies typically competed for the concession by putting forward a bid, generally combined with a signing bonuses to gain the license to the concession rights.⁸⁷ As with most mining concessions, the resources (i.e. oil and gas) underground belong to the State and not to the owner of the land. As such, the State - that is, the Government or a designated legal Authority - grants the rights and obligations to a company for to explore a designated 'block' of agreed territory (onshore or offshore) for the potential to develop and produce oil and gas resources.⁸⁸ Concessions agreements are generally long-term (e.g. at least 20 years) and based on a two-phase financial structure: exploration and production. Following the granting of a concession license to an IOC or companies, a fee is paid by the licensee to the State per barrel for the right to produce the oil or gas, generally calculated according to market price. All costs and risks are borne by the operating company, but all profits exceeding the agreed fee go to the operating company.⁸⁹ Modernised concessions are different to the traditional types in that there is generally a smaller concession area, for a short duration, increased control by the host state, equal profit sharing and updated bonus / tax payments, and a 'far more dynamic and flexible [approach to] accommodating different perspectives and interests of the contracting parties'.90

The number of concessions granted to foreign entities increased during the latter half of the 19th century. Included in the concessions was the right to undertake the development facilities, a right which could be bought with a cash payment or by negotiating other outcomes.⁹¹ The first of many concessions was granted to a British company to construct a telegraph network. Although the network would provide a quicker communication pathway into India, it effectively handed over control of Iran's technological developments to a foreign power.⁹²

Iranian concessions to foreign entities during the 19th century were broad in scope and often achieved as a result of political pressure.⁹³ The government of Iran at this time was

92 Ibid.

⁸⁷ Keith William Blinn et al. *International Petroleum Exploration and Exploitation Agreements: Legal, Economic and Policy Aspects* (Barrows, 2nd ed, 2009) 47.

⁸⁸ Mahmoud Fard Kardel, *The Development of Iran's Upstream Oil and Gas Industry: The Potential Role of New Concession Contracts* (Routledge, 2020) 72.

⁸⁹ Ibid 72–74.

⁹⁰ R Global (n 91).

⁹¹ Steve Muhiberger, *History of Islamic Civilization* (Nipissing University, 1999) 2.

⁹³ This pressure was effective because Iran was in a crossfire between regions controlled by the British Empire and regions controlled by Russia.
experiencing fiscal difficulties due to its debts to Russia and the UK.⁹⁴ In addition, the chaotic approach to governance in Iran prompted foreign powers to increase their concession demands on Iran. The British, particularly, took advantage of the chaos by establishing a monopoly over the production and sale of the natural resources of Iran.⁹⁵ Indeed, with the Concession terms and conditions favouring the foreign entity there was considerable potential for the national resources of Iran to be exploited. As such, the Concessions scheme was particularly unsustainable given the prohibitions placed around additional commercial engagements between the parties, leaving Iran to feel resentful that it was being exploited.⁹⁶ The following section provides a brief overview of the major concessions in this period.

2.3.1 The de Reuter Concession

On 25 July 1872 the first petroleum contract was negotiated between Baron Julius de Reuter of Britain and Shah Nasr-ed-Din of Persia. de Reuter had considerable influence in European (including British) financial and political circles at the time⁹⁷ and ultimately secured a wide-scoping Concession agreement.⁹⁸ de Reuter agreed to finance natural resource ventures in Persia (now Iran) following the visit by de Reuter's agent, Cotte, to Tehran in 1872. During this visit Cotte offered cash to Shah Naser-ed-Din Shah in exchange for a Concession. Although the Concession did not target oil reserves directly, its terms were sufficiently broad to cover these resources.⁹⁹ Included in the Concession was exclusive rights to Persia's mineral reserves excepting gold, silver, precious stones for 70 years, along with the sole right to the construction of rail and tram networks, the building of canals and irrigation systems, access to forests and uncultivated land, and the establishment of a bank and public works of every description.¹⁰⁰

Lord George Curzon, the eldest son of Baron Curzon and the UK Foreign Secretary in 1919, said at the time that the Concession marked a total and 'extraordinary surrender'¹⁰¹ by Persia to the interests of foreign entities. Put simply, Lord Curzon described it as 'the most complete concession which a nation can ever grant, giving up all its possession to foreigners'.¹⁰² However, the Concession was withdrawn after 15 months as a result of both internal (the

⁹⁴ Nasar Farshadgohar, *A Survey on Iran's Oil Agreements (Seyri Dar Gharardadhaye Naftiye Iran)* (Tehran, 1381) 79.

⁹⁵ Ibid.

⁹⁶ Muriel Atkin, Russia and Iran, 1780–1828 (University of Minnesota Press, 1980) 147.

⁹⁷ He is more well known outside of Persia as the founder of the news agency.

⁹⁸ Laurence Paul Elwell-Sutton, 'The Iranian Press, 1941–1947' (1968) 6(1) Iran 65, 67.

⁹⁹ Ibid.

¹⁰⁰ Ibid.

¹⁰¹ George N Curzon, Persia and Persian Question (Cambridge University Press, 2015), 480.

¹⁰² Rabiei Faranak, *Contracts Law* (Behnami, 2002) 71.

Iranian public) and external (Russia) objections.¹⁰³ The British government also withdrew support for de Reuter due to concerns around international tensions leaving the Shah to cancel the Concession¹⁰⁴ and confiscate the £40,000 deposit de Reuter paid to the Persian government.¹⁰⁵ Notwithstanding the short duration of the Concession agreement, it laid the foundation for Iranians to be highly cautious of foreign entities exploiting its national resources.¹⁰⁶

2.3.2 Hotz Concession

A Concession of note early in the 19th century was that granted to the Hotz Company of Britain Operating out of Booshehr in the Persian Gulf. The Hotz Company was granted a Concession in 1884 giving it the right to the extraction of oil from the Dalky and Qeshm Islands.¹⁰⁷ Despite several drilling attempts to discover oil through 1890 to 1893, none was found and after the annulment of all mining concessions by the Persian Government in 1899, the Hotz Concession was voided in 1901.¹⁰⁸

2.3.3 Imperial Bank of Persia Concession

After the de Reuter Concession was annulled, the company remained committed to making a return on its investment. In 1889, the British Minister, Sir Henry Drummondolf, sought agreement from the Chief Minister of Iran, Amin al Soltan Atabak, for elements of the de Reuter Concession to be enacted.¹⁰⁹ A new concession agreement was eventually agreed on which included the creation of the Imperial Bank of Persia.¹¹⁰ The terms of the agreement included 14 articles including exclusive rights to mint banknotes, along with the payment of one million Francs (around £40,000) by de Reuter to Naser-Aidin-Shah by way of a loan, at 16% interest per annum. The £40,000 deposit previously 'confiscated' by the Persian government was used as capital by the Imperial Bank.¹¹¹

Article 11 stated:¹¹²

¹¹⁰ Rabiei (n 107) 21.

¹⁰³ Elwell-Sutton (n 103) 12

¹⁰⁴ Ibid.

¹⁰⁵ Hamed Alavi, *History of Oil Industry in Iran*, (California Institute of Asian Studies, 1978) 26.

¹⁰⁶ Amir Hossien Mahdavi, *History of the Iranian Foreign Relations (Tarikhe Ravabete Kharejiye Iran)* (Tehran, 1985) 308.

¹⁰⁷ Hooshang Guilak, *Fire Beneath the Ashes: The United States and Iran: a Historic Perspective 1829–1947* (Xlibris Corporation, 2011) 60.

¹⁰⁸ Ibid.

¹⁰⁹ Article 20 of the Concession dealt with the introduction of a bank, hoping to prevent the dissolution of Reuters.

¹¹¹ Ibid.

¹¹² Iraj Zoghi, Economic and Political Issues of Iran's Oil (Masael Eghtesadi va Siyasi Naft Iran) (Pazhang, 1962), 59.

As the Imperial Bank declares that they are ready to exploit all of the natural minerals, everywhere in the country, immediately, the government will give a concession to this bank to exploit all minerals, including iron, copper, lead, coal, oil, and manganese; provided that the government had not granted this to others in the past. If the bank does not start to exploit any particular resources within ten years from the time that the bank was established, the government would assume that they had abandoned their title to those minerals.

It took almost 17 years for de Reuter to resolve its issues with the Iranian government regarding the concession and ultimately achieve an advantage in terms of oil deposits.¹¹³

2.4 Twentieth Century Pre-Nationalisation Oil Concessions

This section discusses the 1901 D'Arcy Concession and other significant oil negotiations during the early 20th century such as the 1933 Concession. Focus is given to the privileges the agreements afforded foreign entities and the efforts by Iran to renegotiate the contracts due to pressure from the Iranian public. In sum, the common concession agreements signed in the Middle East prior to the 1950s had the following characteristics:¹¹⁴

- The rights granted to foreign petroleum companies to explore for petroleum reserves to develop often covered large expanses of area, sometimes even an entire country.
- The duration of the contracts covered an extensive time period.
- The foreign petroleum company had control over the schedule and manner of the development of the mineral reserves.
- The foreign petroleum companies were not required to produce a set number of barrels per day. As a result, the petroleum companies could reduce or increase petroleum production based on falls or increases in the price of petroleum without incurring a penalty. This left the host nation with limited rights other than the receipt of a payment based on production.

¹¹³ Farshadgohar (n 99) 39.

¹¹⁴ Homayoun Mafi, 'Iran's Concession Agreements and the Role of the National Iranian Oil Company: Economic Development and Sovereign Immunity' (2008) 48(1) *Natural Resources Journal* 407, 412-414; Valerie Marcel, *Oil Titans: National Oil Companies in the Middle East* (Brookings Institution Press, 2006) 16-20.

2.4.1 D'Arcy Concession and foreign involvement

The signing of the Concession to William Knox D'Arcy in 1901 marked the 100-year anniversary of the foundation of Iran's petroleum industry.¹¹⁵ With the support of the authorities in Persia, the concession was granted to D'Arcy by Mozaffar Ed-Din Shah on 28th May 1901 in return for £50,000 worth of shares and other payments.¹¹⁶ In accordance with Article 4 and Article 16 of the D'Arcy Concession, May 1903 saw the first company established for the exploitation of oil.¹¹⁷ Despite expert opinion being optimistic about the potential of the drilling operations, the lack of financial support from D'Arcy was a major obstacle to success. D'Arcy was not prepared for the significant costs incurred to support related projects. By 1904 he had outlaid more than £220,000, a huge amount of money at that time.¹¹⁸

The harsh provisions of the D'Arcy Concessions for Iran laid the foundation for the Shah government's decision to regain control of the oil field for Iran and to limit the increase in foreign influence.¹¹⁹ The concession provisions regarded as most controversial included: Article I

The Government of His Imperial Majesty, the Shah, grants to the concessionaire by these presents a special and exclusive privilege to search for, obtain, exploit, develop, render suitable for trade, carry away and sell natural gas, petroleum, asphalt and ozokerite throughout the whole extent of the Persian Empire for a term of sixty years as from the date of the signing.¹²⁰

The extent to which oil exploration and exploitation was allowed under Article 1¹²¹ is extraordinary given that it appears to grant exclusive rights for the exploitation of oil fields over a significant area of land the subject of the concession for 60 years. Furthermore, the Article covered an unparalleled amount of valued resources and supplies relative to the other petroleum contracts which were much smaller in terms of land area and scope: Article 7

¹¹⁵ Alavi (n 110) 30.

¹¹⁶ Ibid.

¹¹⁷ Sundhya Pahuja, Rethinking Iran and International Law: The Anglo-Iranian Oil Company Case Revisited. In James Crawford, Abdul Koroma, Said Mahmoudi, Alain Pellet (eds) *The International Legal Order: Current Needs and Possible Responses: Essays in Honour of Djamchid Momtaz* (Brill 2017) 55.

¹¹⁸ Ibid.

¹¹⁹ Zoghi (n 117) 63.

¹²⁰ Ibid.

¹²¹ Some provinces were open to exclusion as stipulated in a later Article.

All lands granted by this agreement to the concessionaire, or that may be acquired by him in the manner provided in article 3 and 4 of these agreements, and also all products exported, shall be free of all custom duties and taxes during the term of concession. All material and apparatuses necessary for the exploration, working and development of the deposits, and for the construction and development of the pipelines, shall enter Persia free of all taxes and Custom House duties.¹²²

The fact that the Concession operations were exempt from having to pay any tax was extraordinary given the lift to Iran's economy that would have come from an injection of such significant amounts of tax revenue. Moreover, the particularly exploitative nature of Article 7 is evidenced when one considers the relatively low share of the profits from the exploitation assigned to the Persian government.¹²³ It is true that all of the equipment and other resources were ultimately to be handed over to the Persian government. However, given the terms of the Concessions ran for 60 years, it was doubtful that the equipment from the operations would be useful for locating viable oil deposits in the region.¹²⁴ In June 1901, shortly after the granting of the D'Arcy Concession, a Royal Decree from the Shah was to be included as an addition to the Concession which reinforced the view that foreign pressure has again resulted in Persia losing its natural resources for minimal financial return. Article 10 states:

Pursuant to the concession granted to Mr. William Knox D'Arcy, as a result of the particularly friendly relation which unites powerful Great Britain and Persia, it is accorded and guaranteed to the Engineer William D'Arcy, and to all of his heirs and assigns and friends, full power and unlimited liberty for a period of 60 years, to probe, pierce and drill at their will the depths of Persian soil; in consequence of which, all the subsoil products wrought of him without exception will remain the property of D'Arcy. We declare that all the officials of this blessed Kingdom and our heirs and successors will do their best to help and assist the honourable D'Arcy, who enjoys the favour of our splendid court.¹²⁵

The language of the Concession addendum reveals the extent to which the British government had influence over the decision-making processes in Persia in relation to concessions in particular.

¹²² Zoghi (n 117) 63.

¹²³ 16% according to art 10.

¹²⁴ Zoghi (n 117) 63.

¹²⁵ Ibid.

2.4.2 D'Arcy Concession dispute and the Armitage-Smith Agreement

The wording of the Concessions was however disputed during the First World War. Regarding Article 10, for instance, which stated that the Company was required to pay 16 percent of its annual net income to the Persian government,¹²⁶ this was interpreted by the Company to apply to companies operating in Persia only.¹²⁷ The Persian government, however, interpreted this to mean tax was to be paid from the operations of all companies. In addition, the wording of Article 14 was disputed, which referred to the Persian government's requirement to adopt all necessary measures to ensure the safety of the operators and to facilitate all operations associated with the Concession's object.¹²⁸ It was claimed by the Company that the Persian government had not upheld its obligation to protect the pipeline and as a result withheld royalty payments to the government.

On the other hand, the Persian government asserted that according to Article 14, 'it was not liable for loss or damage caused by acts beyond its control'. The government then referred to Article 17 and made a request for arbitration. The two parties eventually signed the Armitage-Smith Agreement in December 1920¹²⁹ which recognised the claim from the Persian government that all companies operated by the Anglo-Persian Oil Company (APOC) were subjected to the profit-sharing provisions in the Concession.¹³⁰ As a result, the Persian government was eligible to collect a royalty payment of 16 percent of the annual net profits made from mining, refining and marketing oil from Persia, irrespective of whether the processes were undertaken by the APOC or by a subsidiary company, either inside or outside of Persia.¹³¹ There was one exception; a royalty payment was not required to be made to the government on the profits made from the transportation of petroleum by ships. In addition, the Armitage-Smith Agreement allowed the APOC to deduct the costs incurred by subsidiary companies for the refinement and distribution of Persian petroleum and its products.¹³² The Agreement also stated that net profits could be adjusted for income tax reasons and the net profit for interest or dividends were not to be subjected to deductions, and all interest and dividends received were not to be included in the profits for which royalties

¹²⁶ As well as an initial fee of several thousand pounds.

¹²⁷ Maximilian Kuhn, Enabling the Iranian Gas Export Options: The Destiny of Iranian Energy Relations in a Tripolar Struggle over Energy Security and Geopolitics (Springer Science & Business Media, 2014) 281. ¹²⁸ Ibid 282.

¹²⁹ This simply clarifies earlier signed documents and is named after the British Treasury official, Sydney Armitage-Smith.

¹³⁰ Kuhn (n 132) 282.

¹³¹ Farshadgohar (n 99) 79.

¹³² Made prior to the calculation of net profits to allocate16% revenue to the Persian government.

were payable.¹³³ Lastly, the Agreement included the provision that all disputes over the payment of royalties should be referred to a Chartered Accountant in England rather than arbitrators from Tehran as stated in in Article 17 of the Concession. The Chartered Accountant was to be nominated by the President of the Institute of Chartered Accountants in England and the decision made by the Chartered Accountant was to be final.¹³⁴ Notably, the claims by the British that there had been violations of the Concession articles were raised after the British government had become the largest shareholder in the D'Arcy Concession. Given the British government had a significant stake in the D'Arcy Concession, the Armitage-Smith Agreement to assign arbitration responsibilities to an accountancy firm in England may be considered as further evidence of the unbalanced nature of the D'Arcy Concession.¹³⁵

The Armitage-Smith Agreement was significantly different to the D'Arcy Concession in its original form, to the disadvantage of Iran.¹³⁶ Furthermore, the intent of the Agreement to link royalty payments to the Persian government along with the profits from trade helped to provide some protection to the Concession if there was minimal or no profit made in some years, or in the event that commercial losses occurred.¹³⁷ On the one hand, this payment method appealed to the Persian government because it needed ongoing revenue from the operations to boost its budget bottom line.¹³⁸ On the other hand the payment method was subject to fluctuations because it was based on the profits drawn by the APOC. As such, it did not ultimately serve the interests of the Persian government. The royalty payment amount, set at 16 percent of the APOC's net profit, reflected the provisions of the 1872 de Reuter Concession, which was higher than the l0 percent of profits initially argued for by D'Arcy.¹³⁹

2.4.3 D'Arcy Concession: Cancellation and the new agreement

When negotiating with APOC, The Iranian government sought an agreement giving Iran the option to own 25 percent of the company.¹⁴⁰ This was in addition to returning 75 percent of the concession region to Iran, demanding that the tax paid by the Company was calculated

¹³³ Stephanie Cronin, 'The Politics of Debt: The Anglo-Persian Oil Company and the Bakhtiyari Khans' (2004) 40(4) *Middle Eastern Studies* 6.

¹³⁴ Alavi (n 110) 44.

¹³⁵ Ibid.

¹³⁶ Kuhn (n 132) 282.

¹³⁷ Ibid.

¹³⁸ Zoghi (n 117) 84-86.

¹³⁹ Gregory Brew, 'In Search of "Equitability": Sir John Cadman, Rezā Shah and the Cancellation of the D'Arcy Concession, 1928–33' (2017) 50(1) *Iranian Studies* 125, 127.

¹⁴⁰ Guilak (n 112) 65.

according to Iran's tax laws and recognising Iran's rights in relation to APOC operations, including after the end of the concession. If APOC agreed to these conditions then the Iranian government would agree to extend the contract for a further 20 years.¹⁴¹ However, due to the protracted nature of the concession negotiations and the ongoing excuses by the company regarding its fall in revenue, Reza Shah decided to have the D'Arcy Concession annulled, a decision enacted by the 10th Session of Parliament.¹⁴² Talks were held between Britain and Iran in both Tehran and Europe until 1933, where they agreed on all terms, excepting the renewal of the D'Arcy Concession, which was due to finish in 1951. The Shah eventually agreed to a 30-year extension requested by APOC.¹⁴³

2.4.4 1933 Concession

After examining its financial relationship with APOC in 1921, the Persian government found a number of irregularities of concern for which the Company was accountable.¹⁴⁴ Then, from 1926 to 1931, APOC used a range of methods to avert changes in the D'Arcy Concession. For instance, in 1930, the Company claimed that it was exempt from having to having to pay tax according to Persian taxation law.¹⁴⁵ The Company also signed a contract with the British Admiralty to supply cheap fuel that the Persian government was unaware of and subsequently did not pay taxes on the deal.

Overall, the D'Arcy Concession made government revenue more sensitive to a fall in APOC income and during the Great Depression it received only small royalty payments due to the decline in APOC's profit as well as to the calculation for 16 percent of APOC's net profits agreed to under the Armitage-Smith Agreement.¹⁴⁶

Although the Persian government cancelled the original concession in 1932, it was willing to negotiate new terms. The British government initially would not accept the cancellation and raised the issue with the Council of the League of Nations.¹⁴⁷ A compromise was reached between the parties in 1933 including terms which saw the Persian government grant APOC the exclusive right, to explore for, extract and refine petroleum within the Concession territory.¹⁴⁸ The new contract also stipulated an annual royalty payment per ton of petroleum

¹⁴³ Ibid.

¹⁴¹ Ibid.

¹⁴² Ibid.

¹⁴⁴ Alavi (n 110) 55.

¹⁴⁵ Ibid.

¹⁴⁶ Fereidun Fesharaki, Development of Iranian Oil Industry (New York, 1976) 13.

¹⁴⁷ Alavi (n 110) 58.

¹⁴⁸ Ibid.

sold,¹⁴⁹ providing the Persian government with a minimum total annual payment¹⁵⁰ and requiring foreign companies to pay 20 percent of the dividends made from the distribution of its products. Nonetheless, Iranian critics of the Concession continued to raise concerns that its conditions were not in the best interests of Iran, especially given the Persian government did not have the power to terminate the Concession and received no compensation for any decrease in the value of the pound, as stipulated in Article 10 (V)(a) of the Concession.¹⁵¹ The rationale underpinning the Persian government's preparedness to agree to the new Concession emerged from its desire to guarantee revenue.¹⁵² A steady increase in oil production, refining capacity, and the payment of royalties followed the new concession agreement, in addition to the opening of new oil fields in Kermanshah province in northwest Iran, and the construction of an oil pipeline to the region.¹⁵³ However, by 1938 the oil production fell into decline, along with government revenue, and the onset of World War Two presented its own set of issues.¹⁵⁴ The next section briefly examines the lead up to the nationalisation of the petroleum industry in Iran.

2.5 World War Two and Politics in Iran

The increasing importance of petroleum to countries was evidenced during World War II, with several attempts made by foreign countries to form an alliance with Iran including Germany, Turkey, Britain and the USSR.¹⁵⁵ None of these attempts were successful. The British and Soviet forces did however occupy oil-producing regions in Iran in 1941 to protect the oil fields from German occupation.¹⁵⁶ In 1941, due to his links to Axis interests, Reza Shah Pahlavi was forced to abdicate his leadership. His son, Muhammad Reza Shah Pahlavi, succeeded his father and adopted a policy of support for the Allies policy and acknowledged the Iranian parliament's pressure for liberal reforms. Towards this end, treaties were signed with the USSR and Britain in 1942 to observe the territorial rights and independence of Iran.¹⁵⁷ The agreement thus sought an end to the occupation of oil producing areas in Iran,

¹⁴⁹ Four shillings every ton of petroleum sold.

¹⁵⁰ Minimum amount is 750,000 pounds.

¹⁵¹ Alavi (n 110) 58.

¹⁵² Mafi (n 119) 411.

¹⁵³ Alexander Melamid, 'Petroleum Product Distribution and the Evolution of Economic Regions in Iran' (1975) 65(4) *Geographical Review* 510, 513.

¹⁵⁴ Mafi (n 119) 412.

¹⁵⁵ Guilak (n 112) 183.

¹⁵⁶ Ibid.

¹⁵⁷ Ibid 129.

and while, unlike the US, both the USSR and Britain were opposed to withdrawing from the regions, a negotiated agreement for their withdraw from Iran was achieved in March 1946.

2.5.1 Soviet Union influence on oil concessions

Iran emerged as a region of competing interests among powerful nations at the start of the Cold War due in part to the global increase in oil consumption and the importance of oil to the growth of national economies.¹⁵⁸ The US adopted a particularly active role in relation to Middle Eastern oil.¹⁵⁹ Given the role that petroleum was to play in the post-war reconstruction period it was not particularly surprising that Iran presented as a country for oil concession hunting. Concessions sought by the Soviets aimed to provide a counter balance to the US requests for petroleum concessions in northern Iran. In return for the Concession, the Soviet Union agreed to protect the security of northern Iran and to encourage a stable relationship between the great powers.¹⁶⁰

In April 1946, the Prime Minister of Iran, Ahmad Qavam signed an agreement to establish the Soviet-Iranian Oil Company to exploit Iran's oil reserves in the north. This concession was valid for 25 years and included several new features including a 51% (USSR) 49% (Iran) split of ownership, changing to a 50/50 split after a further 25 years; the profits were apportioned according to the number of shares held, with no provision for royalty payments.¹⁶¹ The British government kept a keen eye on the proceedings, concerned that the forces of nationalism in Iran opposing the Soviet Concession might also challenge the Anglo-Iranian Concession in Iran's south. The US, at least in its rhetoric, defended Iran's right to decide for itself how to dispose of its natural resources.¹⁶²

At this time, the number of Iranian parliamentarians opposed to the granting of any concessions grew daily. This led to their initial rejection of the concession deal with the Soviet Union following lengthy parliamentary negotiations.¹⁶³ A Bill was then put before parliament by Dr Rezezade Shafag, who was later to sit on the Majlis Oil Commission of 1950-51, containing the following:

- 1. The oil agreement with the USSR was to become null and void.
- 2. Iran uses its own capital and resources to implement a five-year plan to explore and exploit its oil resources.

¹⁵⁸ Oil supported nearly 4% of world's energy in 1900 and nearly 18% in by 1940.

¹⁵⁹ Richard John Eden, Michael Vivian Posner, Richard Bending, Edmund Crouch and Joe Stanislaw, *Energy Economics: Growth, Resources and Policies* (CUP Archive, 1981) 77.

¹⁶⁰ Catherine Skrine, *The World War in Iran* (Constable, 1962) 227.

¹⁶¹ Ibid.

 ¹⁶² Lawrence Paul Elwell-Sutton, *Persian Oil: A Study in Power Politics* (Lawrence & Wishard, 1955), 117.
 ¹⁶³ Ibid 118.

- 3. If oil is discovered, the Iranian government can arrange for its sale to the USSR following informed approval from the Majlis.
- 4. Iran no longer agrees to grant concessions to foreign entities or to participate in partnerships with foreign oil companies.
- 5. In cases where Iran's territory and resource rights are breached, particularly in relation to oil exploitation in the south, the government should enter into negotiations to restore its national rights, while informing the Majlis of the outcome.¹⁶⁴

The Bill, which was ratified in October 1947, indicated Iran's determination to have control over its natural resources. As such, it had significant implications for the sustainability of the AIOC and the oil exploitation operations in the south. The impetus of the Bill was national pride in Iran and the people's desire to stand up for their nation's sovereign rights.¹⁶⁵

2.6 Overview of Pre-Nationalisation Era

Within Iran, these political interactions created the perception that Iran was being manipulated and exploited by the world's powerful nations. This left a lasting mark in the minds of Iran's politicians and citizens alike, leading to the cautious and untrusting attitude the Iranians have towards foreign intervention today. The inadequacy of the Anglo-Iranian Oil Company (AIOC) in terms of the royalties paid to Iran became an increasing point of contention. Although the AIOC had provided an access point to advanced technical knowledge and the company returns were as high as 150% in some years, there was a developing sense among Iranians that they should have greater control and draw a greater profit from their natural resources.

As a result, the time had come for many Iranians to take action to nationalise the nation's petroleum industry and to recover their natural resources. The elections for the 14th, 15th and 16th Majlis saw important changes in both the domestic politics in Iran and the international economy, which, under the hegemony of the US, now had access to relatively inexpensive petroleum. Therefore, following World War Two the revenue from the sale of petroleum provided a platform for the Iranian government to introduce an ambitious national development plan and to intervene more actively in the economy. This, combined with the

¹⁶⁴ Ibid.

¹⁶⁵ Ibid 119.

past objections of Iranians to foreign participation in the Iranian petroleum industry, saw the introduction of the Buy Back contract (BBC). Thus, the nationalisation of the petroleum industry in Iran in 1951 emerged from a period of what Iranians saw as manipulative and unfair treatment from foreign powers. In this sense, the Nationalisation initiative was both politically and economically motivated.

2.6.1 Concession issues and their renegotiation

The Iranian government passed the First Development Plan in 1949, to be funded largely by revenue from the petroleum industry. As such, alterations to the agreement with the AIOC were proposed given that the British government made more profit from taxing the AIOC than the government of Iran made from the receipt of royalty payments.

In May 1949, an agreement was reached in which the new provisions were inserted into the clauses of the Concession:

- Clause 3 (a) royalty payments to the Iran government were to increase by two shillings per ton of exported or sold oil.
- Clause 7 (a) the AIOC agreed to increase its tax payments from 9 pence to 1 shilling per ton after the first 600,000 tons of production.
- Clause 4 (a) the AIOC agreed to pay the Iranian government, as ratified by the Majlis, £5,090,909 from the company's general reserves. Additionally, the Government was to receive 20 percent of the annual amount in the AIOC general reserves to offset the impact of the British income tax.¹⁶⁶

Clause 4 (b) – saw the guaranteed minimum annual payment increase by £4,000,000. These new provisions appeared to move the contract between the British and Iranian governments towards a fairer and more balanced footing, with greater focus also placed on engaging the Iranian workforce. However, the increasingly nationalistic public sentiment in Iran meant that any remaining bias in the Concession towards the foreign entity could not be ignored.¹⁶⁷

2.6.2 Rejecting the compromise agreement

The Mossadegh Committee appointed by the new Majlis examined the new Agreement and concluded that it was still not meeting Iranian interests. The sense of exploitation associated with the AIOC deal was compounded by the deal negotiated between Saudi Arabia and the

¹⁶⁶ Zoghi (n 117) 251–256.

¹⁶⁷ Zoghi (n 117) 256.

Aramco Oil Company.¹⁶⁸ In response, the AIOC agreed to a 50-50 profit sharing deal in addition to a £5 million future royalty payment and a monthly advance of £2 million up to the conclusion of 1951. The Mossadegh Committee passed its decision to nationalise the petroleum industry in Iran and the AIOC proposal was subsequently rejected.¹⁶⁹ The Bill supporting the nationalisation of the petroleum industry was enacted in 1951 and included the article:

For the happiness and prosperity of the Iranian nation and for the purpose of securing world peace, it is hereby resolved that the oil industry through all parts of the country, without exception, be nationalized; that is, all exploration, extraction and exploitation operations shall be carried out by the government.¹⁷⁰

The British government responded quickly and broadly, withdrawing all technicians from Iran, declaring an embargo on Iranian petroleum, freezing Iranian assets and prohibiting exports to the nation. This resulted in a dramatic reduction in oil production in Iran and the protracted hostilities between the AIOC and the Iranian government caused the Iranian economy to suffer. In turn, the subsequent political upheaval saw Parliament dissolved in 1953 while the petroleum industry in Iran experienced a significant downturn.¹⁷¹

2.7 Nationalisation of the Oil Industry and its Consequences

The law to nationalise the Iranian petroleum industry included several articles which are worthy of attention. For instance, Article 2 formed the legal parameters of the nationalisation process. In turn, to implement nationalisation the AIOC was forced to dispose of its assets and to give control to the Mixed Board of senators and deputies which was established by the Majlis Oil Committee (MOC).¹⁷² In addition, Article 6 formed the basis for the gradual transition from a reliance on foreign expertise to the use of Iranian experts to coordinate the operations. Lastly, Article 7 formed the basis for the sale of oil, which was to continue at the sales volume undertaken by the AIOC.¹⁷³

In response to Britain's concerns about the contractual obligations that Iran had previously agreed to, the newly-appointed Board of the National Iranian Oil Company (NIOC) was

¹⁶⁸ The profits were split 50/50 with the Saudi Arabian government.

¹⁶⁹ Zoghi (n 117) 256.

¹⁷⁰ Ibid.

¹⁷¹ Mohammad G Majd, 'The Oil Boom and Agricultural Development: A Reconsideration of Agricultural Policy in Iran' (1989) 15(1) *The Journal of Energy and Development* 125, 127.

 ¹⁷² Neveen Abdelrehim, Josephine Maltby and Steven Toms, 'Corporate Social Responsibility and Corporate Control: The Anglo-Iranian Oil Company, 1933–1951' (2011) 12(4) *Enterprise & Society* 824, 854.
 ¹⁷³ Ibid.

instructed to comply with the existing practices of the AIOC including the continuance of employment of AIOC employees.¹⁷⁴ The period from 1951 to 1954 proved to be highly unsettled in terms of Iran's capacity to strengthen its nationalised oil sector. The ongoing challenge to Iran's decision to nationalise by the British government in the International Court of Justice (ICJ) resulted in Iran's refusal to accept an interim measures imposed by the Court while trying to resolve the dispute.¹⁷⁵ The attempt by the US government to resolve the international dispute via the establishment of the Harriman Commission to facilitate negotiations between Iran and Britain also failed.¹⁷⁶

In 1952, the ICJ reached a decision regarding its competence to rule on the Iran and Britain dispute. The decision by the Court was that it did not have jurisdiction to rule on the matter brought by the British Government, and the Court thus called for all interim measures to cease.¹⁷⁷ The subsequent rejection of proposals and counter proposals by both Britain and Iran on how to resolve the dispute, and the ultimate inability of the US government to effectively mediate the dispute, saw Iran strongly defend its position that any outcome must satisfy the requirements of the new nationalisation law. The Iranian government's decision in 1952 to close down the British Embassy was in response to the covert attempts it believed were being initiated by the British government to replace the Mossadegh government with a government friendlier to its oil interests.¹⁷⁸

2.7.1 The 1953 Coup D' Etat and its implications for foreign investment

With the support of the US, the British government planned a coup d'état to overthrow of the Mossadegh government.¹⁷⁹ On the back of support from some conservative groups in Iran including some mullahs, senior members of the police force, army officers, and members of the media, the Shah and other members of the Royal family also supported the possibility of a coup, with Mossadegh to be replaced with General Zahedi.¹⁸⁰

The following months saw the home of Mossadegh attacked, and key officials supportive of Mossadegh's rule kidnapped in a bid to destabilise Iran. However, the plan to destabilise Iran was uncovered and the key players in the Iranian government were arrested.¹⁸¹ By June 1953

¹⁷⁴ Kuhn (n 132) 286.

¹⁷⁵ Ibid.

¹⁷⁶ Mohsen S Haery, *The Legal Implication of the Dispute Over the Iranian Nationalization of the Oil Industry* (University of California, Berkeley, 1955) 108.

¹⁷⁷ Ibid.

¹⁷⁸ Ibid.

¹⁷⁹ Ibid.

¹⁸⁰ Homa Katouzian, Musaddiq and the Struggle for Power in Iran (I.B. Tauris & Co, 1990) 178.

¹⁸¹ Ibid 182.

however, the US had given its backing for the coup to take place in August. Much of the media was now controlled and giving its support to the Shah.¹⁸² Mossadegh had managed to thwart the first attempt at the coup, and the Shah and his family had been forced to flee Iran to Baghdad and eventually to Rome. However, General Zahedi successfully seized control of the Government and Mossadegh was arrested just days after an attempted escape. Mossadegh was imprisoned for three years and then placed under house arrest where he remained until his death in 1967.¹⁸³ Within a year of the successful coup, the Zahedi government had negotiated the 1954 Agreement with a consortium of international oil companies bringing an end to the nationalistic stance adopted by Mossadegh. The coup d'état in 1953 was thus a pivotal element in the efforts of the British to protect their investment interests in Iranian petroleum which were being threatened by the nationalisation of the petroleum industry.¹⁸⁴ Hence, the strict terms and conditions that the Iranian government insisted upon when negotiating petroleum agreements with foreign oil companies is understandable. The West played a role in the overthrow of the nation's first democratically elected government and this, combined with the subsequent capitulation of the Shah to foreign interests, arguably laid the foundation for the 1979 Revolution. This points to the importance of the association between the oil agreement made between Iran and foreign countries and the issues of national security and economic prosperity. During the first half of the twentieth century there emerged serious discord in Iran related to the terms and conditions of oil agreements with foreign entities. Of most concern was the question of how best to ensure the economic benefits of oil agreements were returned to the Iranian economy.¹⁸⁵

2.7.2 The 1954 Consortium

As such, 1953 saw the emergence of a new era in the relationship between Iran and foreign oil companies.¹⁸⁶ The new government was willing to negotiate with the AIOC to settle the oil dispute as well as to secure an increase in petroleum exports.¹⁸⁷ Moreover, the foreign oil companies indicated their willingness to enter into further contracts to explore, refine, and market Iranian petroleum. However, it was also recognised that the AIOC could no longer have exclusive rights to the production and sale of Iranian petroleum and that all future

¹⁸² Ibid 188.

¹⁸³ Ibid.

¹⁸⁴ Ibid.

¹⁸⁵ Kirsten Bindemann, 'Production-Sharing Agreements: An Economic Analysis' (Oxford Institute for Energy Studies, 1999) 5.

¹⁸⁶ Heidari (n 14) 1758.

¹⁸⁷ The willingness to conclude contracts emerged from the needs of budget-based large-scale development program.

contracts would have to be 'within the framework of the nationalisation law, even if only formally'.¹⁸⁸

2.7.3 The National Iranian Oil Company and the International Consortium

As relations between Iran and the US returned to a more normalised state, Washington agreed to provide the new government in Tehran with financial and military support and to assist it to resolve its petroleum disputes. By December 1953, the President of the AIOC met with several foreign oil companies from the US, London, France, Holland to determine how best to bring Iranian petroleum to the international market, with the US arguing that 'a consortium was the best possible answer to the problem'.¹⁸⁹ However, the US push to establish a provisional consortium did not have the support of the British government or the AIOC. The US adopted the position that the AIOC should have 40 percent share of the consortium, US companies should combine to also have a 40 percent share, with the French and the Dutch oil companies having the remaining 20 percent.¹⁹⁰

The US at this time was clearly trying to increase the participation of its national oil companies in the production and sale of Iranian petroleum. This is evidenced on 30 January 1954 when the five major oil companies in the US involved in the attempt to settle the Iranian oil dispute were granted antitrust immunity by the US Department of Justice.¹⁹¹ This meant that they were now free to join with the British, Dutch and French petroleum companies to form a consortium to explore for, refine and sell Iranian petroleum.¹⁹² This move was supported by Prime Minister Zahedi in Iran with the creation of a special petroleum commission to coordinate the establishment of a consortium of primarily US, British, and Iranian oil companies to develop Iranian oil.¹⁹³ In August 1954 a consortium agreement was signed and then confirmed by the Majlis in the following October. The consortium composition included the British Petroleum Company (previously the AIOC) 40 percent, five US oil companies (Standard Oil, New Jersey; Standard Oil, California; Mobil, Texas Oil and Gulf Oil) combining to hold 40 percent, Royal-Dutch-Shell 14 percent, and Compagnie Francaise des Petrols 6 percent.¹⁹⁴ Within nine days of the agreement being ratified, Iranian

¹⁸⁸ Alavi (n 110) 116.

¹⁸⁹ Ibid 128.

¹⁹⁰ Ibid.

 ¹⁹¹ 'Anti-trust immunity' allowed companies to carry out typically prohibited competitive tactics including dominating a market and determining all prices, leading consumers to have limited choice.
 ¹⁹² Alavi (n 110) 129.

¹⁹³ Ibid.

¹⁹⁴ Ibid.

petroleum began to flow into US, British, Dutch and French tankers for sale in the world market.

There were several significant provisions in the agreement worthy of consideration:

- The Consortium was to pay the NIOC for all equipment and to train personnel to facilitate the production, export and sale of the crude products.
- Iran would purchase oil for its own consumption at around the cost of production.
- The agreement was for 25 years, with a provision in place for a maximum of three fiveyear extensions.
- After the initial three-month period, the income to Iran was to increment over the following three years from 31 million in the first year, 62 million in the second year, and 67 million in the third year.
- The Consortium was to guarantee the export of a minimum 78 million tons of oil over first three years.
- The Consortium agreed to pay Iran income tax, in Sterling, at approximately half its net operating revenue.¹⁹⁵

The Consortium members were subsequently granted by Iran and the NIOC the sole rights to explore and produce petroleum in southern Iran and to operate a refinery at Abadan according to the terms in the Agreement. Although the NIOC, with the Iranian Government as its major shareholder, was owner of all fixed assets in the nation's petroleum industry, the foreign oil companies were granted unrestricted use for the duration of the agreement.¹⁹⁶ The agreement also specified that all petroleum products required for consumption in Iran would be delivered to the NIOC via the appropriate foreign operating company. As such, it was regarded by the Iranian parliament as a way to secure greater participation in the management of the nation's natural resources and to earn higher oil revenues while placing the greater financial burdens and risks onto the foreign oil companies.¹⁹⁷

2.7.4 Reforming the petroleum agreements

By 1957, three years after the signing of the Consortium Agreement, the first Iranian Oil Bill was passed by the Majlis allowing for further contracts to be signed with foreign investors for the exploration and production of petroleum outside of the operating area occupied by the

¹⁹⁵ Mostafa Elm, *Oil, Power, and Principle: Iran's Oil Nationalization and Its Aftermath* (Syracuse University Press, 1994) 310.

¹⁹⁶ The Iranian Oil Refining Company and the Iranian Oil Exploration and Producing Company were involved.
¹⁹⁷ Mafi (n 119) 410.

consortium.¹⁹⁸ The Consortium Agreement, along with later miscellaneous contracts, were made in accordance with the 1957 Petroleum Law and were defined by 75 percent / 25 percent profit sharing agreements. That is, the NIOC as partner received half of the profits of the exploitation and the Iranian Government would receive 50 percent of the profit as income tax.¹⁹⁹ In some later agreements, the amount of tax payable to the Iranian government increased to 85 percent as changes to tax laws were implemented, along with mandatory royalty payments to the NIOC.²⁰⁰ Additional agreements were signed by the NIOC and Italian oil company Azinde Generate Italiana Petrole (AGIP) in August 1957; the Pan American Petroleum Corporation (a subsidiary of Standard Oil of India) in June 1958 – forming the Iranian Pan-American Oil Company (IPAC), and the Sapphire Petroleum Company of Canada in June 1958.²⁰¹ Notably, these joint agreements had different terms and conditions to the 1954 Consortium Agreement and indeed to those generally agreed to in Middle Eastern oil concessions.

2.8 Contractual and Commercial Changes between 1960 and 1979

The petroleum industry in Iran underwent many changes throughout the 1960s and 1970s due to initiatives led by the Shah to modernise the sector and to increase foreign investment. The sustained development of the economy in Iran was regarded as fundamentally dependent on the rapid modernisation of the sector and its infrastructures as discussed below.²⁰²

2.8.1 Petroleum transactions and the Iranian economic development plans

Substantial increases in oil revenues coincided with the campaign to modernise Iran under the reign of Mohammad Reza Shah. This included the adoption of Western industrial policies by the Shah during the 1960s to achieve three main outcomes: growth of the industrial sector, the transition to modern manufacturing methods²⁰³ and better wages management.²⁰⁴ The subsequent reforms and development plans, sometimes referred to as the "White Revolution", included US\$1.5 billion in investment into the agriculture sector and the allocation of profits from the petroleum sector to build industrial infrastructure. The most substantial progress was

¹⁹⁸ Alavi (n 110) 142.

¹⁹⁹ Ibid.

²⁰⁰ Ibid.

²⁰¹ Ibid 144.

²⁰² Alavi (n 110) 145–148.

²⁰³ The significant funds and technology required could be gained by entering into oil contracts with more technologically advanced foreign companies.

made in the petroleum, transport and communication industries; for example, the building of a refinery²⁰⁵ and an petroleum pipeline into the USSR.²⁰⁶ However, due to a higher priority being given to the industrial sector over the agricultural sector, there emerged a widening in the capacity of wealth accumulation from the industrial sector compared to farming and crop-growing.²⁰⁷

Indeed, the underlying assumption of the modernisation agenda was that industrialists earning robust returns on investments would reinvest in the Iranian economy and thus revitalise it via further development.²⁰⁸ The reality however was quite the opposite, with much of the profits gained by industrialists contributing to a widening of the gap between the middle and industrial classes and the build-up of social discontent. This was compounded by the jump in the petroleum price from 1973 to 1977, increasing inflationary pressure, corruption among officials, and high levels of migration to urban centres.²⁰⁹ This highlights the role that petroleum revenues can play in Iran in both national progress and development and national instability and conflict.

2.8.2 The 1965 agreements

In March 1964, a report commissioned by the Iranian government presented it findings from a survey of almost 50,000 square kilometres of the Persian Gulf for the presence of oil. The findings were of a significant amount of oil in the Gulf, prompting the NIOC to invite all interested oil companies to submit to the Iranian government a business plan for the oil extraction and refinement.²¹⁰ As a consequence, by the beginning of 1965, a number of JVAs were signed by the NIOC with several oil companies. The contracts were based on the 50 percent partnership arrangement with the NIOC as detailed above.²¹¹

2.9 The Islamic Revolution and its Impact on Petroleum Contracts

The advent of the Islamic Revolution in Iran in 1979 had a significant impact on the agreements. Following the Revolution, the newly installed Constitution put an end to the Oil Concession Agreements with Iran, with Article 81 stating: "It is absolutely forbidden to give foreigners the right to establish companies or institutions in commercial, industrial, and

²⁰⁹ Ibid.

²⁰⁵ A concerted effort was undertaken to reduce Iran's dependence on foreign technology by developing their own facilities.

²⁰⁶ Elm (n 200) 42.

²⁰⁷ Ibid 40.

²⁰⁸ Ibid.

²¹⁰ Haleh Afshar, Iran: A Revolution in Turmoil (Springer, 1985) 103–104.

²¹¹ Ibid.

agricultural fields, as well as in mines and in the service sector".²¹² The NIOC annulled all contractual agreements with foreign oil companies signed prior to the Revolution.²¹³ The NIOC was acting on the decision by the Revolutionary Council which argued the importance of revising all business agreements with foreign entities. The annulment of the contracts had both immediate and wide-reaching effects to Iran's economy, including the costly and protracted legal action taken against the Iranian government by the foreign oil companies.²¹⁴ Although Iran offered to compensate the consortium members for damages resulting from the cancellation, the aim of the legal action was to achieve a legal verdict against Iran rather than just a payment for damages. Moreover, the withdrawal of foreign oil companies from their operations meant that Iranian officials were forced to manage the fields and try to curb disruptions to the flow of oil.²¹⁵

2.10 Iran-Iraq wars

Further disruption to the petroleum sector in Iran emerged as a result of the 1980 Iran-Iraqi war. Oil and gas infrastructures were targeted by air and on the ground with the attacks designed to interrupt both the production and export of petroleum.²¹⁶ The Iraqi armed forces enjoyed some success in this campaign, destroying and/or damaging crucial petroleum production, refinery and exporting sites. By 1986 the impact of these disruptions was being acutely felt, with the average barrels of production each day falling from 2.3 million barrels between 1982 to 1986 to almost half that in 1987.²¹⁷ Iran's economic challenges were only made worse by the financial sanctions against Iranian petroleum imposed by the US, Japan, US and France.²¹⁸ The damage inflicted on Iran's petroleum sector as a result of the war with Iraq prompted the need for foreign involvement and investment in the reconstruction of the sector. As such, this established the tone and the parameters for contractual agreements with foreign entity oil companies into the 1990s.²¹⁹

²¹² Constitution of Iran, art 81 <<u>https://www.wipo.int/edocs/lexdocs/laws/en/ir/ir001en.pdf</u>> 18.

²¹³ Paasha Mahdavi, 'Oil, Monarchy, Revolution and Theocracy: A Study on the National Iranian Oil Company', in David G. Victor, David R. Hults and Mark C. Thurber (eds), *Oil and Governance: State-Owned Enterprises and the World Energy Supply* (Cambridge University Press, 2012) 243. ²¹⁴ Ibid.

²¹⁵ Ibid.

²¹⁶ Ibid 51–52.

²¹⁷ James D Hamilton, 'Historical Oil Shocks' (Working Paper No. w16790 National Bureau of Economic Research, 2011) 17.

 ²¹⁸ BBC News, 'Six Charts That Show How Hard US Sanctions Have Hit Iran' *BBC* (2019)
 https://www.bbc.com/news/world-middle-east-48119109 (access 12 January 2020)
 ²¹⁹ Valentina Svalova, *Risk Assessment* (BoD – Books on Demand, 2018) 52–53.

2.11 Conclusion

Examining the historical context of Iran's petroleum industry is vital to gaining a deep understanding of the conditions and functions of modern Iran Petroleum Contract (IPC). The reasons why Iran approaches contractual agreements with foreign entities with caution and some suspicion are readily discernible. History shows that instances of foreign domination, contractual terms and conditions unfavourable to Iran's interests, and growing nationalistic sentiment towards the protection of natural resources against foreign exploitation are key factors shaping the modern petroleum industry in Iran. The petroleum sector in Iran and the sale of exploration rights and revenue gained from the sale of petroleum is critical to an economic reform agenda. In turn, this implies the need for the Iranian government to engage in commercial cooperation with foreign investors to ensure a robust and profitable petroleum sector. The original Buy Back system introduced in the 1990s and the more recent transformation to the IPC demonstrate the need of the Iranian government to effectively balance how it accesses foreign expertise, technology and investment with its internal national priority to maintain complete control over its petroleum resources.

3 REGULATORY FRAMEWORKS

3.1 Introduction

The domestic laws of a country provide the framework for regulating and monitoring the contract terms and conditions for foreign investment, exploration, production and payment of taxes / royalties. This chapter examines the Oil Law and legislation in Iran following the victory of the Islamic Revolution. It provides a detailed discussion on such aspects as the non-voids bale article of approved petroleum contracts of 1940 by the Council of the Islamic Revolution, the complimentary law establishing the Ministry of Petroleum of 1941, the Oil Law approved in 1918, Annual Budget Laws and rules linked to the five-year plan of economic development, laws supporting and encouraging foreign investment of 1962, the Oil Law amendment of 1918 approved in 1971, the duties of the Ministry of Petroleum, and also common conditions, structure and pattern of upstream approved oil and gas in 1975.

3.1.1 Single article non-voiding the contracts

The Islamic Revolution victory in 1989 changed the rules and regulations governing Iran's petroleum industry. The Revolutionary Council of the Islamic Republic of Iran in August 1940 passed a bill to establish the Ministry of Petroleum for the first time in history. According to Article 1 of the bill, the NIOC, National Petrochemical Company and National

Iranian Gas Company fall under the authority of the Ministry of Petroleum and its affiliates.²²⁰

It was also at this time that the Guardian Council of the Constitution was first convened. The Council is a 12-member group mandated under the Constitution of Iran to, among other things, ensure the compatibility of the legislation passed by the Islamic Consultative Assembly [i.e. Majlis] with the criteria of Islam and the Constitution.²²¹ As stated in Article 98 of the Constitution: 'The interpretation of the constitution is the responsibility of the Guardian Council. This is determined with the approval of three-fourths of its members.'. As such, the economic relations that Iran has with foreign countries may be the subject of constitutional interpretation by the Council, which has the authority to veto laws passed by the Majlis.²²² In addition, the Council supports the influence of the IRGC on the economic and cultural life in Iran.²²³ The IRGC was founded by Ayatollah Ruhollah Khomeini after the revolution in Iran as a branch of the Iranian Armed Forces. However, while the Iranian Army is charged with defending Iranian borders and maintaining domestic order according to the constitution, the IRGC is charged with the role of protecting Iran's Islamic republic political system. This includes providing protection the Islamic political system against interference from foreign countries. Under the President Mahmoud Ahmadinejad's administration, the IRGC has expanded its role in the social, political, military and economic affairs of Iran, leading some analysts in the West to claim that its political power exceeds even that of the Shia clerical system in Iran.²²⁴

The first year after the Islamic Revolution also saw major changes in the rights of Iranian petroleum in relation to existing contracts. The Revolutionary Council of Islamic Republic of Iran passed a bill for the formation of a special commission on the petroleum contracts to identify all contracts which were inconsistent with the laws underpinning the nationalisation of Iran's petroleum industry.²²⁵ They were to be placed in the hands of a special commission for the Minister of Petroleum to determine if they were void. The Commission also had the authority to settle disputes arising from such contracts.²²⁶

²²⁰ Daniel, Brumberg and Ariel I. Ahram. 'The National Iranian Oil Company in Iranian Politics' (Paper Prepared in Conjunction with an Energy Study Sponsored by Japan Petroleum Energy Center and the James A. Baker III Institute for Public Policy, Rice University, 2007) 2.

²²¹ Constitution of Iran (n 217) art 94 21.

²²² Yasmin Alem, 'Duality by Design: The Iranian Electoral System' (IFES, 2011) 21.

²²³ Ibid 18–20.

²²⁴ Ibid.

 ²²⁵ Reza Yeganehshakib, Iran's New Generation of Oil and Gas Contracts: Historical Mistrust and the Need for Foreign Investment' (2015) 3(4) *The Journal of Political Risk* 5..
 ²²⁶ Ibid.

3.1.2 Bill to establish the Ministry of Petroleum, 1941

The complementary law to establish the Ministry of Petroleum was approved in March 1941. The aim in establishing the Ministry of Petroleum was to apply national ownership principles to Iran's gas and oil reservoirs; on the oil, gas, and petrochemical installations; and on those related to exploitation and marketing (paragraph 1 of the Act).²²⁷

The basic tasks of the Ministry of Petroleum include:

- Proper and correct preservation of resources and exploitation of oil and gas resources of the country.
- Maintain oil and gas facilities and the creation of the petrochemical and other associated facilities.
- Efforts to develop technology and relevant industrial knowledge to become more self-sufficient and to be independent of foreigners.
- Supervision of the export of oil and petroleum products; gas petrochemicals and related products, and final approval of contracts and other contracts relating to the above matters.
- Coordinate the activities of different units of oil, gas and petrochemical industries and coordinate with government agencies on the country's energy policy.
- Determine policies and common rules for governing subsidiaries and affiliates.
- Implement and integrate operational plans, financial and commercial subsidiaries, and formulate a comprehensive program within the framework of national programs and the government's economic policies and the executive government.
- Plan major capital projects and the supply of necessary funds.
- Develop existing activities or create new activities.
- Exercise technical, administrative, legal, commercial, financial and administrative supervision and control over the performance of subsidiaries.
- Ensure general subsidiary companies observe the approved principals and follow general policies.
- Monitor the implementation of regulations and general policy assessment and classifications.

²²⁷ Samad N Ebrahmi and Abdul, Shiroui Khouzani, 'The Contractual Form of Iran's Buy Back Contracts in Comparison with Production Sharing and Service Contracts' (Middle East Oil Show, Society of Petroleum Engineers, 2003) 150.

- Supervise international relations and monitor affairs in the Ministry of Petroleum outside the country.
- Establish and maintain good relations with the legislature and the civil and military authorities.
- Represent Iran in the international oil community.
- Represent the petroleum industry in the Supreme Council for Energy and Economic Council.
- Investigate the conditions necessary for the creation, dissolution, disintegration or integration of subsidiaries and affiliates and legal authorisation from the competent authorities.²²⁸

The Ministry of Petroleum was given the authority to change the form of existing companies according to the operation necessary for change (analysis, integration, cancelation, or establishment of new companies).²²⁹ In addition, it could undertake the necessary changes after approval by the Committee composed of the Ministry of Petroleum, a supervisor for planning and budget, and the General Secretary for Administrative and Employment Affairs (Article 2).²³⁰

The statute for the companies listed in Article 2 was provided by Ministry of Petroleum and the changes commenced following approval of the Board and the Ministers (Article 3).²³¹ However, Article 4 stated that the validity of existing statutes related to the NIOC and national petrochemical industry to approve companies' statutes referenced in Article 3 remains in force.²³² Before approval of new statutes (Article 3), the option to select members to be on the Board of Directors of subsidiary companies was delegated to the Ministry of Petroleum (Article 5).²³³ Lastly, this rule stated that to meet the human resource needs the Ministry of Petroleum has the authority to hire workers from subsidiary companies as officers who administer the employment rules (Article 6). They were also to cover all staffing costs from the budget of the subsidiaries and affiliates that hired the workers.²³⁴

²²⁸ Ibid.

²²⁹ Ibid.

²³⁰ Law Amendment Act art 2.

²³¹ Law Amendment Act art 3.

²³² Law Amendment Act art 4.

²³³ Law Amendment Act art 1 para b established the Ministry of Petroleum on 29.03.1359

3.1.3 Petroleum Law 1947

In addition to the provisions of the Constitution after the revolution, in 1947 another Oil Law was ratified by the Iranian parliament. The provision and approval of this rule was likely influenced by the political circumstances and a xenophobic mood to avoid foreign influence in the petroleum industry.²³⁵ In general, the Petroleum Law designates petroleum contracts as private and the NIOC accepts mitigation in accordance with the terms and conditions of the contract. However, the principle that a nation has sovereignty over their natural resources in international law ensures that nations have the right to govern the development of their natural resource while compensating the IOC via the profits they incur.²³⁶

However, the law does not provide clear regulations about petroleum contract patterns and the mixing of investment and oil installations leads to confusion about how to manage the legal status of such investments.²³⁷ Article 2 of the Oil Law states the country's oil resources are a part of public wealth and according to Article 45 of the Constitution are owned by the Islamic government.²³⁸ All installations, equipment, properties and investments by the Ministry of Petroleum and subsidiaries belong to Iran and are the possession of Islamic government. Acts of sovereignty and ownership of the petroleum interests belong to Islamic government. Based on the regulations and powers prescribed in this Act, it is the responsibility of the Ministry of Petroleum to act according to principles and general plan of the country.²³⁹ This rule repeats Article 45 of Constitution and when comparing the two rules of 1935 and 1947 it is evident that the new rule just addresses the generalities and principles. Thus, when compared to the old rules, the new rules have many shortcomings. According to Article 12 of the Oil Law 1947, the enactment of the Act means any rules against this rule are annulled.²⁴⁰ It is difficult to say that the rule of 1935 maintained its validity and power. Furthermore, due to Article 6 of the oil rule of 1947, whereby foreign investment in petroleum operation was banned, the legal barriers related to partnership contracts in production continued. Article 6 states that: 'All capital on the basis of budget operation units the Ministry of Petroleum proposed and after approval from the General Assembly will be included in the national budget. Foreign investment in these operations will

²³⁵ Mohsen Mohebi, *Legal Issues Facing the Oil and Gas in the Light of International Arbitration* (Publication of Knowledge, 1967) 16, 20.

²³⁶ Ibid 49.

²³⁷ Ibid.

²³⁸ Law Amendment Act art 2.

²³⁹ Mohebi (n 240) 20.

²⁴⁰ Law Amendment Act art 12.

not be allowed at all'.²⁴¹After the Islamic Revolution, the selection and use of "reciprocal" type contracts was significantly more common in the petroleum industry (see Chapter 4, section 4.4). This was because of the constraints and requirements of the Constitution and other laws, and the economic sanctions and requirements.²⁴² Plans to use reciprocal transactions in Iran's petroleum industry go back to the Oil Law, 1935 and the years prior the Islamic Revolution. The legislation imposed significant limitations to the participation of foreign oil companies in oil and gas upstream activities. Article 3 of the Oil Law stipulates that Iran's oil resources and the exploration, development, production and distribution of petroleum throughout the country and continental shelf is exclusively the responsibility of the NIOC directly or through its agents and contractors.²⁴³ In Iran's Reciprocal Transaction Contracts, foreign investment companies undertake all investment funds such as installation of all equipment, start-up, and technology transfer. After setting up the equipment and the production to contract stipulations, return of capital, and costs and benefits, is provided by instalments within a certain time (e.g. seven years). This is in the form of delivery of products produced from these investments and facilities and no other form of reciprocal trading exists.²⁴⁴ Article 8, Section 3 states the Executive Regulations for the approval of Reciprocal Transaction Contracts of April, 1965 by the state board: 'If the exploration operation does not lead to the discovery of trading fields in mentioned area, the contract will be terminated and the parties have no right to demand any money'.²⁴⁵

3.2 Additional Regulations following the Islamic Revolution

In the First Five-Year Development Plan approved in 1989, it reads: 'The government can meet part of the needs of the mining industry in matters of production, exports and related investments, and mutual transactions by up to ten billion dollars'.²⁴⁶ In the Second Five-Year Development Plan, such permission has been granted, and in paragraph 'a' of this clause it states:

The relevant executive agencies will be allowed within the validity of the law ... up to six billion and five hundred million dollars using Buy Back Contract methods ... and

²⁴¹ Law Amendment Act art 6.

²⁴² Chow et al (n 22) 4.

²⁴³ Law Amendment Act art 3.

²⁴⁴ Tamara Milenkovic-Kerkovic, 'Drafting an International Buy Back Agreement' (2004) 2 *Economic and Organization* 165, 167.

²⁴⁵ Ebrahimi and Khouzani, (n 232) 8.

²⁴⁶ Law of the First Five-Year Economic, Social and Cultural Development Plan 1989, para (a) cl 29.

up to three billion and five hundred million dollars on the planning and financial obligations.²⁴⁷

The Third Five-Year Development Plan implemented by the Iranian parliament from 2000-2004 permits the conclusion of this type of contract.²⁴⁸ In the Fourth Five-Year Development Plan commended in 2005, the use of Buy Back Contract for the exploitation of oil and gas fields is emphasised. It states in paragraph (b) of Article 14:

The Ministry of Oil is authorised to act for discovery and exploration of more oil and gas fields across the entire country, as well as to the transfer and use of new technologies for exploration operations of onshore and offshore joint fields shared with neighbouring countries in which the relating exploration operations concerns risk-taking by the contracting party and leads to discovery of fields capable of commercial production.²⁴⁹

3.2.1 Annual Budget Law

Until 1949, foreign investment in oil and gas existed only in the downstream sector. With the end of the Second world War there was the need to rebuild the destroyed industries, attract foreign investment, and develop a program of economic, social and cultural rights.²⁵⁰ Iran's Annual Budget Laws and the licenses and guidelines for concluding the necessary contracts are important to this objective. For instance, Iran's 1953 Budget Law after the war saw foreign investment grow substantially leading to tangible outcomes and reflects the priority of the government to undertake Buy Back contracts with foreign companies. Moreover, Budget Law 1966 authorised further investment through the Buy Back Contract.

The Council of Ministers in the *Executive Regulations Promotion and Protection of Foreign Investments Act 1962* provided further incentives and facilities for foreign investment.²⁵¹ However, although the new law encourages investment protection and regulation of the world economy, it is arguable that it does not reduce the scepticism of foreign investors. It is the same scepticism that the authorities have towards foreign companies. To convince foreign investors to invest in Iran's industries, particularly the petroleum industry, requires greater transparency.

²⁴⁷ Law of the Second Five-Year Economic, Social and Cultural Development Plan para (a).

²⁴⁸ Law of the Third Five-Year Economic, Social and Cultural Development Plan art 85.

²⁴⁹ Law of the Fourth Five-Year Economic, Social and Cultural Development Plan art 14 para (b).

²⁵⁰ Chow et al (n 22) 4.

²⁵¹ Executive Regulations Promotion and Protection of Foreign Investments Act 1962, arts 2–3, art 4 para (a, b).

3.2.2 Annual budget laws and the five-year development plan

After the Islamic Revolution in Iran other legislation for the petroleum industry was passed. The Annual Budget Laws and the regulations in the Five-Year Economic Development Plans were adopted after the end of the Iran-Iraq War. The laws outline the rules around petroleum contracts and the Annual Budget Law provided the platform to attract foreign capital to restructure the oil and gas industry and to facilitate economic development. Prior to this, only limited economic plans and short-term options were utilised. The laws passed by Parliament in 1950 allowed the government up to 10 billion dollars in bilateral transactions with the central bank guarantee to meet the needs of the mining industry.²⁵²

The *Annual Budget Law 1953*²⁵³ and *Annual Budget Law 1954*²⁵⁴ gave the license to use reciprocal contracts to meet the needs of petroleum industry. The Second Five-Year Economic Development Plan (1955)²⁵⁵ addressed the financial needs of development projects and the needs of industrial and mining projects as defined in the law. Similarly, the Budget Law 1954²⁵⁶ allowed foreign investment up to the amount of \$3.5 billion in oil and gas projects. In accordance with Clause 22, paragraph (m) of the Second Five-Year Economic Development Plan, the Budget Law of 1957 reaffirms the issue of licenses for contracts with the NIOC. In Budget Law of 1958²⁵⁷ however, companies were allowed to settle debts and obligations arising from the financing of productive projects and job creation at a rate of \$4.5 billion from the production of the same design. This provision was repeated in the Budget Laws of the following years. The Budget Law 1959²⁵⁸ also set rules regarding the reimbursement of the cost of the reciprocal transaction.²⁵⁹

In reference to obtaining foreign finance for projects mentioned in Article 11 of the Third Economic Law (1960) – including petroleum in the form of budget bills – the Iranian government is permitted to use reciprocal transactions (in general).²⁶⁰ This is, however, subject to approval by the Economic Council following confirmation of its technical and economic feasibility.

The last notable change in the rules of petroleum contracts after the Islamic Revolution is Article 14 (b) in the Five-Year Development Plan adopted in 1965. The government and the

²⁵² Annual Budget Law art 29 para (a).

²⁵³ Annual Budget Law 1953 art 22.

²⁵⁴ Annual Budget Law 1954.

²⁵⁵ Ibid art 22 para (m).

²⁵⁶ Ibid cl 29 para (b).

²⁵⁷ Ibid prov $\overline{29}$ para (L-1)

²⁵⁸ Sect (L) pr 29.

²⁵⁹ Mohebi (n 240) 16.

²⁶⁰ *Third Economic Law* (1960) art 85 para (e).

NIOC use the same permissions in legislation to sign several contracts with foreign companies for oil and gas exploration. Criticisms were also directed towards the necessity of foreign investment, especially in the field of oil and gas exploration. This gradually persuaded the government to reconsider its policy towards exclusive use of reciprocal transaction contracts and open options to use other contract patterns. This included partnership contracts to participate more effectively in the global oil and gas arena while still maintaining national rights over the petroleum. However, the Fourth Economic, Social and Cultural Rights (ratified in 1965)²⁶¹ meant the government was permitted to encourage and support international agreements to attract foreign capital and resource use.

3.2.3 Legal support for foreign investment law

In the Regulations Act²⁶² there is emphasis on the pattern of participation and the foreign investment restrictions (whether direct or indirect) under the provisions of the law. In practice, however, this foreign investment law is not applied in the upstream petroleum industry. In fact, IOCs did not use the law and protection of foreign investment because the contracts did not fit within the framework of the law. In total, there are numerous laws and regulations governing the petroleum industry, the most important of which are the Oil Law 1947 amendments in 1971, the Fifth Five-Year Development Plan, the duties and powers of the Ministry of Petroleum, and the Annual Budget Laws. Also of importance are the general conditions, structure and patterns of upstream oil and gas contracts approved in 1975.

3.3 Conclusion

The historical development of the laws and regulations for petroleum contracts in Iran reveal that the element of sovereignty and ownership of petroleum resources had a significant role in the formulation of laws and petroleum contracts. An examination of the history of the petroleum industry in Iran makes clear that during the Qajar era the nation witnessed the arrival of outside experts to address weaknesses in the fields of finance, information and technology for research, and exploration and exploitation of petroleum resources. Granting concessions is understood historically to be a negative step. The Iranian government and people soon realised the importance of the nation's resources, and to access a greater share of the benefits called for more involvement in the operation. This led to the nationalisation of

²⁶¹ Art 14(b).

²⁶² Art 1(b) art 4(a-3) art 4 (b).

the petroleum industry in 1911 as certified by Dr Mohammad Mosadegh, Iran's Prime Minister from 1951-1953.²⁶³

Problems with foreign investors and domestic and international conspiracy against the nationalisation of the petroleum industry resulted in a coup against Dr Mosadegh in 1914. The reinstatement of consortium contracts saw Iran's resources plundered again. In 1918, the approval of partnership contracts attempted to address the role of national interests and sovereignty over national interests and shares in petroleum production. However, Iran's share of interest income from petroleum extraction was only properly addressed in the Oil Law of 1935. This emphasised that property and the sovereignty of the country was paramount and was intended for the bolster the oil and gas industry in Iran.

Service Contracts (reciprocal transaction contracts) emerged after the Islamic Revolution. Legal constraints including Articles 44, 80 and 81 of the Constitution (source privilege and foreign borrowing), Oil Law of 1947, and the limitations of the Annual Budget Law meant that such contracts were the only way Iran could use its upstream projects. One of the main objectives of the contracts was to ensure government sovereignty over oil and gas resources, and to maintain and monitor government operations. This was because Iran had shown special sensitivity when applying the oil rules of 1940 and 1947.

The history of petroleum contracts in Iran over the last century confirms that they challenged the interests of IOCs as well as the interests of the host government. In this sense, the type of contract and its conditions, which are comprised of many important factors, are crucial to the interests of the contracting parties. Petroleum companies and host countries will inevitably have interests in different types of contracts. The petroleum companies are understandably looking to ensure the security of their significant investment.²⁶⁴ As such, they have an overarching interest in sharing the risks associated with the exploration and development of oil fields with the host nation, and for there to be adequate provisions within the contract to expand operations to exploit opportunities for greater profitability.²⁶⁵ Conversely, the host country has an overarching interest in securing maximum returns from the foreign company for the exploitation and sale of its natural resources, along with access to advance technology and know-how to develop their natural resources sector.²⁶⁶ Despite there being little likelihood that either party will have all of their interests meet, Iran considered the benefits of

²⁶⁶ Ibid.

²⁶³ Parviz Kambin, A History of the Iranian Plateau: Rise and Fall of an Empire (iUniverse, 2011) 85-87.

²⁶⁴ R Global (n 91).

²⁶⁵ Ibid.

the existing contracts to be too much in favour of the foreign entities' interests . As a result, changes were introduced into the contract designs to give more consideration to how Iran can better benefit from the exploitation of its petroleum resources.

4 MODERN CONTRACTUAL FORMS: THE EMERGENCE OF BUY BACK TRANSACTIONS

4.1 Introduction

Global petroleum supplies were inevitably disrupted as a result of the frequent political and economic crises in Iran, particularly due to the Islamic Revolution and the Iran-Iraq War. Indeed, the early stages of the Iran-Iraq War brought petroleum production in Iran to a virtual standstill, providing a stark reminder of the sensitivities of petroleum production to tensions in the region.²⁶⁷ Buyers who ordinarily purchased petroleum from Iran turned now increasingly to Saudi Arabia and Kuwait for their supply, fearful of losing their tankers if they stayed with Iran.

By the 1990s, the petroleum production and export industry in Iran had stabilised but it did not reach the output achieved during the 1970s. Contributing factors were that oil wells were old and required upgrading if production capacity were to improve. An increase in production however meant the need for new oil fields and wells. This, in turn, required more capital and new technologies; namely, foreign investors.²⁶⁸ Thwarting access to further foreign investment was the ban placed on foreign investment in Iran by the US. However, some European and Asian interests invested in the development of the petroleum and gas industry in Iran, thus limiting the success of the US sanctions.²⁶⁹

Within Iran, there were also concerns from the government regarding the rapid growth in domestic energy consumption and its potential to reduce petroleum exports and thus affect the budget bottom line.²⁷⁰ To reduce the demand for petroleum in Iran, the government sought to increase the price of petroleum, to switch from petroleum to gas, and to simultaneously increase petroleum production. Iran at this time was eager to increase its energy exports but internal politics over the construction and maintenance of pipelines presented significant challenges to this outcome.²⁷¹

4.2 International Petroleum Agreements

 ²⁶⁷ Roger Stern, 'The Iranian Petroleum Crisis and United States National Security' (2007) 104(1) *Proceedings of the National Academy of Sciences* 377, 378.
 ²⁶⁸ Ibid 379.

²⁶⁹ Ibid 379. ²⁶⁹ Ibid 378.

²⁷⁰ Paul Rivlin, 'Iran's Energy Vulnerability' (2006) 10(4) *The Middle East Review of International Affairs* 103, 103.

²⁷¹ Ibid 103–104.

The domestic laws of host countries typically impose strict regulations and monitoring mechanisms in relation to petroleum contracts.²⁷² Such regulations cover all aspects of the contracts from the creation of a legal framework to stipulate the nature of foreign investment, to the exploration and production procedures and the setting of taxes. The regulations around revenue-sharing regulations are made explicit, and it is incumbent on International Oil Companies (IOCs) to consider their statutory obligations along with the standard terms of the contract set down by the host country when signing a petroleum agreement. Agreements that include the transfer of rights or ownership over natural deposits involve the government or a government-mandated company and typically fall under one of two regimes: concessionary systems or contractual systems.

As previously established in Chapter 2, the concession system represents the original legal framework established for exploration and production of petroleum and is the most commonly used contractual framework with states.²⁷³ An exclusive license is granted for the exploration of hydrocarbons and although the state has ownership of the petroleum *in situ*, upon drilling a well into the petroleum flow the concessionaire assumes title to that petroleum. In turn, the host nation is to be paid taxes and additional royalties including government representatives at all levels (local, regional, state and federal) and agencies controlled by the governments excluding state-owned companies.²⁷⁴ The concessionaire accepts all operational and investment risks. As a counterbalance to undertaking these risks, the IOC is permitted to locate, exploit, transport and sell the natural resource, typically within a specified region and over a specified time period. Taxes are then paid to the government upon sale of the resource along with the payment of other agreed expenses.²⁷⁵ Several types of petroleum contracts are included in concession agreements; with those signed prior to the 1950s including a wide range of concessions to IOCs and those signed after that relying more on a 50-50 share agreement formula. Within these agreements, the IOC assumes the risks and costs of exploration, while receiving remuneration via a fixed portion of the output, meaning the resources do not legally change hands.²⁷⁶

Non-concessionary contracts come in two categories: Production Sharing Agreements (PSAs) including JVAs²⁷⁷ and Service Contracts including Buy Back Contracts and IPCs. Service

²⁷² The territorial jurisdiction of the host government and where the foreign company has economic interest.

²⁷³ Bindemann (n 190) 1.

²⁷⁴ Christian Ule and Alexander Brexendrorff, *Investing in the Oil & Gas Industry* (Mena, 2005) 20.

²⁷⁵ van Groenendaal and Mazraati (n 2) 3715.

²⁷⁶ Ibid.

²⁷⁷ Also referred to as 'Production Sharing Contracts' (PSCs).

Contracts may be further divided into three types; pure service contracts, risk service contracts, and technical assistance contracts.²⁷⁸ PSAs and Service Contracts are more complex than concession schemes, setting out in extensive detail the regulations underpinning the host nation and IOC relationship. Table 4.1 provides a snap-shot of the main characteristics of each non-concessionary contract.

Туре	Description
Production Sharing	State retains ownership of the resources
Agreement (PSA)	Foreign oil company manages and operates petroleum field
	State and company negotiate a profit-sharing system
	Oil company accepts most exploration and development risks /
	costs
Joint Venture	Stakeholders (foreign oil company / host State) in venture execute
Agreements (JVAs)	business goals jointly
	Stakeholders share costs and risks of venture
	JVAs allow host country partner greater control over the venture
Service Contracts	Provide host Governments with the greatest control over the
	resource exploration and exploitation
	Foreign oil company performs a specified service and does not
	typically share in the revenue produced.
	Host Government maintains control of the resource
	The three main types of service contracts are:
	(1) risk service contract – oil company is contracted to provide all
	risk capital for exploration and production of petroleum. Successful
	discovery and production of commercial quantities of oil entitles oil
	company to recoup expenses and remuneration for services. Host
	Government maintains sovereignty over natural resources at all
	times.
	(2) pure service contract – Host Government provides risk capital
	for exploration and production of petroleum and oil company

Table 4.1 Non-Concessionary Contracts

²⁷⁸ Dariush Mobaser et al, Autopsy of the Oil Contract (Buy Back)' (2000) 21.

	provides states services for an agreed fee irrespective of discovery.
	Mostly used by countries with very high petroleum deposits.
	(3) technical assistance contract – mainly for rehabilitation,
	redevelopment or enhanced petroleum recovery projects. Oil
	company assumes operations of existing field including existing
	equipment and personnel to provide capital and technical know-
	how assistance
Buy Back Contracts	Capital costs are capped and recovered by the contractor as agreed
	at the commencement of the contract
	In addition to development costs, contractor receives an agreed rate
	of return from production revenue
	Contractor acts as operator during exploration and development
	phases, with State assuming operatorship at production phase
Iranian Petroleum	Costs recovery based on annual work programme and budget
Contracts	approved by contracting parties (NIOC has final right of approval
	in Iran). Contractor reimbursed for petroleum costs and paid a
	remuneration fee linked to production rates and revenue to costs
	(i.e. R factor) calculation indexed to market prices
	Contractor required to be a joint venture partner with State

Sources: R Global (n 55); Parris, Richard and Louis Skyner 1-4. Clifford Chance LLP.²⁷⁹

Comparison and analysis of the different types of petroleum contracts and their relative strengths and weaknesses is integral to understanding why Iran initially chose to use the Buy Back model as well as ultimately adopt the new IPC form.

4.3 Addressing the Imbalance

Development of Iran's petroleum sector and the level of cooperation it has committed to with foreign companies has been driven largely by decisions made by the Majlis Energy Commission (MEC). Over time and with the emergence of petroleum as a new and important energy source, the unsustainable inequalities in the concession system became increasingly

²⁷⁹ Key comparisons of the new Iran Petroleum Contract and Buy Back' <<u>https://www.cliffordchance.com/content/dam/cliffordchance/briefings/2017/01/key-comparisons-of-the-new-iran-petroleum-contract-and-buyback.pdf</u>> (accessed 1 December 2019).

apparent. Some host nations, being aware of what was at stake, raised their concerns and sought to renegotiate the inequitable terms in the contracts.²⁸⁰ The host nations challenged the legitimacy of the concession system on the grounds that they lacked balance, demanding *rebus sic stantibus* and *pacta sunt servanda*.²⁸¹ To clarify, in international law, the concept of *rebus sic stantibus* (translated as 'things standing thus') requires that in the situation of a fundamental change of circumstances, a party is permitted to withdraw from or terminate the contract.²⁸²

Conversely, the principle of *pacta sunt servanda* (translated as 'agreements must be kept') in international law seek to ensure the binding or enforceable nature of international contracts.²⁸³

Disputes continued for many years (e.g. Aramco Arbitration, 1958; Sapphire Arbitration, 1967; Texaco Arbitration, 1977) with formal adjudication by the Courts including the ICJ and the Permanent Court of International Justice (PCIJ).²⁸⁴ As such, the governments in some host nations felt the need to act unilaterally by gradually introducing ad hoc national legislation and then by either enforcing state participation or by the nationalisation of the petroleum sector. The concessions thus resembled a type of micro Constitution with its own laws enacted in domestic law and with disputes mostly settled through international arbitration.²⁸⁵

As previously mentioned, although concession agreements in the Middle East typically covered large regions for exploration and were generally of long duration (i.e. for more than 50 years), their terms and conditions were sometimes subject to renegotiation or amended by mutual consent in some instances of changing circumstances.²⁸⁶ The major concessions however gave the concessionary company the right to fix production rates and prices, thus affording them a superior negotiation position.²⁸⁷ Before the late 1960s, host nations typically accepted these conditions as they were guaranteed high rates of production and that their

²⁸² Encyclopædia Britannica (online) <<u>https://www.britannica.com/topic/rebus-sic-stantibus#:~:text=international%20law&text=The%20concept%20of%20rebus%20sic relevant%20island%20ha s%20become%20submerged</u>>.

²⁸³ Ibid <https://www.britannica.com/topic/pacta-sunt-

²⁸⁰ Mafi (n 119) 410.

²⁸¹ Piero Bernardini, 'Stabilization and Adaptation in Oil and Gas Investments' (2008) 1(1) *Journal of World Energy Law & Business* 98, 98.

servanda#:~:text=known%20by%20the%20Latin%20formula,would%20be%20binding%20or%20enforceable>

²⁸⁴ Mafi (n 119) 414, 415, 419.

²⁸⁵ Ibid 408.

²⁸⁶ Ibid 430.

²⁸⁷ Ibid.
product made it to market. Consumer nations maintained stable pricing which, although low, were not of great concern to host nation because they had not incurred costs of exploration and development; that is, had not borne significant risks for their revenue returns.²⁸⁸ Within the concessionary system,²⁸⁹ the transfer of the rights for exploitation followed an evaluation of the agreement by the host nation, based on the desire to ensure the interests of the state were met for conserving and exploiting natural resources. This is compared to the previous system where the rights to exploit the natural deposits were transferred without limits, and with decision making such matters as pricing and the rate of exploitation left to the IOC's discretion.²⁹⁰ This resulted in the payment of only small and fixed amounts of tax and royalties.²⁹¹ The political revolution in Iran during the 1950s and the subsequent nationalisation of the petroleum industry was the catalyst for significant changes to this arrangement, with host nations now having greater say in pricing and with the reconsideration of previous contracts.²⁹² For example, OPEC was established in 1960, passing Resolution XVI in 1968, calling for all previous terms in the contracts to be altered to include a gradual increase of state control (i.e. to 51%) by 1983. This change resulted in most member-nations acquiring partial or full control over the development operations by the 1970s.²⁹³

The case of Iran demonstrates how concession agreements did not always adequately serve the interests of the parties and, notwithstanding several attempts at re-interpretation and change, led the government to design their contracts in a completely different way. Even though the concession agreements had previously been more than satisfactory to the foreign petroleum companies with whom Iran was dealing, the desire to keep the original concession and the preparedness to renegotiate their terms was due to several factors. First, the petroleum companies were aware that the terms of the original contracts were particularly favourable to them. As a result, they were concerned that any refusal to renegotiate new terms and conditions may result in hostile action by the Iranian government such as tighter regulations of the petroleum industry or the seizure of assets.²⁹⁴ Second, given the extent to which the original concession terms favoured the foreign petroleum companies, agreeing to more equitable terms would still lead to the profitable production of petroleum. As a result, the

²⁸⁸ Judith Rees, *The International Oil Industry* (London, 1987), 60.

²⁸⁹ Also referred to as the 'Tax and Royalty System'.

²⁹⁰ Heidari (n 14) 1758.

²⁹¹ Ibid.

²⁹² Bernardini (n 286) 99.

²⁹³ Ibid.

²⁹⁴ Rami A Kamal, 'Overview of the Petroleum Industry in the Middle East (1869–1950)' (2005) 24(8) *The Leading Edge* 818, 818.

Consortium was prepared to be enter into new contract arrangements if they could continue to turn a profit from the exploitation of Iran's vast petroleum resources.²⁹⁵

Most modem concessions follow the example of the agreements made in Oman in 1967 and Abu Dhabi in 1974.²⁹⁶ These agreements gave foreign petroleum companies exclusive rights to the exploration, development, and exporting of petroleum, but over much shorter contract periods, clearer production obligations, the inclusion of a relinquishment clause, higher royalty and bonus payment requirements. In addition, the state or national petroleum company took a more active role in the venture. As such, the three areas given particular focus in the restructured concession system were the:

- level of control over operations afforded to foreign petroleum companies,
- share of revenue allocated to each party, and
- rights and obligations applied to foreign oil companies while operating in the country (e.g. production and investment requirements)²⁹⁷

The emergence of OPEC during the 1960s combined with an increase in the number of smaller 'independent' operators altered the contractual landscape to some degree. Despite several attempts to agree to new arrangements during the late 1960s and early 1970s (e.g. the Tehran Agreement on petroleum prices), the member states of OPEC undertook unilateral action to control the pricing of petroleum, thus bringing an end to the old concessionary regimes in 1974.²⁹⁸

In the Middle East, the trend to maintain state ownership and control over operations continued.²⁹⁹ By the mid-1960s, Production-Sharing Agreements (PSA) (discussed below) used by the Indonesian government were more commonly utilised in response to the general antipathy towards concessionary agreements. The following section discusses in more detail the various forms of petroleum contract. It also discusses the Buy Back contract as a precursor to the modern petroleum contract, the IPC.

4.4 Contractual Systems

To evaluate the competitiveness and attractiveness of the new IPC in the global petroleum market, an analysis of the alternative transaction systems is required. This section discusses

²⁹⁵ Ibid.

²⁹⁶ Bindemann (n 190) 10.

²⁹⁷ Ibid 9–10.

²⁹⁸ Ibid.

²⁹⁹ Rees (n 293) 61.

the main elements of the alternative schemes, considers the legal basis for the various frameworks and their suitability for both the foreign petroleum companies and Iran. It leaves the discussion of the most recent form of contract, the IPC to Chapter 5.

4.4.1 Production Sharing Agreements

Production Sharing Agreements (PSA) are legal instruments to regulate the relationship around petroleum extraction entered into by a government entity and a commercial petroleum company.³⁰⁰ Within the PSA, the government entity retains ownership of the resource and hires the IOC to provide technological and investment utilities during different phases of the extraction and production processes. The government entity typically acts through a proxy; namely, a national petroleum authority. The IOC is typically granted an agreed percentage of petroleum output as compensation for undertaking the risks of production and for the development of necessary facilities and services. The remaining output remains under the ownership of the government entity.³⁰¹

In addition, the government entity is permitted to be actively involved in certain exploitation and exploration operations, usually under the supervision and direction of a common governing body. In terms of the background to PSAs, they were first introduced into Indonesia in 1966 as unfavourable national sentiment towards foreign companies increased and national authorities were no longer prepared to endorse concessions.³⁰² The PSA sought to balance national interests against the need for foreign investment in petroleum production and the subsequent revenue. From Indonesia, the passing of legislation took place in "all oilproducing regions with the exception of western Europe" to permit the use of PSAs to ensure state ownership of natural resources.³⁰³ A degree of scepticism remained among foreign petroleum companies with operations in Indonesia based on their inability to legally possess or manage the petroleum reserves.³⁰⁴ An additional concern was that accepting the terms of PSAs in Indonesia would result in similar demands for the use of PSAs in other countries. As the large petroleum companies hesitated over whether to enter into such agreements, smaller operators took advantage of the opportunity to sign the PSAs and subsequently to wrestle some of the commercial power away from the larger operators.³⁰⁵ It was fairly common in practice however for the IOCs to maintain direct control over production of the petroleum

³⁰² Ibid 1.

³⁰⁴ Ibid 67–68.

³⁰⁰ Bindemann (n 190) 1-2.

³⁰¹ Ibid.

³⁰³ Ibid.

³⁰⁵ Ibid 61.

fields. In turn, as acceptance of PSAs became more common in the global petroleum market, they emerged as the preferred system in 52 states including Egypt, Syria, and the Philippines just to name a few.³⁰⁶

4.4.1.1 Key mechanisms of a PSA

The PSA operates according to the following process. First, the government as legal owner of the resource hands over the responsibility for locating and exploiting the petroleum reserves within a specified area and over an agreed span of time, with the expectation of an eventual compensation payment. Under this arrangement, the contractor accepts all risks and expenditures. Both parties are equal under the law in a PSA with penalties imposed for any breaches of the contracts terms and conditions,³⁰⁷ and provision for international arbitration in response to any emergent conflicts.³⁰⁸ In most countries around the world, the government maintains full control over the natural resource thus allowing it to permit access to the resource to one entity to the exclusion of others. However, there are limitations placed around the exploitation and production rights of the contractor as follows:³⁰⁹

- a) Only the types of operations agreed to in the contract are legal
- b) Only the natural resource agreed to in the contract may be extracted
- c) All other provisions of the contract function as limiting elements

4.4.1.2 The PSA and the state

The PSA is a private contractual arrangement in which the government retains power as representative of the nation to exercise all legal mechanisms at its disposal for the benefit of its citizens. This has the potential to create tension within the contractual arrangement because, although both parties are equal within the scope of the agreement, the state may pass laws that impact the agreement. As such, the host nation is automatically entitled to any petroleum produced by the foreign petroleum company, whereas the foreign company is only entitled to the production of petroleum via the PSA.³¹⁰

Although the PSA grants the foreign petroleum company the right to conduct operations within a specified region, there is no proprietary interest attached. As a result, the host government retains ownership of all output. This means that the foreign investor is acting on

³⁰⁶ Ibid 48.

³⁰⁷ Jenik Radon, 'The ABCs of Petroleum Contracts: License-Concession Agreements, Joint Ventures, and Production-Sharing Agreements' in Svetlana Tsalik and Anya Schiffrin, *A Reporter's Guide to Energy and Development of Open Society Institute* (Open Society Institute, 2005) 61–63.

³⁰⁸ Bindemann (n 190) 30.

³⁰⁹ Ibid 69.

³¹⁰ Ibid 68–69.

behalf of the government and is paid according to the level and distribution of the output.³¹¹ As mentioned earlier, the contractor accepts all expenses and risk, and there are no refund arrangements embedded into the PSA should no deposits be found, or if the project does not achieve commercial viability.

There is however the option to have a clause inserted into the PSA permitting some form of the compensation arrangement for the outlay of expenses.³¹² This arrangement allows a point to be determined within the terms of the contract (e.g. at the exit of the mine shaft) where an agreed amount of the output, termed the cost-recovery product, is allocated to the foreign company as compensation for their costs.³¹³ The remaining output, termed the profit product, is subsequently divided between the host country and foreign entity according to the terms agreed to in the PSA.³¹⁴ The PSA therefore allows the host nation to have access to a significant output share without having to invest significant funds or to take on commercial risk.³¹⁵ The amount of tax paid to the host government is largely defined in the contract terms. To avoid disincentives for investment, the amount of tax to be paid is typically offset by amount of royalty payments and profit product granted the host nation in the terms of the PSA. As a general rule, the lower the profit product and royalty payments to the government, the higher the level of income taxes paid on the product.³¹⁶

The tax regime applied to the foreign investor under the PSA is unique; that is, general tax payments are replaced by an allocated amount of output and extends for the duration of the contract.³¹⁷ This does not equate to the provision of tax exemptions or special privileges to the foreign investor however, but is rather a form of payment in kind.

Such production sharing arrangements thus provide a safety net for foreign company investors against_fluctuations in the nation's tax regime of a country or potential changes to tax law whilst also protecting the resource and revenue interests of the state.³¹⁸ Given that the PSA may last for decades, this can make a significant difference to the investor.

As such, PSAs emerged as an attractive contract arrangement to both the host nation and the foreign investor. The three main benefits of the PSA to the host nation were:

³¹⁵ Ibid.

³¹¹ Ibid 30.

³¹² Bindemann (n 190) 9.

³¹³ Ibid.

³¹⁴ Silvana Tordo, 'Fiscal Systems for Hydrocarbons: Design Issues' (World Bank Working Paper, no. 123, 2007) 44.

³¹⁶ Bindemann (n 190) 73.

³¹⁷ Ibid 13.

³¹⁸ Ibid 69.

- They attract substantial foreign investment for the petroleum resource related exploration, extraction and production thus contributing to economic stability in the country, and ultimately its national security. It is the case for many petroleum producing nations that the government would otherwise not be able to develop its petroleum industry.
- 2. They are typically long-term agreements which allow the host nation to forecast future growth in the rate of output to engage in budgetary planning more accurately.
- They allow the host government to determine the receipt of a fixed share of the output and thus avoid the conventional tax system which can be difficult to calculate on production output and equally difficult to collect.³¹⁹

The contractual relationship between the foreign entity and host government is nearly always exclusive and offer the foreign investor a degree of legal stability within the terms of the contract.

In terms of ensuring that Iran's interests are balanced against the interests of foreign companies, PSAs are restricted under the Constitution of Iran in the sense that all contractual agreements must support the Islamic Principle base on the stated Islamic criteria.³²⁰ However, the Iranian government permits such agreements for exploration of certain areas, if the contract terms are compatible with the legal limitations set down in the Constitution.³²¹ According to Seyyed Mehdi Hosseini, Deputy Managing Director of NIOC, the advantages and disadvantages of PSAs are best described as:

The risk the companies accept in PSAs is that of exploration, whereas most of our projects are for development and the exploration costs are not very high compared to development expenses and the revenues gained. This is particularly true in countries with a very high potential for petroleum discoveries, such as Iran. Therefore, the rewards can be huge and disproportionate to the risks. ... From the viewpoint of an oil-producing country, contracts such as PSAs or concessions may be best when the potential for discovering petroleum is low.³²²

³¹⁹Irina Paliashvili, 'The Concept of Production Sharing', Outlines of the Presentation of Dr. Irina Paliashvili, the President of the Russian-Ukrainian Legal Group, at the Seminar on the Legislation on Production Sharing Agreements, 14 September 1998 http://www.rulg.com/documents/The_Concept_of_Production_Sharing.htm (accessed 7 January 2020).

³²⁰ Art 4, 'Islamic Criteria'. The restrictions are also provided in art 44, art 45, and art 81 of Iran's Constitution. ³²¹ Bindemann (n 190) 9.

³²² Mehdi Hosseini, 'National Iranian Oil Company Deputy Managing Director, Speaks Up (Part One)' (May 2005) <www.payvand.com/news/05/may/1146 html> (accessed 19 December 2019).

4.4.2 Revenue sharing contracts

Revenue sharing contracts resemble PSAs in several ways, but with a different payment method. Similar to a PSA, the foreign investor ensures the provision of capital and technical expertise to support the exploration and production of the resource.³²³ The Revenue Sharing Contract then differs to the PSA in that the investor is paid according to an agreed share of revenue rather than production. The foreign investor thus seeks to recover the costs of a successful exploration via sales revenue. Hence, Revenue Sharing Agreements grant the IOC a stake in the petroleum following its discovery. Current law in Iran does not permit such an agreement.³²⁴ The Constitution of Iran stipulates that the nation's natural resources remain wholly owned by the state. Article 45 Public wealth and property states that such resource locations include uncultivated land, bodies of water (i.e. seas, rivers and lakes, public waterways), natural terrain (e.g. mountains, valleys, forests) unenclosed pastures, properties of unknown ownership, and public property regained from usurpers. All these locations remain available to the Iranian government for use in accordance with national interests.

4.4.3 Joint Venture Agreements

The host nation may at times enter into a commercial partnership agreement with an IOC or become a member of a consortium. These JVAs require the host country to contribute a percentage share of the capital investment to establish the commercial enterprise in return for an equal share of the 'cost oil' and 'profit oil'.³²⁵ Provisions in the JVA typically allow for the IOC's portion to be separated and this results in some of the risk being passed on to the government. Iran adopts a conservative perspective on JVAs and although it does not reject them outright, its preference was to use the Buy Back contract and since 2015 the IPC. In general, IOCs prefer to own part of the joint venture because it will remain a partner in the petroleum well for as long as it remains productive. In contrast, the Buy Back contract and the IPC do not permit the IOC to own any part of the well and once the invested capital and interest has been received, the oil field is vacated.³²⁶

The NIOC has entered into JVAs in the past;³²⁷ for example, two JVAs related to petrochemical deals were signed with 2001, one with ENI, an oil company in Italy, and one

³²³ Zhuo Feng, Shui-Bo Zhang and Ying Gao, 'On Oil Investment and Production: A Comparison of Production Sharing Contracts and Buyback Contracts' (2014) 42 *Energy Economics* 395, 396.

³²⁴ Ibid.

 ³²⁵ Abdolhossein Shiravi and Seyed Nasrollah Ebrahimi, 'Exploration and Development of Iran's Oilfields through Buyback' (2006) 30(3), *Natural Resources Forum* 200-201.
³²⁶ Ibid.

³²⁷ Dayarayan, 'Iran at a Glance' http://www.parstimes.com/business/iran_at_a_glance.pdf (accessed 19 December 2019) 4, 6.

with LG, the Engineering and Construction company in South Korea. In 2002, the National Petroleum Construction Company (NPCC) based in Abu Dhabi and Naft Sazeh Qeshm (NSQ) of Iran entered into a JVA in cooperation with Shell. The aim of the agreement was to build the necessary infrastructure to exploit the petroleum deposits of Soroosh. This was the largest ever contract for surface facilities at the time (i.e. to develop the Soroosh/Nowrooz oil fields) and the NPCC-NSQ joint venture was agreed to because it tendered the best proposal in terms of costs, scheduling and Iranian content. As such, the NPCC-NSQ joint venture was perceived to show real support for the development of Iran's petroleum industry.³²⁸ In addition, it aimed to capitalise on the expertise of NPCC in offshore fabrications to develop the Naft Sazeh Yard as well as to create employment and training opportunities for Iranians.329

Thus, notwithstanding that Iran generally prefers not to endorse JVAs, it is willing to enter into such agreements when the circumstances show that it is prudent to do so.

4.4.4 Service Contracts

Service Contracts are entered into for specific work to be done by a contractor at a fixed price or for an agreed share of the production profit. The three main types are the Pure Service Contract, the Risk Service Contract, and the Technical Assistance Contract.

4.4.4.1 Pure service contracts

Pure Service Contracts commit the contractor to undertake the exploration or development work at an agreed fixed price.³³⁰ The state adopts all the risk in this contract form. Countries in the Middle East often have the financial resources to develop their natural resources but lack the expertise and technology to do so. Pure Service Contracts are typically agreed to in order to gain access to a specific technical service at a specific point in time. It is generally the case that the service company will provide the equipment and personnel required to complete the work.³³¹ Payment for service may be a daily rate, fixed turnkey rate, or another specific fee arrangement and is generally not attached to project outcomes or market factors. In cases where payment for service is attached to agreed performance outcomes (e.g. operating cost reductions), the risk adopted by the service company is most often confined to overruns on non-recoverable costs or in relation to contract disputes (e.g. losses due to a contract breach or default).

³²⁸ Ibid.

³²⁹ Emirates News Agency, 'Abu Dhabi-Iran joint venture in Shell deal'' (January, 2002) < http://wam.ae/en/details/1395227046863> (accessed 9 August 2019).

³³⁰ Feng et al (n 328) 395. ³³¹ Ibid.

4.4.4.2 Risk service contracts

Risk Service Contracts required the service provider to invest capital into the project. It is mostly similar to the Pure Service Agreement however payment is based on the initial investment amount rather than a flat fee.³³² These contracts typically include a reserve agreement to be satisfied prior to the provision of any payments. Risk Service Contracts are mostly entered into for operations where expertise, technology and substantial investment are required to improve.³³³

4.4.4.3 Technical assistance contracts

This is a modernised version of the pure service contract where the contractor transfers technology to the host country in addition to supporting exploration and production activities.³³⁴ The host country maintains ownership of the resource and equipment and facilities as managed by the national oil company.³³⁵ The IOC then provides the technical services for the project. As such, the IOC performs a defined task for fixed compensation and cannot acquire an interest in the resource.³³⁶

The first Service Contract in Iran was concluded with the French company, ERAP. The primary advantage of the service contracts was that absolute ownership of the oil deposits could be maintained by the host nation. In this way, it is similar to a PSA in that it favours resource nationalism.³³⁷ Moreover, service contracts allowed the host nation to gain access to the IOC's expertise and capital without having to grant them ownership rights of the field and production.³³⁸ This is a vital consideration for the host nation because maintaining sovereignty over the natural resources increases the potential for proper oversight over the operations of the IOC. That is, they provide the state-owned oil company (e.g. NIOC) with a stronger position in its 'regulatory, supervisory and operatorship roles'.³³⁹ Moreover, it provides a pathway to improving the training provision to local workers through the transfer of technology and technical knowledge. The transfer of technology is considered as a major "externality" of a petroleum contract agreement and as playing an integral role in the eventual level of development of the host country. Externality in this context is defined as the

³³² Ibid.

³³³ Ibid 397.

³³⁴ R Global (n 91).

³³⁵ Ibid.

³³⁶ Ibid.

 ³³⁷ Ali Zamani, *Challenges in the Development of Iran's Oil Industry* (Desert Publications, 1960) 151-152.
³³⁸ Ghandi Abbas and C-Y. Cynthia Lin, 'Oil and Gas Service Contracts Around the World: A Review' (2014) 3 *Energy Strategy Reviews* 63, 63.
³³⁹ Ibid 66.

unintended or unanticipated benefit to Iran society and to its energy sector more specifically to emerge from the petroleum project itself.³⁴⁰ The positive transfer of technology within contract arrangements would typically include the requirement for nationals to receive training in operational practices directed by the foreign entity and for the promotion of local research and development initiatives to increase the level of indigenous technology use over time.³⁴¹

Notwithstanding the advantages of service contracts in regard to sovereignty concerns, this contract form was not automatically the preferred choice for the host nation or foreign oil companies. In terms of the host nation, a primary disadvantage of a pure service contract for instance was that the host nation was required to take on substantial operational and financial risks. The pre-determined remuneration to the IOC over the contracts lifetime make it difficult for the state-owned oil companies to facilitate changes to optimal production levels.³⁴² Regarding the IOC, the main disadvantage in a risk service contract for example, is that it is required to cover all the costs of exploration. Thus, the risks associated with a failure to find oil reserves were significant for the contractor.³⁴³

4.5 Buy Back Contracts

Buy Back Contracts are short-term agreements whereby the IOC is entitled to a share of the profit following the commencement of petroleum production. At the completion of the contract in Iran, the company is required to transfer all land and facilities to the NIOC.³⁴⁴ Under the terms of a Buy Back agreement, the IOC is required to recover the costs to explore for or develop of natural resource field. Repayment for all capital expenditure, interest paid, and the pre-agreed share of production is then facilitated through the sale of the oil or gas produced.³⁴⁵

Although Buy Back Contracts included fewer incentives for foreign investors than Production Sharing Agreements (PSAs), the IOCs were nonetheless willing to sign them. Several reasons can account for this. First, IOCs cannot discount the possibility that a favourable change in the contract structure may occur over time. Second, there is the strategic

³⁴⁰ Yinka Omorogbe, *The Oil ana Gas Industry: Exploration and Production Contracts* (Malthouse Press, 1997)90.

³⁴¹ Kameel Khan, 'The Transfer of Technology and Petroleum Development in Developing Countries: With Special Reference to Trinidad and Tobago' (1986) 4(1) *Journal of Energy & Natural Resources Law* 10, 17. ³⁴² Ibid.

³⁴³ Zamani (n 342) 152.

³⁴⁴ Shafiei Khah and Amiri (n 24) 378.

³⁴⁵ van Groenendaal and Mazraati (n 2) 3709.

concern that not signing a contract presents opportunities for rival companies to do so, which may in turn reap the benefits of any change in the contract structure should it come to pass. During the mid-1960s major IOCs were cautious about signing the new PSAs, but this form of contract was to become the most common.³⁴⁶ From the point of view of Iran, short-term contracts were the preferred option where foreign investment was needed to build infrastructure; whereas, the NIOC could manage the development and exploration of petroleum reserves.³⁴⁷

In Iran, the NIOC has only a supervisory role in a Buy Back Contract. The allocation of shares to each party in the contract is calculated by transferring gross production into gross revenue and then subtracting the costs of operations.³⁴⁸ The net revenue is then divided between the parties based on an agreed formula. The Buy Back Contract is typically defined according to a two-stage process: exploration of a natural resource field by the IOC; and then a declaration of its commercial or non-commercial viability. If the site is deemed to be non-commercial, the IOC accepts all costs and the contract is subsequently terminated.³⁴⁹ The site is regarded to have commercial potential by the NIOC if production output can provide a minimum rate of return following the subtraction of all capital costs, bank fees, operating expenses, and IOC costs.³⁵⁰ However, although the IOC may have explored and found a commercially viable field, it does not automatically get granted approval to develop the site. Rather, it only has the right to negotiate with the NIOC first to enter into a contract to develop the field.³⁵¹ If the negotiation is not successful, the contract goes through a tender process in which the successful IOC agrees to receive costs for its expenditure in addition to an agreed fee.³⁵²

Buy Back Contracts are of relatively short duration (i.e. between 5-7 years) and include an upper limit on amount of capital expenditure allowed (any increase in the amount must have NIOC approval). In addition, an important condition in the Buy Back Contract is the treatment of price risk.³⁵³ For instance, if there is a fall in revenue to a level that where the monthly entitlement of the IOC cannot be covered, the NIOC may accept a reduction in its share, or the amortisation period may be extended if this remains insufficient to cover the

³⁴⁶ Bindemann (n 190) 29.

³⁴⁷ Ibid 78.

³⁴⁸ Ibid.

³⁴⁹ Ibid 131–132.

³⁵⁰ Bindemann (n 190) 14.

³⁵¹ van Groenendaal and Mazraati (n 2) 3709.

³⁵² Bindemann (n 190) 78.

³⁵³ van Groenendaal and Mazraati (n 2) 3715.

IOC's entitlements.³⁵⁴ The main distinguishing aspect of this contract is that the contractor does not gain ownership of the petroleum (unless it is bought directly from the host nation). The main differences between a PSA and a Buy Back Contract are what made the latter less attractive to potential investors. For instance, the Buy Back Contracts lacked flexibility and security in terms of tax and investment arrangements given the potential for changes to legislation to be introduced in Iran, and the onus on the contractor to accept all exploration and development risks.

The Buy Back Contract risk is that commercial production is not a right. In this regard, the Fourth Five-Year Development Plan declares: "... If at the end of the exploration phase, commercial areas in any part of the area are undetected, the contract will be terminated, and the parties will not have the right to claim any aspect".³⁵⁵ Arguably, the main differentiating factors between Buy Back Contracts and PSAs is related to risk. The contractor in a Buy Back scheme is reimbursed for exploration and development costs from the sale of petroleum production.³⁵⁶ In addition, a contracted company is penalised under the Buy Back scheme if it fails to produce the amount of petroleum agreed as it is subsequently reimbursed to a lesser amount than in earlier agreements. Thus, Buy Back Contracts have increased risk compared to other finance deals.³⁵⁷

4.5.1 International Buy Back Contracts

Within the structure of the Buy Back Contract is the transaction for the use of machinery, equipment, knowledge and technical expertise throughout the construction of the production facilities. Although there are generally only two negotiating parties in a Buy Back Contract, there is the possibility for a three-party or larger transaction to take place. For example, goods or materials may be provided by a third party other than the primary-transaction contractor, who subsequently sells or leases it to the primary-transaction company.³⁵⁸ In addition, the primary-transaction company may seek to transfer some of its Buy Back obligations; for example, transferring responsibility for the sale of the product to a third party rather than undertake this responsibility itself. In such circumstances, the third party is required to sign

³⁵⁴ Bindemann (n 190) 80.

 ³⁵⁵ Law of the Fourth Five-Year Economic, Social and Cultural Development Plan art 14 para (e).
³⁵⁶ Dr. Hossein Afarideh, 'Iran Ready to Give Production Sharing Agreement' (Menas Associates' Iran Energy Focus, February 2003) <<u>http://www.payvand.com/news/03/mar/1026 html</u>> (accessed 12 June 2019).
³⁵⁷ Ibid.

³⁵⁸ United Nations Economic Commission for Europe, 'International Buy Back Contracts' 1 <<u>http://ierc.bia-bg.com/uploads/vtcontract/files/vtcontract</u> 00743d3417d145769033e1b5c06e4113.pdf (accessed 9 December 2019).

its own contract with the host nation.³⁵⁹ Such third party (International) Buy Back agreements can be very complex and may be as a single contract or be designed as multiple interconnected agreements.³⁶⁰ A single contract is typically entered into when the product is precisely described and there is no third-party involvement. Notwithstanding the flexibility of the international Buy Back Contract structure, the termination date is typically clearly defined.³⁶¹

Similarities are apparent between Iranian and international Buy Back Contracts in regard to the rights and obligations accepted by the contracting parties including third-party operators supplying technology and equipment. Where they differ, however, is in regard to the special Buy Back provision where the party who establishes operations in a Buy Back Contract is contractually bound to purchase the end petroleum produced through the operation.³⁶² Any disagreements or disputes to occur as a result of the complex and often onerous nature of the contract obligations are resolved according to the pre-agreed laws governing the agreement. Prior to taking the dispute to Court, however, an attempt to negotiate an acceptable outcome is advised. In the event of a failure to reach a compromise, international arbitration procedures may be used, but the choice of arbiter and arbitration rule must be included in the contract.³⁶³

4.5.2 Limitations of Buy Back Contracts

The three main elements underpinning the IPCs relate to collaboration (primarily through its JVA structure), technology transfer and technical know-how, and simplification and consolidation of previous model agreements.³⁶⁴ These principles emerge from criticisms levelled against the Buy Back Contract model, particularly regarding its limited capacity to meet the national interests of Iran.³⁶⁵ See Section 5.2 below for a comprehensive discussion of the 'pros' and 'cons' of the Buy Back Contract model for Iran's oil and gas sectors and the entities (i.e. NIOC) operating within it. The main criticism of the Buy Back Contract by IOC's and that which provides this thesis with its answer to RQ2 (see Section 1.4) is related to the capped cost recovery regime. Specifically, the expectation that the IOC would invest in the development of designated areas on the understanding that any costs incurred beyond the

³⁵⁹ Ibid 3

³⁶⁰ Ibid.

³⁶¹ Ibid 9.

³⁶² Ibid 8.

³⁶³ Ibid 3.

³⁶⁴ James Dallas and Alistair Black, 'New Model Iranian Petroleum Contract' *Dentons* (2016) <<u>https://www.dentons.com</u>> (accessed 1 December 2019) 3.

³⁶⁵ Ardalan et al (n 26) 350.

pre-determined budgets were irrecoverable. This acted as a disincentive to IOCs to invest in projects that carried greater risk or with marginal prospects only of success.³⁶⁶ In turn, the new IPCs aimed to address this limitation, along with the narrow remuneration structure in place to reward enhanced productivity, by implementing three key changes:

- full costs recovery amortised over 5-7 years and annual costs recovered and service fee repayments kept to 50 percent of total annual revenues
- 2. the remuneration of finance and other associated development fees (e.g. income tax and customs duties) incurred throughout the amortisation period
- replacement of the fixed-cost regime with a standardised yearly work programme and budgeting process³⁶⁷

The Buy Back Contract model was also criticised for its remuneration structure, with Contractor fees tied to a fixed percentage of capital costs up to the budgeted cap.³⁶⁸ This element did not allow for any incremental revenue for exceeding production targets or any upside linked to an increase in the price of petroleum. The new IPC model could offer a volumetric fee structure for the duration of the agreement.³⁶⁹ In turn, a revised cost-recovery structure combined with a true volumetric tariff (with incentives for incremental production) would incentivise foreign investment in the technology required to optimise green and brownfield reserve production.³⁷⁰

4.6 Conclusion

This Chapter has conducted a broad examination of the petroleum contract schemes in order to establish the context to the IPC by identifying and discussing the precursors to the current model. The Chapter has explored the function and legal basis of PSAs, Concessionary Models, Revenue Sharing, JVAs and Buy Back Contracts to assess their compatibility to, and function within, Iranian law. The analysis revealed that Buy Back Contracts are amongst the least profitable options for IOCs, justifying to some degree their dissatisfaction with the scheme.³⁷¹ However, the Chapter also demonstrated that other contract options included terms and conditions too favourable to the foreign investor company, thus creating a

³⁶⁶ Ibid 351.

³⁶⁷ United Nations Economic Commission for Europe (n 363) 4.

³⁶⁸ Ibid 5.

³⁶⁹ Ibid.

³⁷⁰ Ibid.

³⁷¹ van Groenendaal and Mazraati (n 2) 3711.

reasonable context for Iran to be unwilling to maintain such agreements due to perception of the unfair and exploitative use of the nation's natural reserves.

Moreover, given the potential for some contract schemes to lack alignment with the Constitution of Iran and some of the nation's statutes, particularly in relation to the ownership of resources by foreign entities, it is evident that the IPC emerges as one of the best alternatives. To support a detailed and accurate analysis of the IPC in terms of its strengths and flaws – and subsequent suggestions for improvement – the following Chapter provides a comprehensive explanation of its legal basis, provisions and comparative benefits in relation to alternative international petroleum contracts.

5 IRANIAN PETROLEUM CONTRACT – AS A RESPONSE TO THE PROBLEMS WITH BUY BACKS CONTRACTS

5.1 Energy Sector in Iran

The energy sector in Iran has been significantly impacted and constrained by the sanctions imposed on the country by the US after the Iranian Revolution in 1979, the UN in 2006, and more recently by the US again in 2018.³⁷² These sanctions have unquestionably impeded the social and economic development and progress of Iran that was acknowledged as a major player in the global energy sector. The sanctions arguably demonstrate the need for Iran to develop a robust and efficient energy sector that has the capacity to develop and operate its energy assets to the betterment of the nation.³⁷³ However, in order to transition Iran's state-dominated energy sector to a modern and globally competitive industry, it must have the capability to attract significant foreign investment capital.³⁷⁴

According to Smousavi, as of 2019, '[b]oth the size of the Iranian energy sector and its influence in the region is expected to grow'.³⁷⁵ This growth was promoted to some extent through the introduction of the IPC in 2015 and the relaxation of sanctions against the nation in 2016. The easing of sanctions especially provided new opportunities for IOCs to engage with Iran to exploit its energy resources. Conversely, the re-imposition of the sanctions by the US in 2018 has placed serious constraints on both IOCs and the Iran to develop energy sector.³⁷⁶

On 14 July 2015, the Guardian Council of the Islamic Republic of Iran signed the Joint Comprehensive Plan of Action (JCPOA), a multilateral nuclear agreement aligned with Iran's constitution and Islamic law more broadly.³⁷⁷ Following proof of compliance with the plan of action by the UN Security Council on 16 January 2016, the sanctions related to the production of nuclear energy were lifted. Yet, the relief for Iran's energy sector was relatively short lived, with the Trump administration withdrawing from the JCPOA In May 2018, followed by the re-imposition of sanctions on Iran's energy sector on 5 November 2018.³⁷⁸

³⁷² Shaghayegh Smousavi, 'Iran: The Energy Regulation and Markets Review - Edition 8' (2019) *The Law Reviews* para 1.

³⁷³ Ibid para 2.

³⁷⁴ Ibid.

³⁷⁵ Smousavi (n 377) para 3.

³⁷⁶ Ibid.

³⁷⁷ Faucon (n 25) para 2.

³⁷⁸ Ibid.

In the context of the current US sanctions regime combined with the desire by the EU and other nations to continue to trade with Iran, it is vital that due diligence is demonstrated by IOCs to assess the extent to which they are exposed to the US sanctions.³⁷⁹ Moreover, IOCs must assess the risks around Iran potentially violating the terms of the JCPOA and the subsequent need to implement the 'snapback' provisions; namely, the reimposition of sanctions on Iran.³⁸⁰ Furthermore, this is in the broader context of Iran's low ranking (128th) on the 'Doing Business' ranking of economies by the World Bank.³⁸¹ Implied in this ranking is the need for IOCs to be alert to the risks of bribery and corruption in Iran's energy sector, and to the challenges presented by Iran's wider infrastructure development needs.³⁸²

Notwithstanding the relatively low global oil price as a disincentive for IOCs to invest significant amounts of capital in the development and production of new oil fields, "the costs of production in Iran are estimated to be significantly lower than the international average".³⁸³ As such, Iran considers this enough incentive for IOCs that it has sought to implement its reforms its energy sector and continue to open it up to foreign investors.³⁸⁴ As previously mentioned, the IPC endorsed by Iran's parliament in October 2015 and eventually ratified by the parliament in September 2016 was designed to facilitate such reforms.

Articles 77 and 125 of Iran's Constitution state that all international agreements have to have the approval of parliament:

Article 77:³⁸⁵ Treaties, transactions, contracts, and all international agreements must be ratified by the Islamic Consultative Assembly.

Article 125:³⁸⁶ All the treaties, transactions, agreements, and contracts between the government of Iran and other governments as well as all the pacts related to the international unions, after they are approved by the Islamic Consultative Assembly, must be signed by the President of the Republic or his legal representative.

³⁷⁹ Hamed Sahebonar, Ali Taheri Fard and Fazel M. Farimani, 'Economic Analysis of New Iranian Petroleum Contract (IPC): The Case Study of the Caspian Sea Fields' 1 <<u>http://www.iaee.org/baku2016/submissions/ExtendedAbs/Sahebhonar.pdf</u>> (accessed 17 February 2020).

³⁸⁰ Ibid.

³⁸¹ World Bank 'Ease of Doing Business Rankings' http://www.doingbusiness.org/en/rankings?region=middle-east-and-north-africa.

³⁸² Sahebonar et al (n 423) 1.

³⁸³ Smousavi (n 377) para 15.

³⁸⁴ Ibid.

³⁸⁵ Constitution of Iran (n 217) 18.

³⁸⁶ Ibid 28.

However, as Smousavi points out, the Iranian parliament has also previously accepted that "contracts in which one side is a government entity or company and the other side is a privately owned foreign company are not international agreements subject to Article 77".³⁸⁷

In March 2018, during the period in which the US sanctions were lifted, a trilateral IPC was signed by the NIOC, Zarubezhneft (a Russian state-owned oil company) and Dana Energy Company to develop the Aban and West Paydar oilfields around the border of Iran and Iraq.³⁸⁸

A further development in Iran's petroleum industry at this time was the reintroduction of the Iran Energy Exchange (IRENEX). Initially established in 2012, the IRENEX permitted the Ministry of Petroleum in Iran to sell crude oil and petroleum products in the international market at a competitive price.³⁸⁹ Operating under the umbrella of the Securities and Exchange Organization, IRENEX aims to organise and supervise the trade of energy carriers. In turn, IRENEX facilitated the purchase of crude oil from Iran during different rounds of the supply.³⁹⁰

5.1.1 Competition in Iran's oil fields

The competitiveness and attractiveness of Iran's energy sector is determined to some extent by the outcome of the debate between conservatives and moderates in Iran regarding the best utilisation of Iran's natural resources. With conservatives pushing for an independent and self-sufficient energy sector, and moderates pushing for the energy sector to drive economic and social advancement (through access to foreign capital and expertise), there remains a degree of risk for IOCs around the political and policy uncertainty in the sector.³⁹¹ The energy industry in Iran has been in public hands throughout the sanction periods, with the IRGC highly involved in its management.³⁹² As such, the interests of vested interest groups must be overcome and any attempt to open up the sector may be regarded with suspicion by the IRGC and the domestic energy sector supply chain. This is because the proposed reforms

³⁸⁷ Smousavi (n 377) para 20.

³⁸⁸ Ibid para 22.

³⁸⁹ Ibid para 23.

³⁹⁰ Ibid.

³⁹¹ Smousavi (n 377) para 44.

³⁹² Ibid.

fundamentally redefine the role and scope of the NIOC and the reliance on foreign investment and technology.³⁹³

The introduction of reforms around competition and tariffs in Iran's energy sector has several potential benefits. The primary objective of the reforms is to make sure that the risks around investing in the energy sector are assigned to the entity that is best able to manage them as well as to drive better investment decision making.³⁹⁴ Competition and the loosening of restrictions aim to shift major performance risks to the private sector, exploit the advantages of competition via the introduction of new technology and international best standards into the sector to the domestic benefits to the economy and citizens.³⁹⁵

Given that the Ministry of Energy in Iran is also the regulator of the energy sectors as well as a direct investor in it, there is significant potential for conflicts of interest. In order to implement effective regulation, there are many advantages to be gained from separating key aspects of the State from the sector.³⁹⁶ However, any reforms to liberalise the sector must be managed carefully so as to make sure that there is not too much upward pressure placed on energy prices during the process which would be difficult to manage, politically.³⁹⁷ This is because such reforms, particularly around the rejuvenation of aging oil fields, are reliant on the large-scale capital and technology from foreign companies, and will inevitably increase domestic oil consumption https://www.govinfo.gov/content/pkg/CHRG-109shrg30856/html/CHRG-109shrg30856.htm

Moreover, the process must be managed carefully to make sure too much pressure is not placed on existing entities to reduce costs and subsequently produce financial problems for the sector.³⁹⁸ The introduction of stronger competition into the sector also requires the building up of capacity of key institutions that will be managing the skills, resources, and expertise as well as educating the entire supply chain on the processes adopted in Iran.³⁹⁹

³⁹³ Ibid.

³⁹⁴ Vahid Aryanpur, Mohammad Saeid Atabaki, Mousa Marzband, Pierluigi Siano, and Kiarash Ghayoumi, 'An Overview of Energy Planning in Iran and Transition Pathways Towards Sustainable Electricity Supply Sector' (2019) 112 *Renewable and Sustainable Energy Reviews* 58, 58.

³⁹⁵ Ibid.

³⁹⁶ Smousavi (n 377) para 45.

³⁹⁷ Ibid.

³⁹⁸ Ibid.

³⁹⁹ Aryanpur et al (n 399) 59.

5.2 Attracting Foreign Direct Investment into Iran's Energy Sector

Foreign investors and developers seeking to establish operations in Iran's energy sector must naturally make decisions as to how they want to enter into, and engage with, the Iranian market. As previously established, the IPC provides one such route for entry and engagement via the creation of a JVA with a local entity. It is therefore vital for IOCs seeking to enter the Iranian energy sector to understand the current laws, regulations and industry frameworks in place to manage the sector. Iran's Constitution dictates that all national laws and regulations are based on "Islamic principles and norms that reflect the heartfelt desire of the Islamic community".⁴⁰⁰ In turn, Iran promotes the participation of foreign entities in its energy sector via the *Foreign Investment Promotion and Protection Act (FIPPA) 2002.*⁴⁰¹ Under the FIPPA, foreign entities can invest in the mining sector and greenfield and brownfield project services upon satisfying of specific criteria. As stated under Article 2 of FIPPA – General Conditions for Admission of Foreign Capital:

Admission of Foreign Investment under this Act and in compliance with other current laws and regulations of the county must be for development and productive activities in the fields of industries, mines, agriculture and services shall be based on the following criteria:

a. Shall lead to economic growth, promote technology, promote quality of productions, increase employment opportunities and increase exportsb. Does not jeopardize national security and public interest, harm the environment, disrupt the national economy, or disturb productions dependent on domestic investments

c. Shall not involve the granting of concession by the government to Foreign Investors; concession means distinctive rights that place foreign investors in an exclusive and monopolistic position

d. The proportion of the value of goods and services produced by Foreign Investment under this Act in comparison with the value of goods and services supplied in the domestic market at the time of issuance of Investment License, in each economic sector, shall not exceed 25% and in each economic sub-sector shall not exceed 35%.

⁴⁰⁰ Constitution of Iran (n 217) 1.

⁴⁰¹ Yeganehshakib (n 230) 1.

The determination of sub-sectors and amount of investment in each will be pursuant to regulations ratified by the Council of Ministers.⁴⁰²

Foreign direct investment (FDI) is permitted in fields where private sector participation is authorised. However, wholly commercial activities are not regarded as foreign investment. Foreign investors can therefore choose either FDI or foreign investment across all sectors within a 'civil participation' framework, as stated in Article 3 of the Implementation Regulation of FIPPA:

Admission of Foreign Investment, based on FIPPA and the criteria set forth in these Regulations, may be carried out within the framework of the following methods. The table of investment methods, features and facilities available under FIPPA shall be prepared and published by the Ministry of Economic Affairs and Finance.

a. Foreign Direct Investment (FDI)

b. Foreign Investment within the framework of contractual arrangements including various types of "Build-Operate-Transfer" (BOT), "Buy Back", and "Civil Participation" schemes.⁴⁰³

In Iran, the Organization for Investment Economic and Technical Assistance of Iran (OIETAI) is responsible for issuing a foreign investment licence under FIPPA as stated in Article 15 of the Implementation Regulation of FIPPA:

Investors shall submit to the Organization their written application together with the documents specified in the relevant form. After conducting necessary investigations and taking the viewpoints of the ministry responsible for the related sectors, the Organization shall bring the investment application along with its expert advice before the Board within a maximum period of 15 working days. Enquiries remained unanswered by the relevant ministry, after 10 days from the date of receipt of the enquiry shall be considered as agreement of that ministry with the investment concerned. On the basis of the decisions adopted by the Board for which the acceptance of the Foreign Investor has already been obtained, the Investment License shall be drafted and, upon confirmation and signature by the Minister of Economic Affairs and Finance, shall be issued.

⁴⁰² Organization for Investment Economic and Technical Assistance of Iran, 'Laws and Implementing Regulations Concerning Promotion and Protection of Foreign Investment' <<u>http://www.satba.gov.ir/suna_content/media/image/2017/02/5236_orig.pdf</u>> 9-10.

⁴⁰³ Organization for Investment Economic and Technical Assistance of Iran, 19.

Note: The Investment License shall include the particulars of the investor(s), type and method of investment, the manner for transfer of dividend and profit gained as well as other terms and conditions relating to the approval of every investment project.⁴⁰⁴

The licence treats foreign investment as equal to Iranian investments as stated in Article 8 of FIPPA:

Foreign investments subject to this Act shall enjoy the same rights, protections and

facilities available to domestic investments in a non-discriminatory manner.⁴⁰⁵ In addition, it permits the resolution of disputes to take place outside of Iran and has provisions for the repatriation of profits. The OIETAI also provides security against noncommercial risks such as the transfer of currency, as stated in Article 4 of the Implementation Regulation of FIPPA:

Methods of investment referred to in Article (3) of these Regulations, in respect of the procedure for investment and the protection coverage of FIPPA and these Regulations, have the following common or specific features and advantages:

- a. Common features and advantages:
 - 1. Foreign Investors enjoy the same treatment as accorded to domestic investors.
 - 2. Import of Foreign Capital, being cash or non-cash (in kind), is only subject to the Investment License and does not require any other license.
 - 3. The volume of Foreign Investment in each individual case shall not be subject to any limitation.
 - 4. Foreign Capital is guaranteed against nationalization and expropriation, and in such cases the Foreign Investor shall be entitled to receive compensation.
 - 5. Transfer of the principal capital, profit and capital gains derived from utilization of capital shall be affected in the form of foreign currency or, as the case may be, in the form of goods, as set out in the Investment License.
 - 6. The freedom to export goods produced by the Investee Firm is guaranteed and, in the event of any prohibition on the export, the goods produced may be sold in the domestic market, and proceeds of sale shall be transferable abroad in the form of foreign currency through the Country's Official Monetary Network.
- b. Specific features and advantages:

⁴⁰⁴ Ibid 24-25.

⁴⁰⁵ Ibid 12.

- 1. Foreign Direct Investment (FDI):
 - i. Investment may be made in all areas where the private sector activity is permitted.
 - ii. There is no restriction on the percentage of foreign shareholding.
- 2. Investment within the framework of contractual arrangements:
 - Compensation for losses sustained by the Foreign Investment resulting from prohibition and/or interruption in the execution of financial agreements caused by enactment of law and/or Cabinet decrees, up to a maximum of matured instalments, shall be guaranteed by the Government.
 - In "B.O.T." and "Civil Participation" schemes where a government agency is the sole purchaser and/or supplier of goods and services at subsidized prices, the purchase of produced goods and services resulting from an investment project by the government agency as a party to the contract, shall be guaranteed in accordance with the relevant regulations.⁴⁰⁶

As for nationalisation and expropriation considerations, Article 9 of FIPPA states:

Foreign Investment shall not be expropriated or nationalized unless for the public interest, through a legal process, in a non-discriminatory manner, and against payment of appropriate compensation based on the real value of that investment immediately before the expropriation.

Note 1: Requests for compensation must be submitted to the Board within a maximum of one year following the expropriation or nationalization.

Note 2: Disputes resulting from expropriation or nationalization will be settled according to Article 19 of the present Act.⁴⁰⁷

Moreover, the regulations around interventions by the government, and government contract breaches are stated in Article 17 of the Implementation Regulation of FIPPA:

The Ministry of Economic Affairs and Finance (State Organization for Tax Affairs, Customs of the Islamic Republic of Iran), the Ministry of Foreign Affairs, the Ministry of Commerce, the Ministry of Labor and Social Affairs, the Ministry of

⁴⁰⁶ Ibid 19-21.

⁴⁰⁷ Ibid 12-13.

Industry and Mines, the Ministry of Jihad-e-Agriculture, the Central Bank of the Islamic Republic of Iran, the General Directorate for Registration of Companies and Industrial Property, the Organization for Protection of the Environment, and other executive agencies determined by the Minster of Economic Affairs and Finance shall introduce their fully authorized representatives to the Organization with the signature of the highest executive authority of the agency. The designated representatives, from the standpoint of the employment regulations, shall be considered as the employees of their respective agencies, and, as situation requires and in proportion to the volume of Foreign Investment applications and enquiries by the investors, shall, upon the Organization's request, 6 be present in the Center in order to respond to the enquiries in accordance with the duties assigned to them under this Article.⁴⁰⁸

Regarding the expropriation and nationalisation of the assets of foreign investors, FIPPA acknowledges the foreign entity's right to obtain immediate compensation at fair market value for the assets on the day prior to their expropriation (see Article 9 of FIPPA above).

5.2.1 Foreign direct investment and the Iranian Petroleum Contract

Foreign direct investment (FDI) includes major investments by foreign entities such as in the development of production facilities or via the purchase of ownership stakes in Iranian companies. The benefits of FDI in its energy sector for Iran include the creation of new jobs, an injection of state-of-the-art technologies, and enhanced understanding of management strategies and workplace practices.⁴⁰⁹

As the Organisation for Economic Cooperation and Development (OECD) explains, FDI is arguably a greater catalyst to domestic development than trade and is a vital component in any open and successful international economic system.⁴¹⁰ However, the benefits and advantages of FDI do not accumulate automatically and evenly for different countries and across different sectors. Moreover, a nation's policies and foreign investment architectures play a key role in attracting FDI and in gaining access to the full benefits of FDI.⁴¹¹ For Iran, the challenges around attracting FDI in its energy sector are associated with ensuring a

⁴⁰⁸ Organization for Investment Economic and Technical Assistance of Iran 25.

⁴⁰⁹ Seyed Mohammad Alavinasab, Determinants of Foreign Direct Investment in Iran' (2013) 3(2) *International Journal of Academic Research in Business and Social Sciences* 263.

 ⁴¹⁰ Organisation for Economic Cooperation and Development, 'Foreign Direct Investment for Development, Maximizing Benefits, Maximizing Costs' (OECD, 2002) 3.
⁴¹¹ Ibid.

transparent, wide-ranging and enabling policy setting for foreign investment that is supported with adequate human and institutional resource capacities.⁴¹²

There are a range of strategies available to a nation to improve its capacity to attract foreign capital into the country. In terms of legal considerations, statuary strategies may be adopted that rely on the legislative power of the parliament. This approach imposes conditions or obligations on the part of the contracting parties and typically reflects the government's economic and social policies.⁴¹³ Alternatively, contractual strategies may be deployed by the host nation to attract foreign capital. Such strategies generally include the provision of incentives or specific terms and conditions that entice the foreign entity into the contract and the transfer of capital into the country.⁴¹⁴

Whatever the strategic approach, the overarching objective is for the host nation to appear as attractive to the foreign investor company. The main points of attraction are generally regarded in terms of the stability of the commercial environment and the capabilities of the host nation to engage in long-term cooperative endeavours to the benefits of both parties.⁴¹⁵ The lifting of sanctions on Iran in 2015 marked the start of a new determination from the Iranian parliament to progress and develop the energy sector on the back of foreign investment.⁴¹⁶ However, the reinstatement of the sanctions by the US in 2018 has clearly had a significant impact on Iran's energy sector and its wider economy.⁴¹⁷

Iranian Constitutional Law, Article 77, dictates that any international agreements entered into by Iran must be approved, ratified and supervised by Parliament.⁴¹⁸ However, contracts entered into by the Petroleum Ministry or the NIOC with foreign companies or governments are not covered by Article 77 and thus are not monitored by the Iranian Parliament directly.⁴¹⁹ There are however other mechanisms within the regulatory frameworks in Iran for its Parliament to monitor petroleum contracts. For instance, if the petroleum contract is worth more than US\$20 million, the government is required to inform the Planning and Budgeting Commission and Energy Commission of Parliament of the conclusion of the contracts and the

⁴¹² Alavinasab (n 414) 259.

⁴¹³ Sheldon Leader, 'Human Rights, Risks, and New Strategies for Global Investment' (2006) 9(3) *Journal of International Economic Law* 657, 657.

⁴¹⁴ Ibid.

⁴¹⁵ Alavinasab (n 414) 25.

⁴¹⁶ Ibid.

⁴¹⁷ Ibid.

⁴¹⁸ Mahdi Haddadi, 'The Nature of Iranian Petroleum Contracts in Upstream Section' (2014) 44(1) *International Letters of Social and Humanistic Sciences* 45, 50. ⁴¹⁹ Ibid.

processes for its finalization. Additionally, in accordance with Petroleum Law Modification, 1987, classified copies of all petroleum contracts of more than five years' duration should be submitted by the government to Parliament.⁴²⁰

The new Petroleum Law passed in 1987 allowed the Petroleum Ministry to conclude petroleum contracts in addition to the NIOC. The model applied by the Ministry was based on the procedures applied prior to 1987.⁴²¹ The modifications to the Petroleum Law (1974) passed in 2011 provided the Ministry with the right to govern and own petroleum resources on behalf of the Iranian state.⁴²² Moreover, Article 16(1) of the modified Petroleum Law (1974) identified the Petroleum Ministry and the NIOC as competent authorities to conclude petroleum contracts in the upstream sector of the petroleum industry.

Operations of IOCs in the upstream sector of the petroleum industry in Iran are undertaken within the terms and conditions of commercial contracts. However, the government maintains ownership of the petroleum resources (as is the case in most countries), can intervene as representative of the Iranian public, and must consider the strategic role of petroleum in domestic economy. As such, there is the potential for the design of such contracts to prioritise public rather than private interests.⁴²³

In October 2013, the Ministry of Oil in Iran established the Iran Oil Contracts Restructuring Committee (ICCRC), chaired by Seyyed Mehdi Hosseini, former Deputy Oil Minister. Within the scope of the Committee's mandate was to assess the most suitable contract models to apply for the development of the nation's oil and gas fields and to increase the overall production capacity.⁴²⁴ Accordingly, this included an analysis of Iran's experiences of the Buy Back Contract model during the previous 20 years. The Committee also considered the feasibility of introducing a new contract model that better conforms with Iran's Constitution, social policies, the Petroleum Act and other relevant upstream legislations.⁴²⁵

5.3 Iranian Petroleum Contract and Iran's Constitution

⁴²⁰ Haddadi (n 423) 50.

⁴²¹ Ibid 47.

⁴²² Law to Modify Petroleum Law (2011) art 2.

⁴²³ Mostafavi Seyed Mostafa and Amani Masoud, 'The Nature of Petroleum International Contracts' (2008) 3(1) *Islamic Research Journal* 22.

⁴²⁴ Iran Business News, 'Oil Contracts Restructuring Committee (2015) <<u>https://www.iran-bn.com/tag/oil-contracts-restructuring-committee/</u>>.

⁴²⁵ Ardalan et al (n 26) 345.

The competitive nature of the global petroleum and gas sector has forced Iran to reform its petroleum contracts.⁴²⁶ Of particular concern is the emergence of new petroleum powers over recent decades and the provisions of new petroleum contract models by rival nations in the Middle East (e.g. Iraqi Service Contracts). Moreover, laws such as the FIPPA introduced by the Iranian government as part of the so-called "reconstruction" era (commencing in 1988) to encourage foreign companies to invest in the nation's oil and gas industries have to date been largely unsuccessful.⁴²⁷

Several aspects of Iran's post-revolution Constitution are related to international contracts and agreements such as the IPC. Principle 77 of the Constitution of Iran states that "International agreements, contracts and deals should be adopted (finalised) in the Parliament".⁴²⁸ Given the nature of the IPC and its objective to establish long-term relationships with international players, this principle prompted proponents of the IPC to argue that they should be able to be finalised without having to involve the Iranian Parliament.⁴²⁹

Opponents of the contract model refer to Principle 139 of the Constitution and its interpretation by the Guardian Council of the Constitution (see section 3.1.1). According to the Council's interpretation, any contract in which one party is a government Ministry or institution and the other party is a private foreign company should not be regarded as an international treaty and therefore was not covered by Principle 77 of the Constitution.⁴³⁰ This interpretation by the Guardian Council is used to argue that allowances for the government of Iran to engage in temporary transactions with foreign companies should not be extended to IPC. This is because the subject of the contract concerns Iran's national interests and any long-term petroleum contract concerning public assets should have its own specific legal and jurisprudential considerations.⁴³¹ The restrictions in the Constitution of Iran related to foreign investment is designed to avoid foreign domination of the Iranian economy. This is emphasised in the Constitutional Principles 43, 81 and 153 as well as in Shia jurisprudence.⁴³² Accordingly, opponents of the IPC are that the contract model violates the principle of jurisprudence denying the domination of Iranians by foreigners.

⁴²⁶ Haddadi (n 423) 50.

⁴²⁷ Yeganehshakib (n 230) 1.

⁴²⁸ Ali Reza Mazlumrahni and Mohsen Esmaeli. 'Legal Analysis and Economic Impact of Iran's Petroleum New Contracts (IPC)' (2018) 3(2) *Örgütsel Davranış Araştırmaları Dergisi* 7.

⁴²⁹ Ibid.

⁴³⁰ Ibid.

⁴³¹ Ibid.

⁴³² Ibid 8.

The position of Shia law in relation to these kinds of commercial agreement is contained in Article 45 of Iran's Constitution, 'Natural Resources as "Anfal". This article identifies the natural resources of Iran as *Anfal* or public wealth and property.⁴³³ The Islamic concept is identified in the Holy Koran where it states:

Anfal i.e., Public wealth and property, such as uncultivated or abandoned land, mineral deposits, seas, lakes, rivers and other public water- ways, mountains, valleys, forests, marshlands, natural forests, unenclosed pastureland, legacies without heirs, property of undetermined ownership, and public property recovered from usurpers, shall be at the disposal of the Islamic government for it to utilize in accordance with the public interest. Law will specify detailed procedures for the utilization of each of the foregoing items.⁴³⁴

Hence, the Holy Koran asserts that while *Anfal* belongs to God and the Prophet, it is left to the government to utilise it in the pursuit of public interest. Given that Shia Islamic teachings dictate that *Anfal* cannot be transferred or sold to another entity, significant barriers emerge in relation to Iran's scope for negotiating an agreement that may provide title to the oil to a private foreign entity.⁴³⁵

5.3.1 Redesigning the contracts

Underpinning the examination of the IPC model was its advantages and disadvantages to the petroleum sector in Iran and the potential for a new contract form to better attract foreign capital, technologies and know-how.⁴³⁶ In turn, the process to assess the current Buy Back Contract model and to develop an improved model saw the ICCRC engage in a number of processes including:

- evaluation of service contracts (partial or in full) from 33 oil producing countries,
- NIOC planning department preparing a report on its operational expectations,
- assessment of extensive research evidence produced both nationally and internationally on the implementation success of service contracts around the world, and

⁴³³ Nima Nasrollahi Shahri, 'The Petroleum Legal Framework of Iran: History, Trends and the Way Forward' (2010) 8(1) *China & Eurasia Forum Quarterly* 111, 120.

⁴³⁴ Holy Koran, Anfal 001.

⁴³⁵ Shahri (n 438) 120.

⁴³⁶ Shana News Agency, 'Who Are Iran's New Oil Managers?' 2019 <<u>https://en.shana.ir/news/286983/Who-Are-Iran-s-New-Oil-Managers</u>> (accessed 5 January 2020).

 analysis of the experts' and domestic / foreign companies' point of views regarding the Buy Back Contract form.⁴³⁷

Based on these considerations, the 10 core principles of the IPC may be summarised as follows:

- 1. Iran retains sole and total ownership of the reservoir.
- 2. Preserve Iran's national interest.
- 3. Due observance of the laws and regulations.
- 4. Win-win contract outcome.
- 5. Alignment in the benefits to both contracting parties.
- 6. Sustainability.
- 7. Partnership.
- 8. Operational efficiency.
- 9. Technology transfer.
- 10. Internationalisation.

Some of the 10 principles cited above have been extensively discussed, such as Iran's desire to maintain sole and total ownership of the oil reserves to protect its national interests and the rationale for entering into partnership agreements to improve operational efficiency and technology transfer. Some other principles, however, require further explanation. Internationalism for instance, refers to the process of regarding "contractual obligations as international obligations".⁴³⁸ As such, any breach of contract by the State (although not the OIC) is regarded as a breach of international law. Traditionally, all investment contracts are managed according to domestic laws, leaving the investor with little sense of protection and equally little recourse in terms of breaches against it. Internationalism attempts to address these concerns via international law which "recognises the absolute sanctity of the contract".⁴³⁹ In turn, any breach of contract by the State may be held open to sanction according to international law.

In terms of the principle of sustainability, this no doubt emerged in response to the heavy reliance of Iran on energy-intensive industries for domestic economic production and export.⁴⁴⁰ Iran is highly dependent on oil products to meet the nation's primary energy

⁴³⁷ Ardalan et al (n 26) 345.

 ⁴³⁸ Chin Leng Lim, Jean Ho and Martins Paparinskis, *International Investment Law and Arbitration: Commentary, Awards and other Materials* (Cambridge University Press, 2018) 37.
⁴³⁹ Ibid 39.

⁴⁴⁰ Morteza Sabetghadam, 'Energy and Sustainable Development in Iran' (Helio International, Sustainable Energy Watch, 2005/2006) 1.

demands and to develop its petrochemical and metal industries.⁴⁴¹ Moreover, the slow process to implement energy price reform along with low energy prices and poor energy efficiency combine to present "a serious threat to the economy of Iran".⁴⁴² According to Sabetghadam, this has increased the imperative for Iran to develop 'coordinated policies and implementation to create a more sustainable energy sector that supports the development and welfare of all Iran'.⁴⁴³

The partnership principle is also related to the development of Iran's energy resources sector via support for and improvement of the capabilities of Iranian domestic entities. Specifically, the IOC is required to gradually transfer the executive management positions to Iranian nationals "in order to facilitate the process and transfer of know-how and managerial skills to the Iranian entity".⁴⁴⁴ With the expectation that under the IPC the Iranian entity will play a larger role in the production activities, the principle of Iran's right of sovereignty and public ownership of all oil reserves is bolstered. Moreover, the principle related to the transfer of technology and know-how will help to ensure the transfer of such technology in Iran to improve managerial know-how and reservoir engineering skills.⁴⁴⁵

On 3 August 2016, the Council of Ministers in Iran passed a regulation governing the general conditions, structure and terms of upstream oil and gas contracts (the IPC Regulation). This represented a material step in securing approval from the Iranian government to the terms of the IPC and thus a major milestone in the nation's endeavour to attract foreign investment. Because the IPC is a risk service contract, the IOC carries much of the financial and operational risks related to the petroleum operations. Such costs are recovered by the contractor and a service fee is paid as compensation for carrying the risk only if petroleum operations are successful. Different to PSA, the IPC entitles the IOC to a fixed fee in lieu of a share of production. If there is agreement between the NIOC and IOC, the fixed fee may be paid by way of an allocation of oil produced.

5.3.2 Iranian Petroleum Contract fiscal parameters

There are several fiscal parameters of the IPC such as capital expenditures (CAPEX) including the company's major, long-term expenses (e.g. buildings, equipment,

⁴⁴¹ Ibid.

⁴⁴² Ibid 5.

⁴⁴³ Ibid 6.

 ⁴⁴⁴ Iran Group, 'The New Iranian Petroleum Contract – Government Approval' (*Herbert Smith Freehills*, 2016)
https://www.herbertsmithfreehills.com/latest-thinking/the-new-iranian-petroleum-contract-%E2%80%93-government-approval> (accessed 22 June 2020).
⁴⁴⁵ Ibid.

machinery), operating expenses (OPEX) including the company's day-to-day expenses (e.g. travel, taxes), Cost of Money (average interest rate for borrowed money), remuneration fees (payment or compensation for services or employment), the amortization period (length of time taken to pay off a loan) and the R-factor or cumulative revenues divided by cumulative costs.⁴⁴⁶ Of these, the main fiscal feature of the new contract model for upstream oil and gas production is the introduction of remuneration payments based on a negotiated fee per barrel of production. This provision was modelled on the fiscal model underpinning Iraqi petroleum contracts which had proved attractive to IOCs. As such, it was included by the Iranian authorities as a way to improve the contract framework to better meet the interests of the operator.⁴⁴⁷

The global energy sector can loosely be divided into the domains of supply and demand. The fiscal model underpinning the supply domain arguably has more complexities than its 'demand' counterpart. This is because of the (potentially significant) technical issues that must be managed during the exploration and production of a field, along with the uncertainty around the behaviours of the different suppliers in the market.⁴⁴⁸ This is not to mention the uncertainty around the geopolitical relations among nations in the Middle East as well as international tensions due to US sanctions on Iran, reintroduced in November 2018. Not surprisingly, geopolitical tensions and their potential to disturb supply (e.g. the tensions over oil supply between Saudi Arabia and Russia), combined with environmental catastrophes (e.g. the spread of COVID-19 virus) and advances in technology all make petroleum supply modelling and forecasting a complicated endeavour.⁴⁴⁹

The basic cost structure of the IPC is illustrated in Figure 5.1:



⁴⁴⁶ Sahebonar et al (n 423) 1.

⁴⁴⁷ Ibid.

⁴⁴⁸ Sahebonar et al (n 423) 1.

⁴⁴⁹ Ibid.

Figure 5.1 Basic costs structure of IPC

Figure 5.1 of the costs related to the petroleum operation shows a depreciation in costs from allocated revenue to petroleum costs. A fee is paid to the contractor in addition to the remuneration for operation and petroleum costs.⁴⁵⁰ All petroleum costs and the remuneration fee are recovered via 50% of revenue from the sale of crude oil from the field. A more targeted description and analysis of the different costs and remuneration fees are provided below:

5.3.2.1 Petroleum costs

In the IPC, the petroleum costs paid by the contractor are divided to four categories:⁴⁵¹

- Direct Capital Costs (DCC): paid during the exploration and development phases of the project to meet the development and production objectives
- Indirect Costs (IDC): paid during the development phase to Iranian governmental agencies. Such costs include Corporation Income Tax, Value Added Tax, Withholding Tax, customs duties and the like
- Cost of Money (COM): paid to compensate the contractor for costs to finance the project, but contrary to bank charges related to Buy Back Contracts, COM and are applicable to the IDC incurred prior to production and petroleum Costs and remuneration fees in relation to delayed or late payments. They are not related to DCC. The rate used to calculate COM is equal to LIBOR in addition to a premium.
- Operating Costs (OPEX): all costs related to the production phase as detailed in DP apart from COM, Capex and IDC.

5.3.2.2 Remuneration Fee

Remuneration fees are paid to the IOC in consideration of the risks and expenditures it 'adopts' during the exploration and development phases, as well as the provision of technology and know-how.⁴⁵² The remuneration fees paid to the IOC are based on the magnitude of risk they take on.

5.3.2.3 Cost recovery

In an IPC, cost recovery commences at the first year of production. The DCC and IDC paid by the IOC prior to production date, in addition to the COM as stipulated in the IPC shall

⁴⁵⁰ Elaheh Ghorbani, "Exploration, Development and Production of Iran's Fields and Reservoirs through the Iran Petroleum Contract (IPC)" (2020), *Journal of Energy & Natural Resources Law* 4.

 $^{^{451}}$ Sahebonar et al (n 423) 5.

⁴⁵² van Groenendaal and Mazraati (n 2) 3709.

depreciate over 5-7 years depending on the project demands and characteristics.⁴⁵³ The 7-year limitation period of the Buy Back Contracts was simply not enough time for the IOCs to procure an adequate return on their investment. This was, of course, remedied via the IPC model with contracts valid for a period of 20 years, with the opportunity to extend it to 25 years upon mutual agreement by the contracting parties.⁴⁵⁴ Upon first production, the IDC and OPEX (but not the DCC amortized over 5-7 years) are to be recovered by the IOC at cost without COM.⁴⁵⁵ Hence, when the cost structures are considered, the IPC presents as a more favourable contract model to international investors / IOCs compared to the Buy Back Contract.

5.3.3 Investment, technology and exploration benefits

The final decision making around the current IPC model was to ensure that it remains "in accordance with the general terms of the most widely used oil and gas production contracts".⁴⁵⁶ Given this context, this thesis sought to answer key questions around the strengths and weaknesses of the IPC (RQ1), the impact of previous contract models (i.e. the Buy Back Contract) on the international petroleum trade (RQ2), and the extent to which the IPC best addresses the development needs of Iran's oil resources sector (RQ3). The IPC model designed by the Ministry of Petroleum of the Eleventh Government of Islamic Republic of Iran aimed to make the contracts more attractive to foreign companies.⁴⁵⁷ The development of the new model is in response to the Iranian economy's ongoing dependence on petroleum and the reality that such resources will one day be exhausted. In turn, the model design has the objective at least to support a long term and win-win relationship between the NIOC and the international petroleum companies.⁴⁵⁸ The characteristic features of the IPC model are its attempt to optimise Buy Back contracts, transition toward global standards in petroleum contracts, attract foreign investors, and create a better balance between risk and reward.⁴⁵⁹

On 30 September 2015, the Parliament of Iran passed the Iran Petroleum Contract (IPC) Bill whereby greater scope is considered for transferring the production and operation of

⁴⁵³ Sahebonar et al (n 423) 4.

 ⁴⁵⁴ Priyanka Prakash, 'Iran Opens Up: New Goals, New Deals' (Centre for Air Power Studies, 2017) 1.
⁴⁵⁵ Ibid.

⁴⁵⁶ Bijan Khajehpour, 'Will Iran Attract International Oil Firms in Post-Sanctions Era?' (2016a) *al-Monitor* <<u>https://www.al-monitor.com/pulse/ar/originals/2016/11/iran-petroleum-contract-ipc-total-cnpc-azadegan.amp html?skipWem=1> (accessed 5 February 2020).</u>

⁴⁵⁷ Mazlumrahni and Esmaeil (n 433) 1.

⁴⁵⁸ Seyed Hassan Mousavi, A Legal Analysis of Service Contract in Oil and Gas Sector in Iran focusing on IPC (GanjeDanesh, 2017) 22.

⁴⁵⁹ Mazlumrahni and Esmaeil (n 433) 4.

petroleum reserves to foreign companies. This is contrary to Article 2 of the Constitution of Iran which, as previously established, limits the extraction and production of crude oil and gas to government sector and bans the transfer of operations to private sector companies irrespective of whether they are domestic or foreign. Nonetheless, the rationale for increasing foreign control over production and operations in IPCs is related to the perception in Iran that it is unable to adequately sustain production of petroleum fields due to lack of investment and technology.⁴⁶⁰

5.3.3.1 Investment

Reports from the Iranian Parliament suggest that the NIOC gets 14.5% of the revenue from crude oil and condensate exports.⁴⁶¹ Iran's refining capacity was more than 2 million barrels per day in 2017 on the basis of domestic allocations. In addition, NIOC exports of crude oil and condensate in 2018 were around 2.5 million barrels per day, combined with domestic sales (discounted by 5%) of around 100,000 barrels per day.⁴⁶² This calculates to revenue of around \$7.6 billion to the NIOC based on an average oil price of \$55 per barrel throughout 2018. However, throughout 2018 it was expected that the NIOC would have expenses of up to \$29 billion.⁴⁶³

According to the *Law of the Sixth Five-Year Law of the First Five-Year Economic, Social and Cultural Development Plan* in Iran, the goal was to have the NIOC produce 4.7 million barrels per day of crude oil and 950,000 barrels per day of condensate by the end of 2020. To achieve such Development Plan goals, NIOC Director of Organisational Planning, Mohammad Delparish announced in 2015 that the NIOC would need in excess of \$65 billion.⁴⁶⁴ However, there are several obstacles to attracting foreign investment for the development of the oil and gas fields and to thus achieving such outcomes. They include the perceived unattractiveness of the new IPC by foreign investors, issues around transparency in Iran's banking system, lack of international credit ratings for Iranian companies, currency exchange rate fluctuations, and international sanctions.⁴⁶⁵

5.3.3.2 Technology

The reasons for Iran's need to access advance foreign technology and know-how are several. First and foremost, such technological advancements were benefitting the oil producing

465 Ibid.

⁴⁶⁰ Ibid 5.

⁴⁶¹ Chow et al (n 22) 4.

⁴⁶² Ibid.

⁴⁶³ Ibid.

⁴⁶⁴ Ibid.

neighbouring countries of Iran, as well as the international petroleum industry more broadly.⁴⁶⁶ There is little question that the political conviction in Iran to rely on domestic petroleum production capabilities for the management of the oil sector in order to satisfy the Constitution constrained the growth of the sector.⁴⁶⁷ It should be acknowledged that the NIOC maintained adequate crude oil production levels although confronted with several challenges and deficiencies.⁴⁶⁸ However, the upstream oil sector infrastructures in Iran were more than 40 years old on average towards the end of the 20th century, with the national pipeline network more than 35 years old, and most of the oil refineries underperforming as a result of inadequate equipment upgrades.⁴⁶⁹ Furthermore, more than 80% of the oil produce in Iran is from a select few mature fields which require significant improvements in their recovery technology. A key issue for the majority of oil fields in Iran is that they have very low recovery rates; rarely above 20-22% for the top five fields.⁴⁷⁰ As such, it is reasonable to assert that the constraints and limitations characterising the oil sector in Iran at this time may have been significantly remedied if advanced technologies international practices were harnessed more constructively after the Islamic Revolution. At present, the NISOC and IOOC⁴⁷¹, two subsidiaries of the NIOC, are responsible for more than 90% of the petroleum production operation in Iran.472

5.3.3.3 IPC role in future potential for increased oil production The NIOC has the potential to increase crude oil production in Iran in two key areas. First, by capitalising on undeveloped and under-developed areas including the West Karun River (Arvandan) and the giant Azadegan field.⁴⁷³ Second, by accessing the more technicallychallenging reservoirs (e.g. the deep-level Bangestan Group of formations in the Khuzestan Province) that demand the use of advanced drilling and stimulation methods to increase the rates of recovery.⁴⁷⁴ In terms of the development of the Arvandan Region, the bulk of Iran's untapped oil is situated in a number of giant and supergiant fields located in Khuzestan

⁴⁶⁶ Ghorbani (n 455) 3.

⁴⁶⁷ Ibid.

⁴⁶⁸ Ibid.

⁴⁶⁹ Chow et al (n 22) 5.

⁴⁷⁰ Ibid.

⁴⁷¹ National Iranian South Oil Company and Iranian Offshore Oil Company.

⁴⁷² Chow et al (n 22) 5.

⁴⁷³ This supergiant field is the largest untapped oil field in the world

⁴⁷⁴ The Ahwaz oil field is one such field. Most wells drilled in this field produce from the shallower Asmari formation, with currently less than 10% of production from the Bangestan formation. Production could potentially increase by 25% (around 220,000 barrels per day extra) by drilling the Bangestan reservoir. Plans to develop the Ahwaz oil field have never come to fruition, however, despite initial interest from BP.

Province, next to the border with Iraq.⁴⁷⁵ It is reported that the NIOC has detected 12 oil fields across the region,⁴⁷⁶ 5 of which are shared with Iraq. These fields have been identified by the NIOC as being of top priority for development.⁴⁷⁷ Moreover, it is estimated by the Arvandan Oil & Gas Company that the shared border oil fields⁴⁷⁸ contain more than 64 billion barrels of oil. In turn, the aim of the NIOC is to have production from these fields of 700 thousand barrels per day by 2022. To date, around 260,000 thousand barrels per day have been added to overall NIOC production following the completion of the first development phase in North Azadegan, North Yaran, and Yadavaran added. However, the development of these fields has been subjected to long delays.⁴⁷⁹

For instance, exploratory drilling commenced in 1976 in Azadegan but exploration in the region was halted due to political unrest in 1978, mass demonstrations and strikes by NIOC worker, and the 1979 Islamic revolution. Throughout the 1980s Iran was embroiled in a war with Iraq, with the Azadegan field being the location of a battle ground. Following the conclusion of the Iran-Iraq War, the risk associated with developing the border oil fields remained high as a result of issues around demining and clearing unexploded ordnance from the region.⁴⁸⁰ The field in South Azadegan, categorised as a greenfield⁴⁸¹ under the IPC, is estimated to contain around 25 billion barrels of heavy oil. The production target for the South Azadegan field is 600,000 barrels per day with an expected investment requirement of around US\$4 billion. In turn, Memorandums of Understanding (MOU) to explore and develop the supergiant field signed by Shell, Petronas, Total, and Sinopec and accepted by the NIOC have been scuppered by the new round of sanctions imposed on Iran by the US.

5.3.4 Foreign interest in Iranian Petroleum Contracts

Early interest from IOCs in investing in Iran's energy sector via the new IPC model was promising for the nation's ambitions. Following the announcement by Iran that it intended to restructure its energy sector contract framework in 2015, the initial interest from international companies was promising.⁴⁸² With more than 28 signed MoUs accepted by the NIOC for joint operations between Iranian domestic companies and IOCs, it appeared that Iran's energy

⁴⁷⁵ Chow et al (n 22) 8.

⁴⁷⁶ Azadegan, Darquain, Sohrab, Yadavaran, Yaran, Omid, Band Karkheh, Moshtagh, Khorramshahr, Susangerd, Arvand and Jufeyr.

⁴⁷⁷ Chow et al (n 22) 8.

⁴⁷⁸ North and South Azadegan, North and South Yaran, and Yadavarn.

⁴⁷⁹ Chow et al (n 22) 8.

⁴⁸⁰ Mahdavi (n 218) 243.

⁴⁸¹ 50,000 barrels per day produced in 2016.

⁴⁸² Prakash (n 459) 1.
sector would have access to the billions of dollars of investment and the transfer of technology and expertise it was seeking.⁴⁸³

The only MoU to transition into a contract is the multi-billion-dollar contract agreed to by the NIOC and the consortium comprising French Total (51% share), the China National Petroleum Corporation (CNPC, 30% share) and Iranian Petropars 19% share) in July 2017 to develop South Pars Phase 11.⁴⁸⁴ This represents the first significant investment from a foreign investor company in Iran after the JCPOA agreement, with the project cost estimated to total 4.8 billion euros. With the contract signed for 20 years, this deal marks the first return of French Total to Iran since the 1990s.⁴⁸⁵

Notwithstanding extensive exploratory drilling to determine the geologic extent of the Azadegan fields, the full petroleum system in Azadegan remains unclear due to a lack of adequate exploration studies.⁴⁸⁶ Although production levels could be maintained at around 400,000 barrels per day for the next two decades with the technology currently available, foreign expertise and technology are needed to support sustained rates of production in newly drilled wells.⁴⁸⁷ According to the NIOC, the priority during the first rounds of IPC negotiations with IOCs is to fast-track the development of its shared oil fields.⁴⁸⁸ In turn, this is confirmed with the only two contracts with international partners to have been signed by the NIOC to date. What is clearly evident, however, is that Iran has largely failed to attract the foreign investment it needs under the IPC. The Trump Administration's withdrawal from the JCPOA on the Iranian nuclear program and the subsequent reinforcement of sanctions has unquestionably and significantly impacted Iran's capacity to attract foreign investment. However, so too has the unattractiveness of the terms and conditions of the IPC.⁴⁸⁹ The domestic exploration and production companies listed as 'partner options' for IOCs wishing to invest include, but is not limited to, eight firms including the IRGC conglomerate KAA and a subsidiary of the NIOC, the Industrial Development and Renovation Organisation

⁴⁸⁹ Chow et al (n 22) 7.

⁴⁸³ Ibid.

⁴⁸⁴ National Iranian Oil Company, 'Iran, Frances Total Sign Deal to Develop South Pars Phase 11', June 3, 2017 <http://www.nioc.ir/portal/Home/ShowPage.aspx?Object=NEWS&ID=c5ae788e-4c87-4e10a41bf64946f6cc08&LayoutID=dfcdcb2a-d23c-4d84-a9ad5104719e9d10&CategoryID=65c76df2-8d90-4f41-

⁹b18- 8b22a277f981> (accessed 20 March 2020).

⁴⁸⁵ Ibid.

 ⁴⁸⁶ Richard Devine and Ida Mokhtassi, 'Iran's Council of Ministers ratifies IPC Terms' (Clyde & Co, 2016) 1.
 ⁴⁸⁷ Ibid.

⁴⁸⁸ Iran shares 19 oil and gas fields with neighbours: offshore fields with Saudi Arabia, UAE, Kuwait, Qatar and Oman; and onshore fields with Iraq.

and the SETAD, a government body controlled by the Supreme Leader of Iran mandated to manage these activities.⁴⁹⁰

The list of countries certified to bid for oil and natural gas tenders to be part of a JVA (see Table 5.1) gives an indication of the relationship between the IPCs and the shifting foreign policy position of Iran. The diversity of international oil and gas companies and their countries of origin, along with the marketing expertise they bring with them, were regarded as advantageous by Iran in its aim to build a more resistant and self-reliant economy.⁴⁹¹ Iran had experienced the consequential effects of Western sanctions of its energy industry and appeared reluctant to become too dependent on IOCs operating out of the US and Europe particularly.⁴⁹² The introduction of IPCs, combined with the strategy to engage diverse types of IOCs thus reflects Iran's attempt to re-engage with the international market to strengthen and improve its domestic economy.⁴⁹³

CNOOC (China)	Korean Gas Corp. (Sth.	PGNiG (Poland)
	Korea)	
CNCP (China)	Lukoil (Russia)	Plus Petrol (Argentina)
CNPW (China)	Maersk (Denmark)	Posco Daewoo (Sth. Korea)
DNO (Norway)	Mitsubishi (Japan)	PTTEP (Thailand)
Eni (Italy)	OMV (Austria)	Schlimberger
		(Netherlands/US)
Inpex (Japan)	ONGC Videsh (India)	Shell (Netherlands/UK)
ITOCHU (Japan)	Pereneco (France/UK)	Sinopec (China)
Gazprom (Russia)	Pertamina (Indonesia)	TOTAL (France)

Table 5.1 IOCs with certification to bid at Iran's oil and natural gas tenders, 2017

Source: David Jalilvand and Friedrich Ebert Foundation494

⁴⁹⁰ Ibid 21.

⁴⁹¹ Shahdani Mahdi Sadeghi and Mahdi Pakzat Seyed, 'Oil Futures Contracts, Obligation to Selling or to Daily Settlement?' (2018) 6(1) *Journal of Asset Management and Finance*, 181.

⁴⁹² Ibid.

⁴⁹³ David Ramin Jalilvand, 'Iran's Gas Exports: Can Past Failure become Future Success?' (The Oxford Institute for Energy Studies, 2013) 7.

⁴⁹⁴ David Jalilvand and Friedrich Ebert Foundation, 'Iranian Energy: A Comeback with Hurdles' (The Oxford Institute for Energy Studies, 2017) 1 <<u>https://www.oxfordenergy.org/wpcms/wp-</u>content/uploads/2017/01/Iranian-Energy-a-comeback-with-hurdles.pdf> (accessed 15 February 2020).

5.3.4.1 Tender model

Of significance is the evaluation criteria applied to determine successful bidders. The fee per barrel is linked to the market and so a primary selection criterion is the IOC's market price percentage "bid" as the base fee per barrel.⁴⁹⁵ The "field risk" adjustment factor as part of the overall fee structure is generally pre-determined. Moreover, the production plan is agreed to by both parties within the broader scope of the Development Plan. As such, it remains unclear as to the bases upon which the bidders differentiate themselves. For example, the fee per barrel and production targets are considered critical tender metrics in the Iraqi contract model.⁴⁹⁶ It is unlikely that the NIOC applies only technical evaluations when selecting successful tenders, and so other specific commercial criteria must also be bid upon. These criteria may include the flexibility to offer an increase or decrease in the fee adjustments relating to each "R" index step change, or an agreement to finance commitments to "Minimum Obligations" in the exploration stage.⁴⁹⁷ The evaluation criteria will naturally differ depending on the prospects of each project (i.e. greenfield or brownfield), but the exact nature of the criteria differences is not explicit in the IPC. It may be that some smaller IPCs do not make their tender public (with the consent of different governing bodies). Furthermore, it is not clearly apparent as to whether there are grandfathering clauses or alternative treatment for those concessions already agreed to under the Buy Back Contract model.498

5.3.4.2 Iranian Petroleum Contracts and the Joint Comprehensive Plan of Action It has been well-established in this thesis that there are many challenges associated with the formulation of petroleum contracts. As such, they are invariably the source of disputes between the negotiating parties irrespective of the prevailing geopolitical tensions in the region. Proponents of the IPC model argue that it can play a vital role in Iran's interactions on the international stage and the development of its resources, despite whatever weaknesses they may perceive the contract model to have.⁴⁹⁹ The key element in the JCPOA is the Iranian government's agreement to limit the capacity or close the nuclear facilities in the

⁴⁹⁵ Moslem Alimohammadlou and Abbas Bonyani, Iran's Energy Policy After the Nuclear Deal for Cooperation with Foreign Oil and Gas Companies' (2019) 12(2) *International Journal of Procurement Management* 199, 199.

⁴⁹⁶ Ibid.

⁴⁹⁷ Dallas and Black (n 369) 7.

⁴⁹⁸ Ibid.

⁴⁹⁹ Amir Ahmadi, 'The Impact of Economic Sanctions and the JCPOA on Energy Sector of Iran' (2018) 13(5) *Global Trade and Customs Journal* 198, 199.

country in return for the lifting of sanctions.⁵⁰⁰ In this post-sanction environment, the government of Iran saw an opportunity to attract IOCs back into the country. To do this, however, the Iranian government believed that the petroleum contract model needed to be more attractive to the interests of foreign entities.⁵⁰¹ The IPC model was also seen by the government as a way to build stronger connections between the economy in Iran and the world economy. That is, they could help to establish stronger ties between the banks in Iran and major foreign banks by encouraging the foreign entities to invest in Iran's energy sector.⁵⁰²

5.4 United States Sanctions

All of the potential investment, technology and exploration benefits potentially accessed via the IPC must of course be considered in the context of the newly reimposed US sanctions. As alluded to earlier, a key reason why the NIOC was unable to attract significant foreign investment through the IPC upon the implementation of JCPOA was the US sanctions imposed on Iran and the associated risk of sanctions snap back for foreign entities.⁵⁰³ Certainly, there were several investment opportunities within different section of the petroleum value chain announced by the NIOC. Examples of such projects include oil field development plans, oil exploration blocks, and crude oil storage facilities, pipelines and pump stations.⁵⁰⁴ On 8 May 2018, the Trump Administration announced the US withdrawal from the JCPOA, simultaneously announcing the reimposition of all US sanctions either lifted or waived as part of the JCPOA⁵⁰⁵. This included reimposing the:

- National Defense Authorization Act (NDAA) 2012 s 1245,
- sanctions under the Iran Freedom and Counter-proliferation Act (IFCA) 2012,
- sanctions under the Iran Sanctions Act (ISA) 1996, and
- sanctions under the Iran Threat Reduction and Syria Human Rights Act (ITRSHRA) 2012.

504 Ibid.

⁵⁰⁰ Ibid.

⁵⁰¹ Mazlumrahni and Esmaeil (n 433) 7.

⁵⁰² Ibid.

⁵⁰³ Paul Rivlin, 'Leverage of Economic Sanctions: The Case of US Sanctions Against Iran, 1979–2016' in Mikael Wigell, Sören Scholvin and Mika Aaltola (eds) *Geo-economics and Power Politics in the 21st Century* (Routledge, 2018) 99.

⁵⁰⁵ National Security Presidential Memorandum/NSPM-11, https://fas.org/irp/offdocs/nspm/nspm-11.pdf>.

Agencies and departments in the US implemented the sanctions based on 90-day and 180day wind down periods. Following the designated period, the relevant sanction would be placed back into full effect. For example, at the conclusion of the first 90-day period on 7 August 2018, sanctions were reimposed on:

- Iran's automotive sector,
- undertakings linked to the issuance of sovereign debt,
- Iranian rial-related transactions,
- Iranian gold and precious metals trade,
- graphite, aluminium, steel, coal, and software used in industrial processes, and
- Iranian government US bank note acquisitions.⁵⁰⁶

Around this time the Trump Administration re-stated its intention to reimpose sanctions on:

- Iran's energy, shipping, and shipbuilding sectors along with port operators,
- petroleum-related transactions by Iran, and
- transactions by foreign financial institutions with the Central Bank of Iran.⁵⁰⁷

In terms of snap back, the Trump Administration indicated that it intended to enforce the sanctions fully, warning that any country which did not wind down its trade activities with Iran would risk severe consequences.⁵⁰⁸ Regarding the petroleum sector in Iran and the attractiveness of the IPC, the sanction by the US to have had the most significant impact is arguably *NDAA 2012* s 1245. This gives authority to the US Secretary of the Treasury to work with the US Secretary of State to impose sanctions on foreign financial institutions. Specifically, the section prohibits foreign financial institutions from participating in the US financial system if they are found to conduct transactions with the Central Bank of Iran.⁵⁰⁹ There are exemptions for some countries importing crude oil under section 1245 (d)(4)(D) of the Act if the President accepts that the country has "significantly reduced" its purchasing volumes, but exactly what constitutes "significantly reduced" has not been well-clarified by the US State Department.⁵¹⁰ The sanctions can include a prohibition on, or the placement of

⁵⁰⁶ Rivlin (n 508) 101.

⁵⁰⁷ United States Treasury, 'Executive Order 13846: Reimposing Certain Sanctions with Respect to Iran' <<u>https://www.treasury.gov/resourcecenter/sanctions/Programs/Documents/08062018 iran eo.pdf</u>> (accessed 3 May 2020).

⁵⁰⁸ Andrea Shalal, 'Pompeo Warns Russia, China Against Ignoring Move to Reimpose U.N. Sanctions on Iran' *Reuters* (August 20, 2020) https://www.reuters.com/article/us-usa-iran-un-sanctions/pompeo-warns-russia-china-against-ignoring-move-to-reimpose-un-sanctions-on-iran-idUSKCN25F2S4> (accessed 7 September 2020).

⁵⁰⁹ Chow et al (n 22) 11.

⁵¹⁰ Ibid.

strict conditions around, companies opening accounts with foreign financial institutions that knowingly conduct or facilitate financial transactions with the Central Bank of Iran or other designated financial institution in Iran.⁵¹¹ Such sanctions significantly impacted the ability of Iran to export petroleum or petroleum products. Moreover, sanctions may be imposed on foreign financial institutions by the US Secretary of the Treasury if said institution knowingly conducts or supports significant financial transaction with the NIOC.

Section 312 of the ITRSHRA had declared the NIOC to be an agent of the IRGC according to Executive Order 13846. When all of these actions by the US are combined they had the effect of placing severe restrictions and constraints arounds Iran's access to foreign investment and technology.⁵¹² Moreover, it put a significant dent in the volume of petroleum exports out of Iran and thus the amount of revenue being injected back into the Iranian economy. Such conditions and outcomes presented significant challenges to the NIOC's capacity to achieve its development goals. It is important to point out however that the most recent round of sanctions placed on Iran by the US have been done so unilaterally and without the support of some US allies (e.g. Germany and France).⁵¹³ In terms of the petroleum sector in Iran, the geo-political implication of the reimposition of severe and lasting sanctions on Iran may be a stronger presence of China and Russia in the exploration and development of Iranian petroleum. These countries, who are arguably more prepared to work around the US sanctions if it is in their best interests to do so,⁵¹⁴ will in turn be more likely to be offered lucrative contract terms to attract their interest and involvement should such interest fade from Western companies (e.g. Total).⁵¹⁵ Indeed, 'China is actively creating financial channels for purchasing petroleum from Iran that circumvent the sanctions imposed by the US'.⁵¹⁶ Notwithstanding these movements from some countries, the reality is that the current round of US imposed sanctions presents a major challenge for Iran to get its oil to the international market and hence to inject much needed capital into the oil and gas sectors. Fulfilment of the Law of the Sixth Five-Year Law of the First Five-Year Economic, Social and Cultural Development Plan is reliant on a significant increase in revenues from oil and gas sales. In

⁵¹¹ NDAA 2012 s 1245.

<<u>https://www.treasury.gov/resourcecenter/sanctions/Programs/Documents/ndaa_publaw.pdf</u>> (accessed 18 December 2019).

⁵¹² Chow et al (n 22) 9.

⁵¹³ Ibid.

⁵¹⁴ Shalal (n 513).

⁵¹⁵ Ibid

⁵¹⁶ Chow et al (n 22) 9.

turn, the sanctions have unquestionably presented significant logistical challenges to Iranian petroleum exports, led to a significant reduction in volume, as well as loss of production. Although the US does not import crude oil from Iran, the formidable range of sanctions it has imposed have severely restricted the extent to which countries – both allies and adversaries of the US, conduct business with Iran.⁵¹⁷

It is worth mentioning that many countries to import oil from Iran are reluctant to participate in sanctioning the nation. Indeed, the European Union, Germany, Canada and other US allies, along with China and Russia, have made clear their intentions to not withdraw from the JCPOA.⁵¹⁸ Hence, Chow et al. have argued that the unilateral decision by the US to reimpose sanctions on Iran "creates the perverse impetus for some countries to come to the aid of Iran" by engaging in initiatives to try to save the JCPOA.⁵¹⁹ Although such initiatives have proved largely to be symbolic, with limited protection provided to the energy resources sector in Iran, they arguably have some positive implications for Iran IPC scheme. To clarify, the sanctions draw the focus of the international community onto Iran's resources sector and on ways to deal with Iran to potentially create trade opportunities once the sanctions have been lifted.⁵²⁰ Furthermore, it has arguably strengthened Turkey's trade relationship with Iran. As a major importer of crude oil from Iran, Turkey has clearly indicated that it intends to keep such trade agreements with Iran in place. In addition, a significant volume of crude oil exports from Iran now go to China and India, which was not the case during the previous round of sanctions.⁵²¹ This has the potential to significantly reduce the effect of US sanctions on the Iranian oil and gas sectors as the sanctions arguably rely on the support from China and India for their impact.⁵²² Nonetheless, the power and the reach of the US Administration to significantly reduce Iran's capacity to develop its resources sector over the long term should not be underestimated.

5.5 Comparing Buy Back Contracts to Iranian Petroleum Contracts

The central focus of this thesis is the extent to which the IPC presents as a better contract model than the Buy Back model for the NIOC as a contracting party and for the development

⁵¹⁷ Ibid.

⁵¹⁸ Tarja, Cronberg. 'No EU, No Iran Deal: The EU's Choice Between Multilateralism and the Transatlantic link' (2017) 24(3-4) *The Nonproliferation Review* 243, 244.

⁵¹⁹ Chow et al (n 22) 11.

⁵²⁰ Ghorbani (n 455) 4.

⁵²¹ Ibid.

⁵²² Reuters (n 27).

of Iran's oil and gas sectors more broadly. The primary objective for developing the IPC was to address and improve upon the flaws in the Buy Back Contract form. In total, the Buy Back Contracts underwent three revisions in around 20 years in an attempt to make them more appealing to foreign investors. However, the short-term contract periods combined with the failure of the Constitution in Iran to recognise reserve booking rights for IOCs, thwarted the potential for the success of the Buy Back Contracts.⁵²³ The petroleum projects completed or in operation under the Buy Back scheme contributed total production of 700,000 barrels per day oil.⁵²⁴

Comparisons can be made between key elements of the IPC and Buy Back Contract models in relation to the insertion of contractual considerations, the development of contractual risks and how to respond to such risks. In turn, one of the main complaints directed towards the Buy Back Contract model was its lack of flexibility for the IOC.⁵²⁵ That is, the Buy Back Contract involved the preparation of a Master Development Plan by the contracting parties. The parties also agreed on an investment ceiling, with the cost-recovery period negotiated by the parties along with an agreement on a fixed ROR.⁵²⁶ The Capex was paid by the IOC who was also responsible for the development of the field. The costs outlaid via Capex was regarded as a 'loan' to the Iranian state, with annuity payments subsequently made from the commencement of production, or when the production target was achieved, until the end of the contract term. These payments were designed to allow the IOC to recover capital expenditures, operating costs and finance charges accumulated during the development phase.⁵²⁷ Alternatively, the IOC may agree to payment in kind. The development phase of the contract was typically 2-4 years and the production phase was generally around 5-10 years duration. While it is generally regarded that the Buy Back Contract was not the optimal form of technical-service contract, the ROR on investment of around 15-18% on investment was appealing enough to attract numerous IOCs to invest in the development of Iran's oil fields during the 1990s and early 2000s.⁵²⁸

Furthermore, the unclear provisions around contract extensions in the Buy Back Contract model left the IOC at greater exposure to risk of not having a request for a time extension

⁵²³ Chengyuan Li, Mingjun Jiang, Haiming Ge, Zhen Li and Dongkun Luo, 'An Operational Risk Analysis of Iran Buyback Contract and its Policy Implication' (2017) 16 *Energy Strategy Reviews* 43, 43.

⁵²⁴ Chow et al (n 22) 7.

⁵²⁵ Li et al (n 528) 43.

⁵²⁶ Ibid 44.

⁵²⁷ Ibid.

⁵²⁸ Ebrahmi and Khouzani (n 232) 150.

granted and thus incurring damages due to delay. This added to the risk for the IOC given the list of events cited as permissible for relieving the IOC from its requirement to complete the operations by the deadline may be limited to such things as failure to provide site possession, undertaking additional or extra activities, and special circumstances beyond the control of the contractor. In contrast, IPCs which include appropriately drafted clauses regarding extension of time provisions can be advantageous to both the IOC and NIOC (or affiliate) in relation to respecting the contracted time to complete works, protecting the rights of the IOC regarding delays caused by the host nation, and preserving the host nation's rights to liquidated damages.⁵²⁹ Moreover, extension of time provisions in exploration and production contracts can provide IOCs with access to more exploration areas and additional time to produce from the fields.

Hence, the IPC is much more flexible regarding the exploration phase in general and the integration of the exploration and development operations for new fields. Such flexibility is evident in the IPC through its allowance of the IOC to explore nearby fields in the case that the originally agreed-to field is non-productive. Under the terms of the Buy Back Contact, the IOC would be obligated to stay with the development plan already agreed-to.⁵³⁰ In addition, it offers greater flexibility around oil recovery from brown fields – improved oil recovery (IOR) and enhanced oil recovery (EOR) – and the development of the common fields with neighbouring countries.⁵³¹ Lastly, the IPC includes more flexible provisions around the exploration, development and production and IOR/EOR, as well as in the development of green fields.⁵³² Hence, a key defining feature of the IPC is the integration of exploration, development and production operations. The flexibilities of the IPC compared to the Buy Back Contract are:

- 1. Flexible Development Plan.
- 2. Work to annual work program and budget rather than to fixed caped costs.
- 3. Full cost recovery for IOC.
- 4. Metter balance between risk and reward.
- 5. Flexible reward structure that considers changes in the price of oil.

⁵²⁹ Aline El Sayed, 'Contractors' Right to Clearer Provisions for Time Extension' (2015) http://www.qatarconstructionnews.com/contractors-right-clearer-provisions-time-extension/ (accessed 12 December 2019).

⁵³⁰ Prakash (n 459) 1.

⁵³¹ Sadeghi and Seyed (n 496) 181.

⁵³² Ibid.

- 6. Flexible fee structure based on the level of risk for different areas.
- 7. Cost Saving Index (CSI).
- 8. More participative decision-making process.
- Opportunity for further exploration operations in neighbouring fields in case of failure.
- 10. Flexibility in long-term cooperation in case of EOR.

To provide a comprehensive discussion of the advantages and disadvantages of IPCs compared to Buy Back Contracts from Iran's perspective, it is necessary to identity their key differences. Table 5.2 presents a comparison of the two contract models including such elements as contract model objective, contract duration, the role and rights of the contractor and others.

Buy Back Contract	IPC
Strategic objective is to preserve the	Strategic objective is to promote IOCs to
sovereignty of the Iranian government over	use the latest technologies in oil reserve
the nation's natural resources	management and petroleum production in
	order to build capacity of the domestic
	resources sector
Contractor acts as operator during the	Favours JVA between Iran (typically NIOC
exploration and development phases, with	or an affiliate) and the IOC to better access
operatorship during the production phase	and exploit their technologies.
assigned to the NIOC.	
Positions the Iranian party more as a	Positions the Iranian party more as a
contractor.	'technical' partner.
Comprises four phases: exploration,	Comprises four phases: exploration,
development, production and	development, production and
enhanced/improved recovery	enhanced/improved recovery
Term of contract typically around 5-7 years	Term of contract generally between 20-25
(2-3-year development period; 4-7-year	years providing foreign investors with more
remuneration).	certainty and investment incentive.

Table 5.2 Comparison of the Buy Back Contract and IPC models

Reflects the Iranian Constitution stating	Includes provisions for the transfer of
ownership of natural resources belongs to	ownership of hydrocarbons to foreign
the "nation" and cannot be transferred.	partners at specified delivery points.
	Ownership rights are not however extended
	to project assets.
Foreign company afforded limited reward	Foreign company rewarded for carrying
for carrying risks. No incentives provided.	risks as 'manager' of the project.
Recoverable capital costs by the contractor	Recoverable costs based on an annual work
are capped (costs are an estimation of	programme and the budget approved by the
project development expenditures). A cap	contracting parties. The NIOC is given final
increase for additional works must be	right of approval.
approved by the NIOC	
Contractor receives a pre-agreed rate of	Favours the development of 'joint fields
return from production revenue allocated by	(NIOC and IOC operated) and higher
the NIOC according to a long-term export	remuneration rates to incentivise
sales agreement	investment.
sales agreement Remuneration fees can be unilaterally	investment. Field development plan includes a minimum
sales agreementRemuneration fees can be unilaterallyreduced by the NIOC if production falls	investment. Field development plan includes a minimum production target approved by the NIOC.
sales agreementRemuneration fees can be unilaterallyreduced by the NIOC if production fallsbelow the level stated in the field	investment. Field development plan includes a minimum production target approved by the NIOC. However, the consequences of not meeting
sales agreement Remuneration fees can be unilaterally reduced by the NIOC if production falls below the level stated in the field development plan.	investment. Field development plan includes a minimum production target approved by the NIOC. However, the consequences of not meeting the targets are not stipulated.
sales agreement Remuneration fees can be unilaterally reduced by the NIOC if production falls below the level stated in the field development plan. Contracting party disputes to be resolved	investment. Field development plan includes a minimum production target approved by the NIOC. However, the consequences of not meeting the targets are not stipulated. Contracting party disputes to also be
sales agreement Remuneration fees can be unilaterally reduced by the NIOC if production falls below the level stated in the field development plan. Contracting party disputes to be resolved via escalation, with arbitration as the final	investment. Field development plan includes a minimum production target approved by the NIOC. However, the consequences of not meeting the targets are not stipulated. Contracting party disputes to also be resolved via escalation, with arbitration as
sales agreement Remuneration fees can be unilaterally reduced by the NIOC if production falls below the level stated in the field development plan. Contracting party disputes to be resolved via escalation, with arbitration as the final method.	investment. Field development plan includes a minimum production target approved by the NIOC. However, the consequences of not meeting the targets are not stipulated. Contracting party disputes to also be resolved via escalation, with arbitration as the final method.
sales agreement Remuneration fees can be unilaterally reduced by the NIOC if production falls below the level stated in the field development plan. Contracting party disputes to be resolved via escalation, with arbitration as the final method. No incentives provided to undertake	investment. Field development plan includes a minimum production target approved by the NIOC. However, the consequences of not meeting the targets are not stipulated. Contracting party disputes to also be resolved via escalation, with arbitration as the final method. Incentives offered to IOCs for undertaking
sales agreement Remuneration fees can be unilaterally reduced by the NIOC if production falls below the level stated in the field development plan. Contracting party disputes to be resolved via escalation, with arbitration as the final method. No incentives provided to undertake exploration / production of higher risk/ cost	investment. Field development plan includes a minimum production target approved by the NIOC. However, the consequences of not meeting the targets are not stipulated. Contracting party disputes to also be resolved via escalation, with arbitration as the final method. Incentives offered to IOCs for undertaking higher risk fields, and EOR on brownfield
sales agreement Remuneration fees can be unilaterally reduced by the NIOC if production falls below the level stated in the field development plan. Contracting party disputes to be resolved via escalation, with arbitration as the final method. No incentives provided to undertake exploration / production of higher risk/ cost fields.	investment. Field development plan includes a minimum production target approved by the NIOC. However, the consequences of not meeting the targets are not stipulated. Contracting party disputes to also be resolved via escalation, with arbitration as the final method. Incentives offered to IOCs for undertaking higher risk fields, and EOR on brownfield projects.
sales agreement Remuneration fees can be unilaterally reduced by the NIOC if production falls below the level stated in the field development plan. Contracting party disputes to be resolved via escalation, with arbitration as the final method. No incentives provided to undertake exploration / production of higher risk/ cost fields. Sanctions and 'snapback' of secondary	investment. Field development plan includes a minimum production target approved by the NIOC. However, the consequences of not meeting the targets are not stipulated. Contracting party disputes to also be resolved via escalation, with arbitration as the final method. Incentives offered to IOCs for undertaking higher risk fields, and EOR on brownfield projects.
sales agreement Remuneration fees can be unilaterally reduced by the NIOC if production falls below the level stated in the field development plan. Contracting party disputes to be resolved via escalation, with arbitration as the final method. No incentives provided to undertake exploration / production of higher risk/ cost fields. Sanctions and 'snapback' of secondary sanctions not applicable	investment. Field development plan includes a minimum production target approved by the NIOC. However, the consequences of not meeting the targets are not stipulated. Contracting party disputes to also be resolved via escalation, with arbitration as the final method. Incentives offered to IOCs for undertaking higher risk fields, and EOR on brownfield projects. 'Snapback' of US secondary sanctions not recognised as a force majeure event. NIOC
sales agreement Remuneration fees can be unilaterally reduced by the NIOC if production falls below the level stated in the field development plan. Contracting party disputes to be resolved via escalation, with arbitration as the final method. No incentives provided to undertake exploration / production of higher risk/ cost fields. Sanctions and 'snapback' of secondary sanctions not applicable	investment. Field development plan includes a minimum production target approved by the NIOC. However, the consequences of not meeting the targets are not stipulated. Contracting party disputes to also be resolved via escalation, with arbitration as the final method. Incentives offered to IOCs for undertaking higher risk fields, and EOR on brownfield projects. 'Snapback' of US secondary sanctions not recognised as a force majeure event. NIOCC <u>not</u> to grant withdrawal rights to contractor

Sources: Creed and Kordvani 2014, 2-3; Kalehsar and Telli, 2017, 165; Parris and Skyner 2017, 1-4.

To sum up, the main issues with the Buy Back Contract are identified as its inflexibility in regard to the scope of the contract, the fixed capital costs and the limited opportunities to

recover cost blow outs.⁵³³ In addition, the contract scheme includes a limited amortization period to recover remaining costs and there is a lack of production operation coverage. Lastly, the Buy Back Contract model does not allow for additional investment in development and IOR/EOR operations, is time-consuming in terms of approval procedures and requires many interventions in decision making.

5.5.1 Pros and cons of the Buy Back Contract

The Buy Back Contract scheme in Iran was estimated to attract around US\$50 billion from foreign investor companies between 1995 and 2005.⁵³⁴ The implementation of sanctions from 2010 to 2014, however, saw a significant down turn in investment as international organisations withdrew their interest to avoid ramifications from the US.⁵³⁵ The development of the contract model emerged in the context of a lack of modern / advanced technologies across the industries supporting the exploration, development and production of oil and gas fields in Iran.⁵³⁶ As previously established, it is a type of service contract granting contractor rights to an IOC to explore and develop the nominated oil or gas field.⁵³⁷ Following NIOC approval, the costs and expenditures of the project for the IOC are to be reimbursed to them from the revenue generated through the sale of the product according to the terms and conditions of the contract.⁵³⁸ Broadly speaking, because the IOC is required under the terms of the Buy Back Contract to make up-front payments for the costs of exploration and development up-front, at no stage is it in a position to gain equity rights under the agreement.⁵³⁹ Table 5.3 provides a snap-shot summary of the main pros and cons linked to Buy Back Contracts from the perspective of Iran's interests.

Table 5.3 Pros and cons of Buy Back Contracts for the oil and gas sectors in Iran

Pros	Cons
Contract model meets Constitutional	Project risks heavily carried by IOC,
requirements including the protection of	meaning the contract model is broadly

⁵³³ Li et al (n 528) 43–45.

⁵³⁴ Li et al (n 528) 43.

⁵³⁵ United States Energy Information Administration (n 38).

⁵³⁶ M. Me, 'The Iranian Buyback Model and Its Efficiency in the International Petroleum Market-A Legal View' (2009) 7(1) *Oil, Gas and Law Intelligence* 22.

 ⁵³⁷ Abbas Ghandi and Cynthia Lin Lawell, 'On the Rate of Return and Risk Factors to International Oil Companies in Iran's Buy Back Service Contracts' (2017) 103(1) *Energy Policy* 16, 16.
 ⁵³⁸ Ibid 18.

⁵³⁹ Mohammad Reza Moghaddam, 'Analysis of Buy Back Contracts and Adduction Optimization Model of Contract for Iran (Farsi)' (2008) 76 *Economy Research Journal* 162, 162.

Iran's sovereignty (ownership) over its	unattractive to IOCs, thus limiting optimal
natural resources	access by Iran to foreign investment capital
Relatively short contract duration forces	Relatively short contract duration limits
IOCs to carry most of the risks around	incentivisation of IOCs to optimise
exploration, development and production.	petroleum production efficiencies and
	opportunities for transfer of advanced
	technologies to Iran's oil and gas sectors
Omission of any clause allowing IOCs to	NIOC capacity to reimburse IOC for
renegotiate the contract terms and	exploration and development costs impacted
conditions due to unexpected events means	by changes in the market price of crude oil
risks related to fluctuations in the price of	
petroleum and their implications for revenue	
raising from volume of petroleum produced	
is carried by the IOC	
Iran's legal system and domestic courts	Complicated bureaucratic administrative
have jurisdiction over the settlement of	processes limit the NIOC's capacity to
disputes and certain commercial	respond quickly and consistently to decision
transactions	making on unexpected and unavoidable
	changes in circumstances.
Fixed ROR provision means risk of not	Fixed ROR does not incentivise IOCs to
being able to generate sales revenue to	maximise total production via pursuit of
cover petroleum production costs of due to	additional reserves and improvements in
falling petroleum prices is carried by the	petroleum recovery and production methods
IOC	

(Sources: Bunter, 6, 7; Ebrahimi and Shiroui, 2; Kuhn and Jannatifar 15; Me, 22)

5.5.1.1 Restrictive strategic objective

A more comprehensive analysis of the pros and cons of the Buy Back Contract model from the perspective of Iran reveals a range of complexities. For instance, depending upon your adopted political position, the strategic objective of the Buy Back Contract model may be viewed as either a pro or a con. Adopting a politically neutral position, this thesis asserts that the strategic objective of the Buy Back Contract model may be viewed primarily as a 'con' or disadvantage for Iran's oil and gas sectors overall. To explain, it has been established that the objective of the Buy Back Contract is two-fold: to preserve the sovereignty of the Iranian government over the nation's natural resources; and to preserve the Iranian government's control over the operations of its resource fields.⁵⁴⁰ Buy Back Contracts thus provide authorisation to IOCs to conduct petroleum field exploration and development on behalf of the NIOC.⁵⁴¹ Following the exploration and development stages, the production stage commences whereby management of the field operations is transferred to the NIOC or the affiliate organisation. The contract stipulates the annual capital expenditures with the IOC being fully reimbursed following approval from the NIOC.⁵⁴² Within all these contract stipulations, the requirement is for strict adherence to Iran's Constitution and Petrol Laws to meet the strategic objective of the Buy Back Contract; namely, the non-negotiable condition that Iran maintain ownership at all times over all resource field developments.

5.5.1.2 Risks carried by IOCs disincentivise field optimisation An important implication of the strategic objective of the Buy Back Contract model is that it contributes to perceptions of a high level of investment risk by IOCs. This is arguably a negative consequence of the contract model for the NIOC and for the development of Iran's resources the in terms of attracting foreign investment into its oils and gas sectors. Of most concern to IOCs is the high level of risk they are required to accept for the rights to explore for and develop a potentially commercial petroleum field.⁵⁴³ Contributing to their risk perceptions is that all assets acquired by the them remain under NIOC ownership, to be offset according to agreed petroleum purchases and remuneration from the NIOC until the control of the assets by the contractor is revoked.⁵⁴⁴ This is in addition to the risks carried by the IOC related to higher than estimated and agreed capital expenditures or higher distribution of capital requirements during the construction of the field. In these circumstances, the IOC is obliged to cover any additional costs.⁵⁴⁵

5.5.1.3 Risks for IOC related to costs recovery

Moreover, an unexpected fall in the price of petroleum in the market can add financial pressure to the operations of the IOC. It should also be noted that the terms of the Buy Back Contract are also potentially disadvantageous to the NIOC, given a fall in the price of petroleum increases its liability as remuneration to IOCs is paid in cash from petroleum

⁵⁴⁰ Ebrahimi and Khouzani (n 232) 2.

⁵⁴¹ Ibid.

⁵⁴² van Groenendaal and Mazraati (n 2) 3709.

⁵⁴³ Tavana Energy, 'Iran Petroleum Contract' <<u>http://tavana-energy.com/iran-petroleum-contract/</u>> (accessed 17 February 2020).

⁵⁴⁴ Ibid.

⁵⁴⁵ Ibid.

sales.⁵⁴⁶ However, the terms of the Buy Back Contract stipulate that the NIOC can postpone repayments on expenditures and remuneration from sales revenue if the price of petroleum goes below the agreed level. This is a clear detraction of the contract model for the investor company because it is required to cover the shortfall while maintaining the agreed rate of return from production.⁵⁴⁷ The risks around having no ownership stake, but still being required to meet rates of production, cover the costs of the exploration and development, and meet the schedule of the project have been raised consistently by IOCs as issues of concern. In essence, the Buy Back Contract does not offer the same level of compensation for these risks compared to alternative contract models.⁵⁴⁸ As a result, the detraction for IOCs is that there is little value in developing and/or deploying advanced technologies if the ownership of the technologies is to be transferred to the NIOC at the conclusion of the development phase.⁵⁴⁹

5.5.1.4 Implications of short contract duration

Heightening the level of risk perceived by IOCs is the relatively short life-span of the contract; namely, between 5 to 10 years along with a five-year investment recovery period. The short contract length is considered to reduce incentives for the IOC to implement processes to maximise the production life of the field. Rather, it arguably incentivises the IOC to implement potentially damaging production rates to maximise cost recovery within the contract period.⁵⁵⁰ Moreover, the short-term nature of the Buy Back Contract has implications for the IOC's capacity to develop a 'comprehensive' understanding of the field and thus the best strategic approach for its development.⁵⁵¹ This may include revising the investment requirements in response to changes in the extent or 'behaviour' of the field, or how to optimise the extraction of petroleum over the field's full life-cycle. This detraction would often steer IOCs towards alternative contract models including PSAs.⁵⁵² The lack of incentivisation of IOCs to optimise petroleum production efficiencies of the field under the terms of the Buy Back Contract thus implies a negative outcome for the

⁵⁴⁶ Tavana Energy (n 548) 162.

⁵⁴⁷ Ibid.

⁵⁴⁸ Roohollah Kohan Hoosh Nejad, Davood Manzoor and Masoud Amani, 'Economic Evaluation of Fiscal Regime of Buy Back Contracts in Comparison with Production Sharing Contracts (Case Study: Azadegan Oil Field)' (2018) 22(2) *Iranian Economic Review* 579, 579.

⁵⁴⁹ Moghaddam (n 544) 161.

⁵⁵⁰ Michael AG Bunter, 'The Iranian Buy Back Agreement' (2009) 3(1) Oil, Gas & Energy Law Intelligence (OGEL) 6, 7.

⁵⁵¹ Ibid.

⁵⁵² Ibid.

development of Iran's oil and gas sectors.⁵⁵³ The short contract duration in which to exploit the field to its full potential places pressure on the IOC to maximise costs recovery via the sale of petroleum during peak production. This limits the capacity of the IOC to actively seek ways to extend the period of peak performance, and therefore increase petroleum sales, before production is transferred to the NIOC.⁵⁵⁴

5.5.1.5 Risks to the contracting parties related to reimbursement of costs The reimbursement mechanism in the Buy Back Contract model however presents some disadvantages to both the NIOC and IOCs as contracting parties. To clarify, an initial amortisation period is established (typically 5-8 years) during which time the IOC contractor recovers costs plus interest based on the London Interbank Offered Rate (LIBOR) in addition to a per-month remuneration fee.⁵⁵⁵ Along with costs recovery, the terms of the Buy Back Contract include the payment of a fixed amount to the IOC to acknowledge the investment risk it has taken.⁵⁵⁶ This remuneration fee is paid upon achievement of the project objectives as set out in the Field Development Plan (FDP)⁵⁵⁷ and following the handing over of the project to the NIOC at the commencement of the production stage.⁵⁵⁸ In addition, the IOC is granted the right to a discount-rate purchase (typically 5% below market rate) of a percentage of the petroleum produced. As such, the NIOC must bear the risk of a fall in the market price of crude petroleum price and hence it capacity to cover the instalment costs to the IOC.⁵⁵⁹ Furthermore, because the IOC does not make a profit from an increase in the level of production beyond what has been agreed, the NIOC also loses out due to the lack of incentive for the IOC to boost output levels.⁵⁶⁰

In terms of the IOC, it is difficult for it to undertake an accurate calculation of costs under the Buy Back Contract model. The contract stipulates that all cost outlays and potential returns from production of the petroleum field are to be detailed accurately in both the FDP and the contract, with the calculation later used to determine compensation.⁵⁶¹ However, the

⁵⁵³ Iranian perspective refers to persons known as domestic critics.

⁵⁵⁴ Moghaddam (n 544) 162.

⁵⁵⁵ Bunter (n 555) 5.

⁵⁵⁶ Ibid.

⁵⁵⁷ The FDP details the scope of the project activities in addition to the plan to develop the field. Hence, it is integral to the Buy Back Contract model and its objective to facilitate the development of resource field operations in Iran.

⁵⁵⁸ Bunter (n 555) 6.

⁵⁵⁹ Ibid.

⁵⁶⁰ Alexander Brexendorff, Christian Ule and Maximilian Kuhn, 'The Iranian Buy Back Approach' (2009) 7(1) *Oil, Gas & Energy Law Journal (OGEL)* 22, 31.

⁵⁶¹ Richard Parris and Louis Skyner, 'Key Comparisons of the New Iran Petroleum Contract and Buy-Back' (Briefing Note, Clifford Chance, 2017) 1.

calculation to determine compensation is a difficult proposition for the IOC. To clarify, it is difficult to predict with a high level of accuracy the size of the petroleum reserves, how much time is needed for maximum production, and the costs of production more broadly prior to the commencement of operations.⁵⁶² Hence, any problems or issues with the processes around the collection of data on the wells may result in incorrect calculations that turn out to be significantly different to the final outcome.⁵⁶³

Another notable detraction of the Buy Back Contract model for IOCs is the fixed rate of return (ROR) provision.⁵⁶⁴ A fall in petroleum prices increases the risk for the IOC that the volume of petroleum produced from the field will not be enough to generate the sales revenue required to cover the costs of production and the remuneration amount agreed to in the contract. Furthermore, given the uncertainties around fluctuations in the price of petroleum, the actual production potential of field, and the final costs of production and so forth, the Buy Back Contract model is a risky proposition for IOCs because it does not include a provision for renegotiation.

5.5.1.6 No option for IOC to renegotiate

All contracts carry a risk for the contracting parties should the circumstances change in favour of one party over another.⁵⁶⁵ The risk can be mitigated to some extent however with the inclusion of a clause allowing for renegotiation of the terms and conditions in light of unexpected events. Because Buy Back Contracts do not offer the IOC – as the party most disadvantaged by a drop in petroleum price or a setback in operations – the option to renegotiate, it must carry a greater share of the overall risk that the NIOC.⁵⁶⁶ Nonetheless, the fixed ROR provision in the Buy-Bank Contract is arguably disadvantageous for Iran's oil and gas sectors. This is because it does not incentivise IOCs to maximise total production return via the search for additional reserves and the pursuit of improvements in petroleum recovery and production methods.⁵⁶⁷ However, the relatively short Buy Back contract period reduces

⁵⁶² Ali Emami Meibodi, Atefeh Taklif, Hamidreza Arbab and Hassan Bovairi Monji, 'Investigating the Effects of Contractual Factors and Arrangements on the Optimum Level of Production in Oil and Gas Projects: Evidence from the South Pars Phases 17 & 18' (2020) 24(1) *Iranian Economic Review* 181, 183.
 ⁵⁶³ Ibid.

<<u>https://www.cliffordchance.com/content/dam/cliffordchance/briefings/2017/01/key-comparisons-of-the-new-iran-petroleum-contract-and-buyback.pdf</u>> (accessed 1 December 2019).

⁵⁶⁴ The IOC's agreed ROR (profit) on its investment differs according to the nature of the project but is typically from 15-20%.

⁵⁶⁵ Brexendorff et al (n 565) 22.

 ⁵⁶⁶ Maximillian Kuhn and Mohammadjavad Jannatifar, 'Foreign Direct Investment Mechanisms and Review of Iran's Buy Back Contracts: How Far has Iran Gone and How Far may it Go?' (2012) 5(3) *Journal of World Energy Law & Business* 207, 208.
 ⁵⁶⁷ Ibid.

the incentive for the IOC to prioritise the acquisition and/or utilisation of modern or advanced technologies. As a result, this limits opportunities for the transfer of such technologies to Iran's oil and gas sectors.⁵⁶⁸

Foreign investor companies also point to the complicated bureaucratic processes associated with the administration of the petroleum mining project according to the terms of the Buy Back Contract.⁵⁶⁹ In particular, they have detailed their concerns that the NIOC was generally unable to respond quickly and consistently to decisions made unexpectedly by the IOC due to unexpected and unavoidable changes in circumstances. At the centre of their concern is that they feel the Buy Back Contracts do not facilitate a cooperative relationship between the IOC and NIOC where the latter acts equally in the interests of both contracting parties.⁵⁷⁰ IOCs have also indicated their frustrations with the process of tendering submissions for the rights to explore and develop oil fields. Issues have been raised about the difficulties of conforming to requirements, which some IOCs believe are unrealistic, as well as about the lack of a standardised approach, often rendering the process slow and too costly.⁵⁷¹

Lastry, an advantage of the Buy Back Contract model to the NIOC is the dispute settlement mechanism. Within Iran's legal system, domestic courts have jurisdiction over the settlement of disputes and certain commercial transactions. In turn, the contract model stipulates in its terms that international arbitration is used to settle any disputes to emerge between the contracting parties in relation to the terms and conditions of the contract and their implementation.⁵⁷² For instance, the approach to resolving disputes between contractors and the NIOC in Buy Back Contracts follows an incremental approach where mediation and negotiation are first initiated, and arbitration is left the final stage.⁵⁷³ The approach to resolving disputes between contractors and the NIOC in IPCs follows a similar incremental approach (i.e. escalation with arbitration as the final stage). However, the parties must first agree to the seat of arbitration and rules and their agreement must subsequently be approved by the Cabinet of Ministers in Iran.⁵⁷⁴

5.6 Summary of Iranian Petroleum Contracts

574 Ibid.

⁵⁶⁸ Bunter (n 555) 7.

⁵⁶⁹ Ibid 16.

⁵⁷⁰ Ibid.

⁵⁷¹ Ibid.

⁵⁷² Ebrahimi and Khouzani (n 232) 2.

⁵⁷³ Key comparisons of the new Iran Petroleum Contract and Buy Back' (n 247).

The IPC developed by the NIOC is a hybrid oil contract form containing some of the elements of the Buy Back Contract as well as some of the advantages of the PSA. The contract aims to address many of the perceived shortfalls of the Buy Back Contract, although the lack of ability for IOCs to book reserves remains an issue for IOCs. In terms of process, at the licensing round for the IPC, the IOCs bid for a stake in different elements of the development plan. For example, contractual work commitment, peak production, fee per barrel, pace of cost recovery and the like. A matrix is used to calculate the 'score' of each IOC (bidder). The additional factors given consideration in the licensing process include the level of complexity (e.g. development vs EOR), and the possibility of success (POS) in terms of any geological blocks to exploration. The main strategic objectives of the IPC are technology and know-how transfer (from the IOC to Iran) and the building of domestic capacity.

To achieve these objectives, the IPC is designed to encourage IOCs to utilise the latest technologies in the field of oil reserve management, optimisation and production. In addition, the IPC is structured around a tiered system of remuneration that allows for an increase in the amount paid to the IOC on each produced barrel if the production target is surpassed. Table 5.4 provides a summary of the highlights of the IPC:

Term	25+ years (compared to 5-10 years for Buy Back Contracts).
Signing bonus	None
Reserves	Provision for the transfer of hydrocarbon ownership to foreign entity
	at a specified delivery point. Reserves bookings are not permitted.
Ownership	Provision for shared ownership of the project's assets.
JVA	The IOC and domestic partner entity enter into a JVA to develop the
	field together. The NIOC oversees the planning and operations.
Capex	Capex is no longer pre-set. The JV parties formulate the MDP which
	includes provisions to permit changes to the Capex if required. An
	annual budget and work plan is developed and ratified by the JV.
Complexities	Different oil fields have different complexities and therefore
	different risks and reward attached. High risk fields naturally require
	higher returns to be paid to IOC compared to low risk fields.
Remuneration	Up to 50% of cost can be recovered by IOCs, in addition to an
	agreed fee per barrel paid in cash or in-kind.

Table 5.4 IPC Highlights

Asset life cycle	Provision for the contract transition across different reserve stages (i.e. exploration, development, production and EOR/IOR) without having to retender.
Marketing	Provision for the IOC to market products.
Corporate Social	IOCs are encouraged to initiate community development projects
Responsibility	(e.g. contribution to the building of hospitals) in oil-producing
	regions.
Transparency	IPCs include provision to ensure better financial transparency to
	minimise the risk of corruption.
Local content	At least 10% and up to 20% of the local partnership, with priority
	given to Iranian companies, followed by the Iranian-IOC JVs.

5.7 Critical Analysis of the Iranian Petroleum Contract

The introduction of the IPC clearly signals Iran's objective to formulate a new investor contract that includes more incentives for IOCs than the Buy Back model. From a broad perspective, the main attraction in the framework for foreign investors is arguably that the terms and conditions can be negotiated on a project-by-project basis. This means the contract can be better tailored to the IOC's needs compared to the uniform contract nature of the Buy Back Contract model. Another point of attraction in the IPC for foreign investor companies is the share of the produced oil they receive which may be sold by the IOC in the global petroleum market. This clearly responds to the disincentivising terms of the Buy Back Contract where companies were paid a fixed fee irrespective of the amount of oil produced. By offering IOC a share of the excess oil they have greater incentive to meet and exceed output targets.

The emergence of IPCs and their aim to generate new foreign investments in the hydrocarbon sector occurred after long consideration of several critical factors including how best to meet the interests of both Iran and IOCs; the contractual conditions required to respond to upstream competitors of Iran, and what type of legal authority was required to best arbitrate or resolve disputes between the parties.⁵⁷⁵

5.7.1 Perceived weaknesses and ambiguities

Critics of the IPC claim there are several weaknesses and ambiguities in the design and elements of the contract. Such criticisms follow the concerns raised by supervising

⁵⁷⁵ Paul Stevens, *Prospects for Iran's Oil and Gas Sector* (Chatham House for the Royal Institute of International Affairs, 2015) 20.

institutions such as the Parliament of Iran and the General Inspection Office to the Ministry of Petroleum which prompted several revisions to the original contract.⁵⁷⁶ From the outset, questions were raised as to the need for a new contract model (i.e. IPC) to replace the Buy Back Contract. While the Ministry of Petroleum responded to such concerns by pointing to the similarities between the IPC and Buy Back Contract forms, opponents pointed to the similarities the IPC shared with production contracts that inflict more costs on the Iranian economy compared to technical service contracts.⁵⁷⁷

The argument was also made that the primary focus of foreign IOCs was on managing project implementation costs and that this exposed another weakness in the IPC form.⁵⁷⁸ In response to these concerns, the Ministry of Petroleum in Iran insisted on the introduction and use of IPCs rather than technical service contracts because foreign companies were demonstrating a preparedness to take a direct role in the different project phases to develop an oil fields.⁵⁷⁹ Moreover, some IOCs had entered into negotiations with Iranian officials from the Ministry of Petroleum and National Petroleum Company affiliated companies.

Opponents of the IPC argue that most of the services provided by foreign companies could in fact be performed by domestic companies. In turn, the Ministry of Petroleum was quick to point out that the benefit of the IPC and its JVA component is that domestic companies may still take a significant role in the implementation of the project while in consultation with, and receiving guidance from, the foreign oil companies.⁵⁸⁰ That is, domestic companies can learn how to manage issues and weaknesses during project implementation which provide long-term benefits to the sector. Nonetheless, opponents of the IPC argue that allowing domestic companies to undertake project implementation without having to enter into a partnership with a foreign company would result in a suite of executive services related to project implementation being transferred more easily to Iranian companies.⁵⁸¹

Further to the role of domestic companies in the JVA of the IPC, it is worth pointing out that its role in the negotiations around the contract is secondary to the foreign company.⁵⁸² It appears that the terms of the IPC stipulate that all costs incurred by the Iranian company are reimbursed from the sale of petroleum produced from the field, similar to the arrangement for

⁵⁷⁶ Mazlumrahni and Esmaeili (n 433) 5.

⁵⁷⁷ Ibid.

⁵⁷⁸ Ibid.

⁵⁷⁹ Sadeghi and Seyed (n 496) 181.

⁵⁸⁰ Mazlumrahni and Esmaeili (n 433) 6.

⁵⁸¹ Ibid.

⁵⁸² Ibid.

the foreign company. This arguably suggests that the motivation, interests and goals of the domestic entity are necessarily similar to those of the foreign entity. This has important implications for the future of Iran's petroleum industry given the assumption that the primary interest of the foreign oil company (and by default, domestic partner companies) is to turn a profit rather than to ensure the long-term sustainability of the petroleum industry in Iran.⁵⁸³

5.7.2 Iranian law

The relationship between the IPC and Iranian law is also important for consideration. There is provision for three IPC types with IPC Regulations:

- Exploration Terms contracts for exploration, development and production,
- Development Terms contracts to develop existing discoveries, and
- IOR Terms contracts for improved/enhanced oil recovery at existing fields.

The Development Terms and IOR Terms IPCs focus on increasing the rates of petroleum recovery from the designated field. As such, it is assumed that the service fee is payable only when the IOC achieved agreed production targets. Due to the fixed nature of the upside from exploration compared to the potentially broad downside for the IOC, service contracts are negotiated for brownfield projects mostly, and rarely for exploration projects.

Based on the desire of Iran to increase the level of production from its oil fields as quickly as possible, the NIOC has initially given greater focus to negotiating IPCs with Development Terms and IOR Terms. This is clearly a strategy to attract IOCs given the greater difficulties associated with trying to lock in investment capital for exploration projects in the current market environment.

The IPC Regulation establishes the principles governing the IPC without disclosing the full IPC terms and conditions. Further clarity is therefore required across several areas. In terms of the JVAs, for instance, the IPC Regulation determines that foreign contractors must enter into a joint venture with one or more Iranian entity. The objective of the JVA according to the IPC Regulation is to encourage technology transfer. As shown in Table 5.1, eight domestic companies were pre-qualified by the NIOC to potentially enter into a joint venture partners with IOCs.

⁵⁸³ Masoud Derakhshan and Atefeh Taklif, 'The Transfer and Development of Technology in Iranian Upstream Oil Sector: Considerations on the Concepts, Requirements, Challenges and Remedies' (2015) 4(14) *Iranian Energy Economics* 33, 34.

5.7.3 Joint Venture Agreements and credit risk for international oil companies

However, there are important considerations in relation to entering into a JVA with an Iranian partner. First, there is a lack of clarity regarding the exact nature of the credit risk assumed by the IOC and how the Iranian partner pays its share of the petroleum operations costs. In addition, it is not clear whether there is a minimum level of interest that the domestic entity must agree to. Second, there are important implications related to fluctuating oil prices for the nature of the alignment between the IOC and the NIOC. The IOC is paid a fixed fee and as such it is arguably protected to some extent from the impacts of the changing oil price. A drop in the price of oil however would potentially result in friction between the NIOC and the IOC as the latter's services would appear to be relatively more expensive. Third, IOCs may reasonably be concerned about the remedies and their effectiveness should a domestic entity partner breach its contract obligations or fail to cover its portion of the petroleum operation costs.

Hence, IOCs must assess the extent to which the IPC aligns the interests of both contracting parties. It is the IOC who must assume all the risks related to the petroleum operations within the terms of the IPC. As such, they must carefully scrutinise the decision-making processes around all major investments. Regarding the terms of exploration particularly, IOCs must develop an understanding of how decisions about the commerciality of discoveries are made and the level of input afforded them in this regard. In addition, the decommissioning and restoration of an oil field is a costly endeavour and it is therefore in the IOC's interest to understand how the costs for such operations are arranged and any liabilities it may have in regard to the decommissioning process specifically.

What is clear from the analysis thus far is that the answers to RQs 1 and 2 particularly must emerge from consideration of the broader geo-political context. According to Dudlák, the Iranian government aimed to use the newly formulated IPC 'to sign 50 contracts with investors concerning exploration and production', including 'Asian companies that helped maintain the Iranian oil and gas production and development during the years of sanctions'.⁵⁸⁴ This speaks to the increasing geopolitical competition between the East and West and the focus towards the growing consumer market throughout East Asia rather than just the traditional IOCs operating out of developed nations.⁵⁸⁵ Moreover, IOCs operating out of the European Union (e.g. French Total, German Wintershall, and Norwegian DNO) have

⁵⁸⁴ Dudlák, (n 68) 465.

⁵⁸⁵ Ibid.

expressed interest in entering Iran's hydrocarbon sector. However, it remains the case (notwithstanding the recent sanctions imposed on Iran by the Trump Administration) that foreign companies remain cautious about entering into the investment process due to concerns about the uncertainty around the regulatory environment.⁵⁸⁶

5.8 Pros and cons of the Iranian Petroleum Contract

Regarding RQ1 (see Section 1.4), although the key terms and condition of the IPC were approved by the Iranian parliament in 2016,⁵⁸⁷ there remain several uncertainties and controversies around the structure of the contracts. Of particular focus is the lack of clarity around the terms of ownership of the resource field and the circumstances under which the IOC can reserve a field.⁵⁸⁸ Table 5.5 presents a snap-shot summary of the pros and cons of the IPC model from the perspective of Iran to facilitate a more comprehensive analysis below:

Pros	Cons
NIOC shares more of the project risks	Contract model may undermine integrity of
making the contract model more attractive	Iran's sovereignty (ownership) over its
to IOCs, thus increasing potential to access	natural resources
foreign investment capital	
Provides a new strategic approach to Iran to	IOCs have a stronger claim a direct
optimise the development / management of	economic interest in Iran's resource fields or
its natural resources	the companies operating them
IPCs provide IOCs with stronger incentive	Viewed as diverting focus and effort away
to invest and bring new technologies to	from Iran developing a 'resistance
Iran's resources sectors	economy' and taking control and ownership
	of the development of its oil and gas sectors

Table 5.5 Pros and cons of the IPC for the oil and gas sectors in Iran

⁵⁸⁶ David Ramin Jalilvand, *What's Fueling Opposition to Iran's New Oil Contracts?* (2016b) *al-Monitor* (<<u>http://www.al-monitor.com/pulse/originals/2016/05/iran-petroleum-contract-framework-delayed html#ixzz4Zz768ZpT</u>> (accessed 5 February 2020).

 ⁵⁸⁷ The terms were approved following the two-day Tehran Summit, on 28-29 November 2015.
 ⁵⁸⁸ Ahmadi (n 504) 200.

Higher foreign investment in Iran's	All strategic and operational decisions
petroleum sector means more domestic jobs	during the development and production
and more competitive modes of production	stages are assigned to the contractor via a
	Joint Steering Committee (JSC)
Dispute resolution mechanism favours	Lack of clarity regarding the point at which
NIOC or its affiliated entity as any disputes	the domestic company can choose to
are initially heard exclusively within the	participate during the development stage
jurisdiction of the Iranian courts rather than	and the type of contribution to be made
international arbitration	

Sources: Creed and Kordvani, 1-5; Dentons, 3; Mazlumrahni and Esmaeil, 1, 5; Seener, 20.

As with the Buy Back Contract model, whether or not specific elements of the IPC are a 'pro' or 'con' for Iran's oil and gas sectors may depend on the assessor's political position. This is evident in relation to the joint venture structure of the IPC and how one perceives the implications it has for Iran's sovereignty over its nature resources and the demands of the Iranian Constitution.

5.8.1 Unconstitutional?

Notwithstanding the 'pros' associated with the new IPC model, opponents of the IPC argue that the joint venture nature of the contract model is unconstitutional and that it weakens Iran's sovereignty over its natural resources. The IPC mandates that successful tendering companies must enter into a JVA with the NIOC or an affiliated domestic company from the energy sector endorsed by Iranian government.⁵⁸⁹ The contracting parties must then commit to joint operations to develop the negotiated petroleum reserve, with the IOC 'managing' the operations during the exploration stage.⁵⁹⁰ This transitions to a 'joint management' relationship during the development and production stages. However, the IPC terms applied to the management of the contracted oil field are somewhat more complex and variable during the different stages of operation than this initial delineation suggests.⁵⁹¹

 ⁵⁸⁹ Joanna Addison, 'The New Iranian Petroleum Contract (IPC)' (Herbert Smith Freehills, February 2016)
 <<u>file://ad.uws.edu.au/dfshare/HomesPTA\$/90923458/Downloads/%D0%AD%D0%B4%D0%B4%D0%B8%D1</u>
 <u>%81%D0%BE%D0%BD %D0%B0%D0%BD%D0%B3%D0%BB%20(1).pdf></u> (accessed 17 February 2020).
 ⁵⁹⁰ Ibid.

⁵⁹¹ Haddadi (n 423) 53.

5.8.2 Undermines Iran's sovereignty over its natural resources

Concerns about the implications this has for the sovereignty of Iran over its resources points to such terms of the contract as the requirement to establish a Joint Exploration Committee (JEC) which includes equal representation of the IOC and the NIOC. The JEC assumes management of the exploration operations, with all decisions taken unanimously by the IOC and the NIOC or an affiliated company.⁵⁹² The NIOC does however reserve the right of control over critical decisions related to reporting requirements, changes to agreed / proposed targets, and issues around commerciality.⁵⁹³ Similarly, all strategic and operational decisions implemented during the development and production stages of the petroleum project as assigned to the contractor via a JSC.⁵⁹⁴ Like the JEC, the JSC has equal IOC and NIOC representation, with the Chair position alternating between the IOC and NIOC each year. The JSC gives final approval to the Annual Work Plan and Budget (AWPB).⁵⁹⁵

5.8.3 Lack of clarity around role of domestic entity

However, uncertainty exists around the JVA structure regarding the point at which the domestic company can choose to participate during the development stage as well as the type of contribution to be made.⁵⁹⁶ Indeed, uncertainty remains around what criteria are to be applied to assess the commerciality of the field.⁵⁹⁷ Evidence of the impact of these uncertainties and controversies for example is found in the IPC contract negotiations between the NIOC and major IOCs including France Total and Shell UK. Although Memorandums of Understanding were reached on investment and field production with the major IOCs, they are yet to be cleared by the Iranian parliament.⁵⁹⁸ Throughout the operation stage, the petroleum project is managed by the JS. As such, Iran is positioned as a third party which, in principle, does not exact shareholding. However, this is determined according to the unique conditions of each project.599

As such, the IPC is viewed as a step backward by commentators such as Ali Reza Mazlumrahni and Mohsen Esmaeil, who argue that it is diverting focus and effort away from

⁵⁹² Ibid 53.

⁵⁹³ Ibid.

⁵⁹⁴ Dallas and Black (n 369) 3.

⁵⁹⁵ Ibid.

⁵⁹⁶ Mohammadi Tavakkoli, Mansour Ahmadnejad and Ali Eshaghzade, 'Identification of Factors Affecting the Development of Accounting and Financial Management Procedures for Joint Operating Agreements (JOA) in Iran's Oil and Gas Industry' (2018) 2(1) Petroleum Business Review 23, 23. ⁵⁹⁷ Ibid.

⁵⁹⁸ Jalilvand (n 498) 8.

⁵⁹⁹ Shervin Darvish, 'Legal Aspects of Iran's New Oil Contracts' (Natural Gas Europe, 18 July 2016) http://www.naturalgasworld.com/legal-aspects-of-irans-new-oil-contracts-30651 (accessed 17 January 2020).

addressing the weaknesses in the Iranian petroleum industry (lack of access to advanced technologies to increase production and reduce costs). These critics argue that Iran is relying too much on foreign companies for relief.⁶⁰⁰ Moreover, they point out that it is against Article 2 of the Act of Executive Policies of Principle 44 of the Iranian Constitution which restricts the extraction and production of crude oil and gas to the governmental sector.⁶⁰¹ Thus, the transfer of oil and gas field operations to foreign entities using IPCs is to these authors unconstitutional.⁶⁰²

5.8.4 More risks for Iran related to international oil company remuneration payments

In addition, risks exist for Iran around the payment of remuneration to IOCs. Debate continues as to whether the amount should be based on the average yearly petroleum export price in Iran (as is the practice), or whether to base it on the standard pricing formula applied throughout Asia.⁶⁰³ The investing company is entitled to remuneration proportionate to the level of risk associated with undertaking the exploration, development and production stages. The remuneration amount is variable as it is calculated according to the ratio of cash received to production costs (R factor) and to production rate. Hence, remuneration is based on a fee for every barrel of petroleum produced.⁶⁰⁴

5.8.5 Recovery of costs

Furthermore, limitations exist around the recovery of some costs outlaid during the development and production stages of the project. There are three main cost associated with the exploration to production phases in the IPC:

- 1. Direct Capital Costs (DCC)
- 2. Indirect Costs (IDC)
- 3. Cost of Money (CoM)

The IPC stipulates that cost recovery occurs 5-7 years after production starts. It includes incentives for decreasing costs via the Costs Saving Index (CSI).

The cost recovery term begins at the initial year of production. The 'direct capital costs' and the 'indirect costs' paid by the IOC before the start of production are amortised within 5-7 years of the start of production. Following the start of production, the IOC's direct capital costs are amortised within 5-7 years from the time of outlay. These costs do not include 'cost

⁶⁰⁰ Mazlumrahni and Esmaeil (n 433) 5.

⁶⁰¹ Ibid.

⁶⁰² Ibid.

⁶⁰³ Ahmadi (n 504).

⁶⁰⁴ Tavana Energy (n 548).

of money' during the development stage and following the initial date of production.⁶⁰⁵ Following the start of production, all additional costs including indirect capital costs are recovered by the IOC or investing entity at cost (i.e. without the cost of money). The amount recovered is based on the invoices submitted by the contractor (every quarter) as approved by the NIOC.⁶⁰⁶ In the event that the IOC cannot recover costs during the terms of the contract, the term is extended to provide extra cost recovery time based on approval by the NIOC.⁶⁰⁷

5.8.6 Take advantage of foreign technologies and expertise

Notwithstanding the issues that remain around the capacity of IPCs to meet all national interests of Iran, they at least provide an opportunity to take advantage of the investment and technical services of foreign and advanced countries.⁶⁰⁸ The capacity to increase foreign investment in Iran's petroleum sector translates to the creation of jobs, a more competitive mode of production, and the transfer of current knowledge and technologies.⁶⁰⁹ This point is also raised by Mohammadi and colleagues⁶¹⁰ who have stated that implementation of the IPC will positively contribute to the acceleration of the movement of petroleum from Iran following the introduction of soft technology and better knowledge transfer mechanisms around petroleum systems and products.

5.8.7 Increase optimisation of natural resources

It is important to note that the IPC were introduced in response to the 'limited easing of sanctions on Iran under the Geneva Joint Comprehensive Plan of Action (JCPOA)'.⁶¹¹ Notwithstanding the maturity of the Iranian petroleum sector, the international sanctions placed on Iran during the early years of the twenty-first century limited the nations capacity to attract foreign investment and technology into its oil and gas reserves, decreasing the nation's overall hydrocarbon output.⁶¹² As such, the introduction of IPCs in Iran reflects the country's strategy (and the strategies of other countries around the world also) to adopt a new approach to the optimisation of their natural resources.⁶¹³ Such approaches can generally be divided into two forms: initiatives designed to "maintain maximum control, ownership and

 $^{^{605}}$ Sahebonar et al (n 423).

⁶⁰⁶ Darvish (n 604).

⁶⁰⁷ Ibid.

⁶⁰⁸ Mazlumrahni and Esmaeil (n 433) 1

⁶⁰⁹ Ibid.

⁶¹⁰ Teymour Mohammadi, Farshad Momeni, Abbas Kazemi Najaf Abadi and Shirkou Bahadori, 'Effect of Petroleum Contracts on Iranian Oil Production Trend' (2016) 12(50) *Quarterly Energy Economics Review* 26.
⁶¹¹ Adrian Creed and Dr Amir Kordvani, 'Iran's New Integrated Petroleum Contracts' (MENA Legal update, Clyde and Co., 2014) 1.

⁶¹² Ibid.

⁶¹³ Ibid.

sovereignty" over the natural resources; or providing IOCs with greater scope to undertake a direct economic interest in the resources fields or in the companies that are operating them.⁶¹⁴ Put more simply, approaches favouring the use of technical services contracts versus production sharing agreements.⁶¹⁵

The emphasis in IPCs on supporting the acquisition know-how and technologies to develop the Iranian oil fields represents a turning point in how the nation approaches the development of its oil and gas industry. Previously, the main consideration in the Buy Back Contract model was to attract foreign investment without violating the Constitution and related oil and gas legislation in Iran. In turn, while the short-term service nature of the Buy Back Contracts between the NIOC and IOCs served this purpose, the model was largely unpopular with IOCs due to the risks they were obliged to carry.⁶¹⁶ IPCs provide an alternative approach designed to "incentivise international oil and gas companies to invest and bring new technologies to Iran".⁶¹⁷ Moreover, a key advantage of the IPC to foreign investors is the opportunity to step up the development and optimal exploitation of oil deposits, and to increase the coefficient of oil extraction.⁶¹⁸

Indeed, Creed and Kordvani assert there are five key objectives underpinning the IPC design:

- 1. To integrate the exploration and production phases.
- 2. To help Iran improve capacity, maintenance and reserve recovery.
- 3. To attract international capital, services, know-how and technology.
- 4. To establish long-term relationships with international partners.
- 5. To reduce investment risks via by more flexible investment costs.⁶¹⁹

5.8.8 Favourable dispute resolution mechanism

Along with the court system, disputes may be settled via other methods including arbitration.⁶²⁰ International arbitration is undertaken within the framework set up by the Iranian Law on International Commercial Arbitration (1997). In addition, ratification of the New York Convention on the Recognition and Enforcement of Foreign Arbitral Awards of 1958 by Iran furthered the positioning of international arbitration within the legal system of

⁶¹⁴ Creed and Kordvani (n 616) 1.

⁶¹⁵ Ibid.

⁶¹⁶ Ibid 2.

⁶¹⁷ Ibid.

⁶¹⁸ Fakhime Hadavimoghaddam and Masoud Mostajeran Gortani, 'Iran Petroleum Contract (IPC) - A New Generation of Oil Contracts' (2019) 13(4) *Actual Problems of Economics & Law* 1522, 1522.

⁶¹⁹ Ibid.

⁶²⁰ Smousavi (n 377) para 64.

Iran.⁶²¹ The two main arbitration entities in Iran are the Tehran Regional Arbitration Centre and the Arbitration Centre of Iran Chamber. The standard practice in international arbitration is that there is no provision for the right of appeal against an award. However, a party in the dispute may seek to have an adjudication set aside on certain grounds.⁶²²

Another positive element of the IPC for the NIOC is the dispute resolution mechanism. Arguably, the contract terms in this regard favour the NIOC or its affiliated entity, stating that any disputes regarding the IPC will initially be heard exclusively within the jurisdiction of the Iranian courts rather than international arbitration.⁶²³ However, in cases of an escalation in attempt at dispute resolution between the contracted parties, the IPC includes a provision for arbitration as the final dispute resolution mechanism.⁶²⁴ Covered in the clause are matters regarding contract breaches, and contract revocation or termination. The terms of condition of arbitration of any disputes are detailed in an annex to the IPC, which generally favour Iran's interests. Central to these conditions is that the IOC (contractor) bears the costs for failure to discover a commercial reservoir, failure to meet the aims of the contract, and when costs cannot be recovered due to an insufficient petroleum reserve.⁶²⁵

5.8.9 International oil company concerns with the Iranian Petroleum Contract

The objective underpinning the IPC was to offer petroleum contracts with more appealing terms and conditions for IOCs compared to petroleum contract from Iraq.⁶²⁶ However, the extensive revisions to the IPC format by the Iranian parliament meant that many IOCs did not necessarily regard the Iranian IPC as superior to the contract previously offered by Iraq. The main issue with the IPC for foreign investor companies is the lack of option to book an oil reserve, which is not permitted according to the Iranian Constitution. Such an option "is important for IOCs in terms of demonstrating their market value".⁶²⁷ In turn, the lack of success in attracting major IOCs from Europe by the Iranian government under President Rouhani is arguably the reason for the current challenges being experienced by operators in Iran's upstream oil sector.

⁶²¹ Ibid.

⁶²² Ibid para 65.

⁶²³ Creed and Kordvani (n 616) 2–3.

⁶²⁴ Mazlumrahni and Esmaeil (n 433) 5.

⁶²⁵ Ibid.

⁶²⁶ Tavana Energy (n 548).

⁶²⁷ Smousavi (n 377) para 21.

In turn, to help to raise capital for exploration and development of Iran's upstream oil sector, the Director of the NIOC, Ali Kardor, announced the 'partnership bond' scheme in 2017. In this scheme, the Ministry of Petroleum in Iran agrees to reimburse Rial investments in foreign currencies such as the Euro or the greenback (US dollars). Reimbursing bond holders has been challenging for NIOC however, with the company having to borrow money from domestic banks to meet its obligations.⁶²⁸ It may be asserted that part of the problem for the NIOC is the structure of the IPC itself. The mandatory requirement in the IPC for IOCs to enter into joint ventures with Iranian partners (excepting during the exploration phase) is arguably problematic. The general inability of domestic companies to provide any significant financial support means the onus is put on the IOCs to provide most of the investment capital.⁶²⁹ Another issue for the IPC in relation to attracting international investment capital is that banks in the West are reluctant to enter into transactions with Iran.⁶³⁰ Furthermore, the list developed by the NIOC of the public, private and semi-private companies in Iran to qualify to form partnerships with foreign companies was not without its issues. To briefly clarify, the partnership is in the form of a JVA for Development operations and the formation of a Joint Operating Company to oversee production operations.⁶³¹ The benefits include the transfer of technology and management skills, fewer interventions in operations, greater financial transparency, more efficient operations, and the internationalisation of operations.⁶³² An emergent issue however is the lack of clarity around the extent to which the private companies are to operate independently from the government agencies or from political interests.⁶³³ At issue is Article 44 of the Iranian Constitution and the implications it has for the scope of operations afforded privatised companies controlled by the government.⁶³⁴

Also, of concern to IOCs is the degree to which the IRGC is involved in some of the domestic companies to qualify for joint ventures. For example, Khatam al-Anbia Headquarters gained qualification from the NIOC and is owned by the IRGC, whereas a, IRGC subsidiary of the IRGC, was considered a private company did not achieve final qualification. Companies considered to have connections to the IRGC are at risk of being

633 Ibid.

⁶²⁸ Chow et al (n 22) 4.

⁶²⁹ Ghorbani (n 455) 2.

⁶³⁰ Dudlák (n 68) 28.

⁶³¹ Addison (n 594).

⁶³² Ibid.

⁶³⁴ Chow et al (n 22) 4.

subjected to sanctions from the US Treasury Department sanctions. However, the situation is made more complex because some of the domestic companies to 'qualify' for partnerships with IOCs are sometimes involved in aviation or aerospace industry projects. This is potentially problematic as these companies which may be subjected to US sanctions in relation to Iran's ballistic missiles program.⁶³⁵

5.9 Implications of the analysis of the Iranian Petroleum Contracts pros and cons

To answer RQ3, a thorough assessment of the objective and structure of the IPC must be undertaken to identify the extent to which they are positive for Iran's oil and gas sectors specifically as well as Iranian society more broadly. The role of OPEC in the global petroleum trade and the fact that the petroleum is exported in crude form by foreign entities means that Iran is trapped in a cycle of 'exporting crude oil for low prices and importing petroleum products for expensive prices'.⁶³⁶ Notwithstanding the objective of IPCs to address some of the constraints imposed upon the development of Iran's oil and gas industries from its protectionist laws, such laws nonetheless still challenge the potential effectiveness of IPCs.⁶³⁷ It remains the case that the prevailing anti-West/anti-imperialist political and religious sentiment in Iran perpetuates a component of distrust in its relations with the West including Western companies.⁶³⁸ This distrust is understandably intensified in the context of renewed and more expansive sanctions placed on Iran by the Trump Administration. Such sanctions have impacted Iran's opportunities to develop its upstream oil and gas industries and also forced foreign investment entities out of Iran which had contracts to develop and renovate resource fields.⁶³⁹ Moreover, the sanctions place restrictions around the capacity of investors to transfer funds into Iran. Hence, not only do the sanctions make it very difficult for Western foreign entities particularly to do business with Iran in general, they undermine the capacity of IPCs to deliver "advanced and up-to-date technologies, tools, and supplies" to is oil and gas industries.640

Indeed, the main point of attraction for IOCs in the design of IPCs is arguably the reduced risk they are obligated to accept to manage the development and production of the oil fields. However, issues of protectionist laws including arbitration of issues within the domestic

- ⁶³⁹ Ibid.
- ⁶⁴⁰ Ibid 2.

⁶³⁵ Ibid.

⁶³⁶ Mazlumrahni and Esmaeil (n 433) 1.

⁶³⁷ Ibid.

⁶³⁸ Ibid.

judicial system and the lack of a "secure environment that guarantees the safety of their investments and profit" must be resolved to enhance the attractiveness of the IPC's pseudosharing production agreements to foreign investors.⁶⁴¹ Furthermore, the response by the Iranian government to the reinstated and more comprehensive sanctions by the West combined with the competition within Iran's political arena are further causes of concern for investors.⁶⁴² Notwithstanding that the re-election of President Rouhani in 2017 in Iran and the general endorsement to his reformist policy that it implies, Iran's capacity to optimise the potential benefits of the IPCs arguably relies on the normalisation of relations with the West.⁶⁴³ This is evidenced in the IPCs negotiation immediately after the nuclear agreement was reached with Iran in 2015. For example, the 2016 agreement between French Total and the NIOC saw the Western IOC successfully negotiate a 50.1% stake in development phase 11 of the South Pars gas field, with China's CNPC (30%) and the Iran's own Petropars corporation (19.9%) making up the other signatories.⁶⁴⁴ This contract, combined with later agreements signed between Norwegian firms and the NIOC to conduct Caspian offshore drillings for example, were described by the media in Iran as the 'breaking of the petroleum dam'.645

The newly imposed sanctions on Iran by the US in 2017-18 however are a significant hindrance to the capacity of IPCs to realise their potential. At first consideration, they would appear to promote the competitiveness of European and Asian IOCs in the Iranian market. However, the legislation in the US to "punishes parallel activities of companies"⁶⁴⁶ in the Iranian oil and gas sectors means for many IOCs the decision to enter into investment agreements with Iran carries too significant a risk.⁶⁴⁷ Indeed, the US legislation has already resulted in European banks suffering losses of up to USD\$13 billion for conducting business with oil and gas operators in Iran.⁶⁴⁸

In terms of risks for successful IOC investors, it is important to note that the IPC does not recognise the "snapback" from the reinstatement of US secondary sanctions 'as a force

⁶⁴¹ Ibid.

⁶⁴² Dudlák (n 68) 28.

⁶⁴³ Diane Munro, 'Past Imperfect, Future Tense: Iran's Oil Industry Post-Sanctions' (The Arab Gulf States Institute in Washington, 2016) 17.

⁶⁴⁴ Ibid 30.

⁶⁴⁵ Arash Karami, 'Iran Signs First Energy Contract Since Nuclear Deal' (2017) *al-Monitor* http://www.al-monitor.com/pulse/originals/2017/07/iran-total-france-south-pars-gas-deal-criticism.html (accessed 5 February 2020).

⁶⁴⁶ Dudlák (n 68) 30.

⁶⁴⁷ Ibid.

⁶⁴⁸ Jalilvand and Ebert Foundation (n 499).

majeure event'.⁶⁴⁹ That is, it does not include a clause which removes liability for natural and unavoidable 'catastrophes' that disrupt the anticipated course of events that restrict the IOC's capacity to fulfil its contractual obligations. Hence, the IOC contractor is not granted any right to withdrawal in the event of such sanction-related snapback. This understandably de-incentivises foreign firms from investing in Iran, prompting foreign firms keen to access Iran's vast oil and gas reserves signing only initial exploratory memorandums of understanding.⁶⁵⁰

5.9.1 Political landscape

The drafting and eventual release of the IPC model must be considered in the context of Iran's political landscape. Although the new model contract was initially announced in Tehran in November 2015, the prospects of its release were clouded due to protests from hard-line political opponents of President Rouhani.⁶⁵¹ They threatened to block its passage through the Majlis. However, the emergence of a "centrists" coalition comprising reformists, moderates and pragmatic conservatives to positions of power in both Iran's parliament and "Assembly of Experts" saw important gains made in relation to controlling hard-line conservatists in Iran.⁶⁵² The shift towards more political centrism in Iran was evident in the increased number of moderates in both houses of Iran's parliament. This had the effect of consolidating the hold on power by President Rouhani and strengthening his position reelected into office for a second term in 2017.⁶⁵³ This, in fact, was the case, with the new political alignment with President Rouhani in the Iranian parliament leading to several key outcomes: (1) greater consensus on economic policy, (2) the pursuit of foreign investment, (3) JCPOA implementation, and (4) legislative and regulatory reforms to the upstream exploration and production activities of Iran's energy sector, which included the IPC.654 In turn, the decision making around these structural reforms to acknowledge and respond to the criticisms made of the Buy Back Contracts was crucial to encouraging foreign companies to invest the approximately \$185 billion needed for the modernisation of Iran's ageing energy sector infrastructures.⁶⁵⁵ Arguably, the responsiveness of key stakeholders in Iran; namely, the NIOC and legislators to act on the calls for change has been vital. The preparedness of

 ⁶⁴⁹ Barak Seener, 'Commercial Risks Entering the Iranian Market: Why Sanctions Make Investment in the Islamic Republic of Iran a High-Risk Proposition' (Jerusalem Center for Public Affairs, 2018) 20.
 ⁶⁵⁰ Ibid.

⁶⁵¹ Dallas and Black (n 369) 3.

⁶⁵² Ghorbani (n 455) 7.

⁶⁵³ Ibid.

⁶⁵⁴ Dallas and Black (n 369) 3.

⁶⁵⁵ Sahebonar et al (n 423) 4.

leadership to draw on the experiences of Iraq (i.e. their technical service contract model) and also seek to be more innovative and flexible with the new contract terms were important signals to investor companies.⁶⁵⁶

5.9.2 Cost recovery

The core elements reflected in the principles of the new contract model are collaboration, technology / know-how transfer, and the simplification / consolidation of previous contract terms. These principles manifest in the use of JVAs (including a revised remuneration scheme that incentivises IOCs to engage in more efficient and enhanced production) and a single contract to cover exploration, development, production and options for enhanced recovery maintenance (including incentives for higher oil recovery).⁶⁵⁷ Indeed, the capped cost recovery condition was unquestionably the target of most criticism levelled at the Buy Back Contract. This is not surprising given the its expectation that IOCs invest in oil field development in the knowledge that any additional costs beyond those stipulated in the pre-determined budgets would not recoverable. Combined with the weaknesses in the remuneration regime that offered no reward for increased productivity, the capped costs recovery did not incentive IOCs to invest in risky projects or those with marginal prospects.⁶⁵⁸ By removing these limitations, the IPC model provides greater incentives to IOCs to enhance production in three key ways:⁶⁵⁹

- First, they are offered full costs recovery amortised over 5-7 years (with the possibility of extension if production is insufficient to recover these amounts). In addition, annual repayments for costs recovered and service fees are limited to 50% of all annual revenue raised from the production of oil within the contracted area. The contractor is also reimbursed in full in cases where the contracted area does not yield adequate revenue to recover production costs in full.
- Second, the remuneration of financing costs in the amortisation period (capped at LIBOR+1%), in combination with all indirect fees related to the development phase such as income tax, customs duties and the like add to the internationally competitive nature of the IOC. Financing fees for example are not recoverable in contracts negotiated with Iraq.

⁶⁵⁶ Dallas and Black (n 369) 3.

⁶⁵⁷ Ghorbani (n 455) 7.

⁶⁵⁸ Dallas and Black (n 369) 4.

• Third, replacing the fixed cost regime in the Buy Back Contract with the standard yearly work programmes and budget process arguably adds more protections around cost for IOC investors.

Implemented via approval from the JSC and NIOC, the use of the committee approach arguably makes the cost control processes more robust. This is important throughout the exploration phase particularly as the set of Minimum Obligations mandated in the IPC differentiate between the technical and operational requirements of this phase. Moreover, by specifying the financial commitment in implementing the Minimum Obligations, there is greater certainty generated around the establishing the budget requirements.⁶⁶⁰ This can place limitations around the degree of flexibility in cost overruns following approval of the figures via the JSC process, with no more than 5% leeway through the production stage, incurring a penalty as part of the production fee structure.⁶⁶¹ Cost overruns are not permitted however through the exploration and development stages, unless there is mutual agreement by the JSC and NIOC due to changes in the project scope or target.

5.9.3 Fee structure

Historical criticisms of the Buy Back Contract were also directed towards its remuneration structure. What emerged as an issue for foreign companies was that the fees they incurred were tied to a fixed percentage of the capital costs up to the cap stipulated in the budget. This structure failed to deliver incremental revenue for going beyond production targets and did not deliver an upside in relation to oil price increases.⁶⁶² The fee structure within the new IPC however is based upon the volume of production extending for the duration of the contract agreement. As previously reported, this can be up to 20 years in cases where discoveries have led to production, with the possibility of a tail extension of up to five years to implement enhanced oil recovery (EOR) methods.⁶⁶³

The pricing structure of the Iranian IPC is more sophisticated than both the Buy Back Contract and the Iraqi contract model. However, a close analysis reveals there are a number of complexities related to the exact way in which the remuneration is calculated. The main factors to influence the remuneration paid to a contractor are the base fee payable per barrel

⁶⁶⁰ Li et al (n 528) 47.

⁶⁶¹ Ibid.

⁶⁶² Dallas and Black (n 369) 5.

⁶⁶³ Ibid.
of production and the link between the fee and market prices.⁶⁶⁴ A percentage of the market price must be determined to represent the base fee and then agreed to by the IOC. In addition, the 'cap' mechanism embedded into the IPC to prevent IOCs from accessing mega profits from an unexpected surge in commodity prices.⁶⁶⁵ The IPC indicates that the cap applies when rolling average market prices go above the market price at the time of first production by a specified predetermined percentage threshold. Nonetheless, the exact percentage is not specified, and it is not indicated clearly if a 'collar' is also applied when the market oil price drops in order to protect the IOC. This is an important consideration given that low market prices draw out the amortisation period for costs recovery.⁶⁶⁶

Base fees in the IPC are subject to further multipliers per barrel to provide an incentive to IOCs to explore high-risk contract areas.⁶⁶⁷ Another volumetric base fee multiplier with the aim to reward incremental production is the separate fee multiplier applied to brownfield sites (or to greenfield sites where EOR methods have been used).⁶⁶⁸ On close analysis however it is not clear in the IPC structure as to whether or not additional variations are available to the pricing target for fields which producing heavier grades of oil. Moreover, the R-actor adjustment mechanism in the IPC is calibrated so that the fee payable to the IOC may be adjusted according to: (i) the production level of the field, and (ii) the ratio of costs recovered to revenue received.⁶⁶⁹ The ratio of costs calculation is made according to the multiple by which the IOCs' received revenues go beyond total expenditure sustained for the contracted field.⁶⁷⁰ There are also additional layers within each band which vary in line with the daily rate of production. Thus, it may be assumed that the fee structure changes in line with the maturity of the contract field as the output and recovery ratio increase. In addition, although cost recovery is permitted if the exploration phase of the field is unsuccessful; namely, the IOC accepts full exploration risk, it is not clear as to whether or not foreign companies are given preferential treatment when adjacent fields are allocated or for future exploration opportunities.671

⁶⁶⁴ Roya Derakhshanalavijeh and Jose Manuel Cardoso Teixeira, 'Cost Overrun in Construction Projects in Developing Countries, Gas-Oil Industry of Iran as a Case Study' (2017) 23(1) *Journal of Civil Engineering and Management* 125, 125.

⁶⁶⁵ Dallas and Black (n 369) 5.

⁶⁶⁶ Ibid.

⁶⁶⁷ Ghorbani (n 455) 7.

⁶⁶⁸ Dallas and Black (n 369) 5.

⁶⁶⁹ Parris and Skyner (n 566) 3.

⁶⁷⁰ Ibid.

⁶⁷¹ Dallas and Black (n 369) 5.

In turn, because there are few working examples to go by due to the reimposition of US sanctions, there remains are great deal of uncertainty as to how the nature of the relationship between the different adjustment criteria. The IPC states that the fee is payable per barrel produced whether or not the 'value' of the hydrocarbons is earmarked for cost recovery (similar to the contract model in Iraq, where the fee is payable per barrel produced). Furthermore, there is a lack of clarity as to the order in which the different fee adjustments are applied.⁶⁷² Nonetheless, the more robust cost-recovery structure combined with the implementation of a true volumetric tariff and extra incentives for incremental production may incentivise IOCs to invest in the technology required to maximise oil production in green and brownfield reserves in Iran.

5.9.4 Joint Venture Agreements and management structures

Arguably, the most significant change to the energy contract structure in Iran was the move to the joint venture model. The primary driver of this move was undoubtedly the objective in Iran to the technology and know-how of IOCs. According to Dallas and Black,⁶⁷³ the technology and know-how was especially needed in regard to modern EOR methods which were lacking across the energy industry. As stipulated within the IPC, the IOC must partner with the NIOC or a domestic energy company according to an incorporated/unincorporated JVA. Arguably, the rotating management arrangements as a framework for domestic entity participation have the potential to provide much needed benefits to Iran's energy sector.⁶⁷⁴ IPC By-law Article 4(4) for instance endorses the transfer of technology and know-how from the IOC to the joint domestic entity and determines the JOC's organisation chart during production operations according to the rotated arrangement. Lastly, IPC By-law Article 8(4) and its definition of the "Joint Management Committee" ("JMC") is arguably of potential benefit to Iran as it provides the NIOC with equal membership on the committees for supervising and monitoring how the petroleum operations are executed. Indeed, the IPC By-laws endorse three types of contracts with foreign entities: (1) exploration and production (E&P) contracts which basically integrates the exploration, appraisal, development and Production Service Contract elements of the project area; (2) development

⁶⁷² Derakhshanalavijeh and Teixeira (n 627) 127.

⁶⁷³ Dallas and Black (n 369) 6.

⁶⁷⁴ Seyed Nasrollah Ebrahimi and Esmat Shahmoradi, 'A Comparative Analysis of Contractual Considerations in Oil and Gas Contracts with an Emphasis on Iranian Petroleum Contracts' (2017) 15(3) *Oil, Gas & Energy Law Intelligence* 4, 6.

and production (D&P) contracts; and (3) IOR/EOR contracts designed for the production stage.⁶⁷⁵

IPC By-law Article 1(4) provides the Iranian governments with greater control over the standards of operations by affording the NIOC with the power to specify the key performance indicators (KPIs) to determine the commercialisation of the field.⁶⁷⁶ By inserting these KPIs into the contrast, combined with regular monitoring and reporting on the petroleum operations, there is enhanced opportunity for the NIOC to identify and address any critical concerns during the operational phases. Moreover, the NIOC may work with the IOC to refine or tweak the agreed KPIs in order to have more control over performance and the achievement of its strategic objectives.⁶⁷⁷

This is because the KPIs allow the main objectives to be quantified and renders the performance of the different teams and departments more 'visible'.⁶⁷⁸ This provides the NIOC with a stronger position that the Buy Back contract for making decisions and for taking action to achieve its desired outcomes.⁶⁷⁹ KPIs are acknowledged to play a key role in diverse types of business success by improving a company's ability to generate performance data and thus assess the extent to which the performance outcomes align with the short-, mid- and long-term objective.⁶⁸⁰ This aspect of the IPC also benefits the NIOC's or affiliated company's role in the development of Iran's energy sector by making more visible the capital 'performance' during operations in regard to costs, risks, resource allocations, returns.

5.9.5 Time extensions

IPC By-law, Article 7 includes the rules governing the execution time of the development and productions phase. As previously mentioned, the contract duration is a maximum 20 years from NIOC approval of the Development and Production Plan (DPP). This time frame may be extended by mutual agreement for up to 5 years. The capacity in the IPC for there to be claims for and grants of extensions of time in completing oil projects is also of importance. The time factor adds a degree of volatility to petroleum contracts given the longer the contract term the greater the risk of changes to laws, market conditions, and

⁶⁷⁵ IPC By-law, art 2.

 ⁶⁷⁶ M Soleimani and Newsha Tavakolian, 'Comparative Financial Analysis of IPC, the New Iran's Oil Fields Development and Production Enhancement Framework IPC vs. Buyback Contracts' (SPE Symposium: Production Enhancement and Cost Optimisation, Society of Petroleum Engineers, 2017) 3.
 ⁶⁷⁷ Ebrahimi and Shahmoradi (n 679) 6.

⁶⁷⁸Guy Redden, *Questioning Performance Measurement: Metrics, Organizations and Power* (Sage, 2019) 19. ⁶⁷⁹ Ibid.

⁶⁸⁰ Ebrahimi and Shahmoradi (n 679) 6.

petroleum prices, all of which reduce contract performability.⁶⁸¹ Conversely, however, it is not feasible to have too much flexibility around time frames embedded into the contract as this increases the risk of misinterpretation and can undermine contract certainty and reliability.⁶⁸²

Extending the period of the contract can be a complicated process with significant financial or non-financial implications for the contract parties.⁶⁸³ Although there can be quite severe consequences to both IOCs and domestic oil companies as a result of delay and disruption to the petroleum project, the commercial advantage of both entities may also be served with time extensions in upstream contracts. The inclusion of clauses in the contract for extensions of time due to unpredicted factors and force majeure can be embedded as rules and principles in an IPC.⁶⁸⁴

The conditions under which extension of time clauses may be enacted include for:⁶⁸⁵

- 1. an act, omission, breach or default by the NIOC or affiliate,
- suspension of works initiated by the NIOC or affiliate (but not due to an act or omission of the IOC),
- 3. a variation imposed by the NIOC or affiliate (but not variation due to an act or omission of the IOC), and
- 4. force majeure, causing delays in critical pathways identified by the IOC within the specified period.

Standard IPCs include provisions for extension of time entitlements to the IOC due to NIOC forced variations or changes, a suspension of activities, force majeure, and changes to the law. As such, well-considered and drafted extension of time clauses in an IPC can benefit both parties. For instance, it can help with preserving a contractual completion time; maintaining NIOC rights to liquidated damages, and by providing IOCs with relief from strict obligations to finish the works by a stated time in the event of delays due to unanticipated 'neutral' events.⁶⁸⁶

⁶⁸¹ Ebrahimi and Shahmoradi (n 679) 18.

⁶⁸² Ibid.

⁶⁸³ Soleimani and Tavakolian (n 681) 5.

⁶⁸⁴ Seener (n 654) 20.

⁶⁸⁵ Ebrahimi and Shahmoradi (n 679) 17.

⁶⁸⁶ Peter Godwin et al, 'The Prevention Principle, Time at Large and Extension of Time Clauses' Construction Dispute Avoidance Newsletter' (2009) <<u>http://documents.lexology.com/4001b4ac8d7a-4d10-b880-ac10e8f54d2a.pdf</u>> (accessed 15 August 2019).

However, the exact way in which the ownership interests of each party is determined, and the extent and timing of each party's participation is not clearly stated.⁶⁸⁷ That is, the point during field development at which the domestic entity chooses to participate is not clearly determined. Moreover, and it is not clearly apparent as to whether the contribution by the domestic entity is to be primarily technical, financial or a mix of both.⁶⁸⁸ It appears that in the IPC the decision-making thresholds are somewhat disaggregated from both the ownership interests of the parties and to the funding regime. For instance, it is expected that the IOC will cover its own costs and the carry of its development partner during exploration, development and production stages.⁶⁸⁹

There are also complexities related to the management structure of the JVAs that are worth consideration. Notably, the management structure varies depending on the exploration or development stage. To explain, the IOC manages the operations and has control over the dayto-day issues during the exploration stage.⁶⁹⁰ In terms of control over operational issues, however, this is deferred to the Joint Exploration Committee (JEC) comprising an equal number of NIOC and IOC members. All decisions made by the JEC must be unanimous, however, the NIOC has the final say in decisions on key issues such as the arrangement of final reports and statements, changes to the targets of petroleum projects, and determinations of a field's commerciality. On this last issue, it is noted that the decision is subject to expert referral in circumstances where the IOC and NIOC cannot reach an agreement.⁶⁹¹ The separation of decision making from ownership and funding responsibilities does not align with the more established PSC models regularly used across the energy sector.⁶⁹² Moreover, there is the need for the IPC to be more explicit in terms of the criteria to be applied to determine commerciality. It is in the interests of the IOC to seek to relinquish a non-commercial prospect and there cannot be the expectation that it continues to invest capital into a marginal prospect without fully understanding its overall economic recovery.⁶⁹³ Notwithstanding that the JVA structure is widely preferred to the risk services model (i.e. Buy Back Contract), the way in which the funding obligations of the IOC are separated from

⁶⁸⁷ Mohammadi Tavakkoli and Ali Eshaghzadeh, 'Developing a Framework for Accounting and Financial Management Procedures of Joint Operating Agreements (JOAs) in Iran's Oil and Gas industry' (2018) 2(1) *Petroleum Business Review* 69.

⁶⁸⁸ Ibid.

⁶⁸⁹ Ibid.

⁶⁹⁰ Dallas and Black (n 369) 6.

⁶⁹¹ Dallas and Black (n 369) 6.

⁶⁹² Tavakkoli and Eshaghzadeh (n 692) 70.

⁶⁹³ Ibid.

the ownership and voting mechanisms has the potential to be problematic. This is particularly the case given that the IOC is obliged to cover all exploration and production costs in addition to the percentage share of carry for the NIOC/local entity during these stages. This presents a range of issues, not least of which is the level of competence (technical) of the domestic entity in the JV, along with any associated sanctions for possible links to the IRGC. Of concern to any IOC is the extent to which it will have to carry its JV partners both financially and in a commercial sense. As such, it is important for the IPC model to make explicit how ownership interests are formulated and the timing and scope of participation by the NIOC/domestic entity in the contracted field.

The clarity of the overall management structure embedded within the IPC is vital to IOCs, particularly throughout the exploration and development stages of the project. This is because it influences which decisions have to be deferred (escalated) to the JEC or JSC and eventually the NIOC. There is some clarity on the conditions to escalate decisions related to the appointment of key personnel or in regard to stipulated decision-making categories. An example is the limited but non-exhaustive list of exploration-phase 'issues' requiring escalation to the NIOC for a final decision. This includes determinations of the commerciality of the field.

It may reasonably be argued that IOCs have a legitimate concern that 'unnecessary' interventions by the JSC may cause delays to the daily field operations. In addition, they may be concerned that the JV partners will interfere in the development initiatives beyond the formulation of the AWPB and Development Plan. Moreover, IOCs will have concerns about whether the NIOC or domestic entity can 'direct' it to undertake an activity that it would not have chosen to perform otherwise (e.g. raising EOR compression). The way in which the Board is composed as well as how the Chairman is appointed would suggest that the only factors mitigating for these risks are the cost recovery mechanism and the ability of the IOC to refer these types of decisions to arbitration.

In terms of the designated domestic entity, whether it enters into a JVA with the IOC directly has implications for decision making. It is noted that in the Iraqi model, the power of the local entity is limited to veto decisions via the introduction of an 'offshore' JVA between IOCs in order to align how they vote.

5.9.6 Timing

It appears that the transition from the Buy Back Contract to the IPC model is the right strategy by Iran, but which has emerged at the wrong time. Iran is endeavouring to reestablish itself in the global petroleum market in the face of many obstacles. The impacts of the US sanctions on Iran's economy and energy sector more specifically has been well documents in this thesis. When these geopolitical tensions are combined with the intensifying debate around the global response to climate changes, many countries are starting to recalibrate their energy strategies. International investor companies are naturally cautious of the financial repercussions for doing business with Iran in opposition to US sanctions. Moreover, the energy sectors in countries around the world are simply looking to reduce their reliance on fossil fuels.⁶⁹⁴ Countries in the twenty-first century are pursuing greener and cleaner fuels and technologies in an attempt to minimise their carbon footprint. New product entrants into the market such as shale oil are also having their impact as they challenge the status quo of global oil powers. Just how Iran manages these complexities in the coming years will arguably go a long way to revealing whether or not Iran can establish a stronghold in the global petroleum market or continue to see a downward slide market share.

5.9.7 Lack of access to technology

In addition to the lack of investment in the Iranian oil and gas sector as a result of the US sanctions is the lack of access to technology for the development of the sector. This thesis has established that development of the energy industry in Iran has been constrained for many decades due to underinvestment and an inability to access technology. Implementation of the JCPOA provided some cause for hope that the IPC would generate a flow of much-needed investment and technology into the resources sector.⁶⁹⁵ Foreign investor involvement has not materialised as anticipated however, and this is not likely to change due to the US withdrawal from the JCPOA.

As reported above, Iran has signed several MoUs with IOC and of foreign entities to jointly develop a number of oil reservoirs, yet the finalisation of any contracts has been sent into disarray as a result of companies being unable to get a sanctions waiver from the US.⁶⁹⁶ Indeed, efforts by European oil companies to participate in developing and exploiting Iran's petroleum industry have largely come to a standstill due to the renewed US sanctions. Almost all Western IOCs with the capabilities to implement petroleum development and production projects worth billions of dollars have ties to the US financial (banking and insurance) systems.⁶⁹⁷ Given that these companies are cautious of snap-back from the secondary

⁶⁹⁴ Prakash (n 459) 1.

⁶⁹⁵ Chow et al (n 22) 13.

⁶⁹⁶ Reuters (n 27).

⁶⁹⁷ Ghorbani (n 455) 5.

sanctions should they enter into business arrangements with Iran, they are reluctant to sign any contracts.⁶⁹⁸ This has forced the NIOC to increasingly rely on domestic contractors, know-how and technologies to continue to develop its oil fields, including the most challenging onshore (e.g. Arvandan in the West Karoun region), and offshore (e.g. IOOC) fields.⁶⁹⁹ It should be noted that domestic contractors have proven to provide satisfactory support overall to the NIOC in the past. However, it is inevitable that the operational capacity of these domestic operators is significantly reduced as a result of the deteriorating economic landscape for domestic players in Iran due to the new sanctions.⁷⁰⁰

In turn, as a result of the limited positive outcomes of the IPC, the NIOC has undertaken to develop more robust contract models to more actively involve domestic contractors (e.g. Arvandan Oil and Gas Company [AOGC], IOOC and Iran Central Oil Fields Company [ICOFC]) in the development of domestic regions. Several (around 34) field developments, flow pipelines, and above-ground facilities have been earmarked by the NIOC requiring investment of between \$6 to \$7 billion.⁷⁰¹ The projects involve developing new oil wells in numerous onshore and offshore fields and undertaking water injection pressure maintenance assessments. In addition, the NIOC considered the introduction of different contract types; namely, the Engineering, Procurement and Construction (EPC) and the Engineering, Procurement and Drilling (EPD) contracts which include a provision for deferred payment to domestic contractors.⁷⁰² Contracts types such as these would provide the platform for the NIOC to continue to develop projects. However, based on the previous discussion, the progress of such projects is significantly impacted by Iran's weakened economic conditions and the lack of technology transfer barriers as a result of US sanctions.⁷⁰³ Given the circumstances currently facing Iran's petroleum sector as discussed above, the IPC as a contract form to help develop Iran's oil fields remains in a state of uncertainty. The failure of the contract to thus far attract foreign investment and to promote access to technology has resulted in significant challenges for the upstream petroleum sector and has prevented the NIOC from achieving production growth targets.⁷⁰⁴ The increasing presence of Russia in the energy resources landscape in Iran may prove to be an integral element of the

⁶⁹⁸ Seener (n 654) 20.

⁶⁹⁹ Chow et al (n 22) 14.

⁷⁰⁰ Ibid.

⁷⁰¹ Ibid.

⁷⁰² Ibid.

⁷⁰³ Ibid.

⁷⁰⁴ Seener (n 654) 20.

Iranian government is longer-term energy strategy to mitigate the effects of US sanctions.⁷⁰⁵ Signs to support such a conclusion were evident in 2017 with the announcement that Rosneft, Russia's largest state-owned oil company, had agreed to a joint venture with the NIOC to work on several 'strategic' projects worth an estimated US\$30 billion.⁷⁰⁶ Iran's collaboration with Russia also involves the transfer of technology to develop the nation's refining and petrochemicals operations.⁷⁰⁷

Undoubtedly, the increasing presence of Russia in Iran's energy sector is a by-product of the vacuum created from the lack of Western companies. Both countries have been the target of sanctions by the US and have some common interests in the geo-political dynamics in the Middle East.⁷⁰⁸ In turn, although Russia's oil companies may not necessarily offer access to the best technology and are not regarded as top tier options for foreign investment, increasing the oil production rates in shared border fields particularly is a top priority of the IPC. Even though "Iran's investment needs in its energy sector goes well beyond the upstream oil and gas sector",⁷⁰⁹ any IPCs signed with Russia and potentially with China to improve the financial and technical outcomes for its upstream petroleum projects would be welcome.

5.10 Recommendations

Notwithstanding the 'grey areas' of the IPC which cause some consternation for IOCs, the key ways in which it improves on the Buy Back Contract model used previously by Iran are welcome. The IPC thus represents a major step in the attempt by Iran to re-open the nation to foreign investment. The transition to IPCs by Iran also reflects "a clear shift from the ideological considerations to a more pragmatic approach in the management of the Iranian economy and its strategic resources".⁷¹⁰ However, it is important to concede that "building trust [with IOCs] is essential" if the IPC model is to realise its potential and provide the anticipated benefits to Iran.⁷¹¹ In turn, the comparative analysis of the Buy Back Contract and IPC model above has clearly established that the Buy Back Contract has not achieved its strategic objective to improve the capacity and resilience of Iran's oil and gas sectors by attracting foreign technology and expertise. The contract model was initially developed by

⁷⁰⁵ Ghorbani (n 455) 7.

⁷⁰⁶ 'Rosneft and National Iranian Oil Company Sign Oil and Gas Strategic Cooperation Agreement' https://www.rosneft.com/press/releases/item/188381/ (accessed 14 September 2019).

⁷⁰⁷ Ghorbani (n 455) 7.

⁷⁰⁸ Chow et al (n 22) 14.

⁷⁰⁹ Ibid 16.

⁷¹⁰ Dudlák (n 68) 35.

⁷¹¹ Ibid.

the government in Iran as a potential pathway to accessing foreign money, technology and know-how for the development and production of costly and complicated mining projects.⁷¹² This is because the lack of access to, and utilisation of, such technologies were both hindering the efficiency with which the fields could be developed as well as the capacity of Iran to develop its domestic resource mining capabilities.⁷¹³ The Buy Back Contract model has not however provided Iran with the 'returns' for which it was designed. The comparative analysis above indicated that the IPC model is closer to petroleum contract terms and conditions recognised internationally and provides a more attractive balance between risk and rewards for investors. With greater security around their investment and profit potential than the Buy Back model, the IPC is thus more attractive to IOCs, which, in turn, increase the likelihood of gaining access to foreign technologies and know-how. Another key finding to emerge from the comparative analysis above is that the structure of the Buy Back Contract model did not necessarily incentivise investor companies to maximise petroleum field development and production. The IPC arguably maximises incentives for IOC across low and high risk areas by providing a more robust mechanism for calculating the rate of return based on revenues from petroleum prices.⁷¹⁴ This has the potential to incentivise IOCs to maximise the extraction potential of the field and motivates them to reap the benefits from higher petroleum prices. Moreover, the greater flexibility offered to IOCs on how they transition through the project stages 'encourages field redevelopment and the use of enhanced petroleum recovery techniques in the process'.⁷¹⁵

There are two key points for consideration here. First, when negotiating petroleum contracts, IOCs aim to comprehensively understand the key 'features of contractual elements which serve to respond to the legal, technical and financial challenges involved'.⁷¹⁶ Second, contract managers working for IOCs adopt a comprehensive approach to managing the risks, complexities and uncertainties in international contracts by adopting a dynamic and strategic approach to the identification of risks and the management of the contract lifecycle.⁷¹⁷ IPC By-law Article 3(6) for instance stipulates that the IOC must carry out all operations to achieve the Petroleum Operations objectives while maintaining the well's Maximum

⁷¹² Dallas and Black (n 369) 3.

⁷¹³ Me (n 501).

⁷¹⁴ A. V. Aliyev. 'Post-Sanction Iran and its "Game Changer" Effect in Energy Market' (2018) 49 *The Scientific* and Pedagogical News of Odlar Yurdu University 124, 125.

⁷¹⁵ Ibid 130.

⁷¹⁶ Ebrahimi and Shahmoradi (n 679) 2.

⁷¹⁷ Ibid.

Efficient Rate (MER). Moreover, the flexibility of the IPC is evident around its core principle and objective to facilitate the recovery of petroleum costs. IPC By-law Article 3(3), Petroleum Costs and the Fees, stipulates that all cost are paid to the IOC form the maximum 50% revenue produced from the field following successful petroleum operations. However, IPC By-law Article 3(4) also states that the contract duration can be extended to support cost recovery when the allocated revenues from the field is not adequate for the term of the contract.

The IPC model is also preferable to the Buy Back Contract model because the obligation under the new IPCs for successful IOCs to partner with domestic exploration and production companies benefits domestic entities operating in Iran's resources sectors. Specifically, elements of the IPC favour the partnership (JVA) approach to enhancing project operations and provide greater flexibility around the scope of the project and the management of changes in costs. As such, they arguably provide an acceptable level of protection against claims by the IRGC that a reliance on foreign investor entities 'would damage the country by shutting domestic contractors out of upstream projects and undermining Iran's sovereignty with 20-year contract terms'.⁷¹⁸ A particular concern of the IRGC is that the IPCs undermine the integrity of the resistance economy in Iran by not adequately considering domestic companies and their abilities to successfully manage various aspects of the oil exploration and production projects.⁷¹⁹ Such opposition around the extent to which IOCs are granted 'management rights' and the size of the revenue from sales only serves to illustrate the risks Western companies continue to face regarding the possibility that their assets could be appropriated by the Iranian government "at a political whim".⁷²⁰ As stated by Seener, '[d]ue to religious and ideological reasons, the risk existed that IRGC would prevent companies entering the Iranian market and competing with it'.⁷²¹ However, the more integrated approach by the NIOC (or affiliate) and the IOC on the exploration, development and production operations provides a stronger structure to maximise the alignment of the benefits of the petroleum mining project for both parties. Moreover, it is likely to encourage the development and uses of the best technologies to optimise the outcomes of the operations.

- 720 Ibid.
- 721 Ibid.

⁷¹⁸ Seener (n 654) 20.

⁷¹⁹ Ibid.

The IPC thus differs from the Buy Back in terms of greater flexibility around the exploration phase and the integration of the exploration and development operations. In addition, there is greater flexibility around the development of common fields with neighbouring countries as well as the exploration, development and production operation in 'high risk' regions and deep waters.⁷²²

⁷²² Seener (n 654) 20.

6 CONCLUSION

The aim of this thesis was to critically analyse the current IPC model applied in Iran to petroleum transactions with IOCs. The analysis it presents was framed around three research questions that sought respectively to identify the strengths and weaknesses of the Iran Petroleum Contract in regard to the international petroleum trade (RQ1), how other contract forms (i.e. the Buy Back Contract) adopted by the Iranian Government impact the international petroleum trade (RQ2), and the extent to which the IPC best addresses the current and potential needs of the oil industry in Iran (RQ3). An overarching goal in addressing these questions was then to assess whether the IPC presents as the best contract reform option to Iran to improve outcomes for all stakeholders. Qualitative interpretive methodology was adopted for critical analysis of historical, government, legislative, industry and research literature to draw conclusions about the merits of the IPC in relation to its commercial, economic, and social objectives.

To briefly restate the context of the analysis, IPCs were first conceived in November 2015 as a contract model to replace Buy Back Contracts. The primary objective of the new contract model for Iran, put succinctly by Batmanghelidj,⁷²³ is to attract IOCs to invest and engage in the nation's oil and gas industries so that Iran has access to much needed capital, technology and expertise to optimise the value of its natural resources. On August 3, 2016, the Cabinet of Ministers in Iran passed a resolution permitting the legislation of the terms and structure of the IPC model a risk service contract. As such, the IPC is a combination of a fourth-generation Buy Back Contract and a Production Sharing Contract (PSC) in Iran. According to the contract terms and conditions, contractor entitlements are paid directly from sales revenues (similar to the Buy Back model). As such, the IPC is generally regarded as a service contract and offer IOCs opportunities to invest in upstream oil and gas projects. The stated aim of the Iranian government is to attract upstream capital investment of up to USD\$13 billion by 2020 and up to USD\$200 billion overall.⁷²⁴

Analysis of Iran's petroleum contract models in this thesis both identified and discussed the implications of the high investment costs for IOCs seeking to engage in the exploration and development of oil fields in Iran. When this is balanced against the general lack of financial resources and expertise in Iran to support the development of its resources sectors, attention

 ⁷²³ Esfandyar Batmanghelidj, 'Emerging Privitization in Iran's Energy Sector Deserves a Second Look' (2017)
 Bourse & Bazaar <<u>https://www.bourseandbazaar.com/articles/tag/petropars</u>> (accessed 15 February 2020).
 ⁷²⁴ Seener (n 654) 99.

was drawn to the key points of negotiation between Iran (via the NIOC) and foreign oil companies for resource exploration and production rights. Hence, the main sources of tension between the contracting parties were identified and evaluated in terms of the extent to which the IPC offers either or both of the contracting parties some 'relief' from what they perceived as the high risk or burdensome aspects of the Buy Back Contract model specifically. The following section summarises the main findings reported in this thesis in relation to each of the research questions.

6.1 Strengths and Weaknesses of the Iran Petroleum Contract in regard to the International Petroleum Trade (RQ1)

In terms of strengths, this thesis established that IPCs grant Iran the right to maintain sovereignty over its hydrocarbon reserves while agreeing to pay IOCs for all direct and indirect expenses, and financial and operational costs via the allocation of a portion (up to 50%) of products or proceeds based on current market prices.⁷²⁵ Section 5.3 highlighted that one of the main complaints directed towards the Buy Back Contract model was its lack of flexibility for the IOC.⁷²⁶ Clearly acknowledged in the literature is that in order for Iran to achieve its strategic objective to develop its energy sector via JV petroleum activities with IOCs, the terms of the contract have to encourage the participation of IOCs.⁷²⁷ That is, participation in terms of committing to significant cost outlays during exploration and development particularly, as well as to meeting the long-term energy sector interests of Iran. Compared to the Buy Back Contract model, the IPC is arguably more likely to achieve these objectives due to its greater flexibility. It is a reality of petroleum contract negotiations that no company can predict the price of oil during the lifecycle of the contract. Therefore, the mechanisms within the IPC including its renegotiation clauses and the option to adjust some fiscal formulas on agreement, provide a degree of flexibility to the IOC. This encourages the participation by the IOC as it provides a stronger role in their capacity to direct contract outcomes. Moreover, it allows for both contracted parties to cope with unexpected contingencies, as well as for the NIOC to introduce incentives to promote cooperative practices.⁷²⁸ Lastly, the flexibility embedded into the IPC terms is to the advantage of the NIOC because it provides the space to renegotiate on some contract terms if the IOC fails to

⁷²⁵ Ibid.

⁷²⁶ Li et al (n 528) 43.

⁷²⁷ Nicky Beredjick, *Petroleum Investment Policies in Developing Countries* (Springer Science & Business Media, 1989) 83.

⁷²⁸ Ibid.

observe the contractual mechanisms.⁷²⁹The greater degree of flexibility in the IPC compared to the Buy Back Contract model also potentially benefits the development of the energy sector in Iran by promoting closer working relationships with foreign entities. Providing scope for renegotiation may facilitate the development of more cooperative and reciprocal relationships as business partners.⁷³⁰ Of course, it may also be reasonably argued that more rigid contract models (e.g. Buy Back) with minimal flexibility in the contractual elements are efficient when conditions are 'certain' and reduce the scope for grievances. However, given the 'uncertain' conditions associated with long-term petroleum operations, there is an increased need to ensure a space for compromise and renegotiation between rigidity and flexibility. As demonstrated by the Buy Back Contract model, in conditions of project uncertainty, renegotiations are often requested.⁷³¹ For example, the IOC may want to renegotiate terms if the discovered oil quality is very high. As such, there is the view that the flexibility of the IPC will arguably help to reduce renegotiation-related grievances between the contracted parties throughout the duration of the contract.⁷³²

It has been well-established, including in this thesis, that a leading source of conflict between the host government and the IOC in a petroleum contract is the preoccupation of the IOC with establishing stable and predictable contractual relations. As previously discussed in Section 5.5 of this thesis, the rigidity of the Buy Back Contracts was perceived by many in the Iranian parliament to be highly detrimental to Iran's interests, and often worked to de-incentivise IOCs to expand production.⁷³³ In turn, although the primary goal of the NIOC and the successful IOC will always be divergent; namely, to promote sector and economic growth and to maximise profits, respectively, the more flexible regime of the IPC provides greater opportunity for both parties to achieve their objectives.

The IPC model also affords the Iranian government with the opportunity to make changes to the contract in response to changing conditions, while not precluding the IOC from negotiating for stabilisation clauses to be included in the contract agreement. IOCs would understandably be opposed to any attempt by the host government to initiate amendments to the contract in an effort to optimise government returns, and thus potentially reducing

⁷²⁹ Ibid.

⁷³⁰ Soleimani and Tavakolian (n 681) 4.

⁷³¹ Ibid.

 ⁷³² Philippe Aghion and Lucía Quesada, 'Petroleum Contracts: What does Contract Theory Tell Us?' (International Workshop on Microeconomics Applied to the Energy Industry, 15 December 2011) 1.
 ⁷³³ Bindemann (n 190) 73.

estimated company profits. Nonetheless, the naturally changing circumstances over the duration of an IPC means that its new clauses allow the Iranian government to capture more (or a fairer share in the view of many Iranian politicians) of the economic benefits of the petroleum operations without breaching the contract terms or disregarding the IOC calls for stability and predictability.

Moreover, the unwavering determination of the Iranian government to maintain sovereignty and ownership of its natural resources at all times, is counterbalanced to some extent by the Open Capex provision of the IPC.⁷³⁴ This By-law supports a more flexible approach to capital expenditure based on the field conditions and behaviour as well as the requirement for additional investment.⁷³⁵ As such, the estimated DCC based on the operational requirements and the performance of the field can be more clearly defined. This helps to the protect the rights of the IOC when changes and adjustments are made to the levels of production.⁷³⁶ Hence, if the NIOC initiates action to limit production due to political or social reasons, the restrictions will not impact the entitlement of the IOC to recover costs and fees. This mechanism, along with other flexibilities around local content requirements and the scope of work in the IPC provide greater ability to the IOC to navigate the different stages and phases of the petroleum operations.

The structure of the IPC has also been found to carry potential advantages to Iran and its participation in the international petroleum trade. The JV component that is embedded in the IPC also encourages participation in the production of oil fields to support national industrial and economic development.⁷³⁷ This is an upgrade on the Buy Back Contract model where the level of State participation was limited. The service nature of the IPCs allows for greater participation in the exploitation of the fields and thus more control in the management and utilisation of its resources. Moreover, the structure of the IPC allows the parties to the contract to determine commercial, political, and other objectives as the outcome goals of the contracts. As the host nation, the Iranian government can establish specific goals and objectives as the criteria entering into a contract with an IOC. Therefore, the government is in a position to establish criteria that act as fundamental principles to govern the IPCs according to domestic laws including the Constitution.⁷³⁸ However, contractors are also afforded the

⁷³⁴ IPC by-laws art 1(27).

⁷³⁵ Ibid art 1(28).

⁷³⁶ Ibid art 3(10).

⁷³⁷ Ibid.

⁷³⁸ Soleimani and Tavakolian (n 681) 3.

right to establish certain criteria as an incentive to enter into negotiations. Such objectives generally refer to addressing apparent geological (e.g. exploitability of the reservoir), financial (e.g. ROR), political (e.g. economic sanctions), and social (e.g. Corporate Social Responsibility) risks.⁷³⁹

6.1.1 The long duration of the IPC and the impact on its efficacy in the context of article 81 of the Iranian Constitution

Article 81 of the Constitution of Iran sets out that it is "absolutely forbidden" for foreigners to have rights "to establish companies or institutions in commercial, industrial, and agricultural fields, as well as in mines and in the service sector".⁷⁴⁰ As worded, the article forbids completely the concessionary system with the terms "absolutely forbidden" as understood in the context of the Constitution meaning that any such attempt by an IOC cannot be legitimised even if it is approved by the Iranian Parliament.⁷⁴¹ In addition to not letting the government of Iran from agreeing to concessions, it restricts the government from affording foreign entities the right to establish companies or institutions that deal with the extraction of mineral resources.⁷⁴² Thus, the government of Iran alone has the authority to legitimately deal with natural resources. It can be then inferred from the Constitution of Iran that concessions contracts, PSAs, JOAs or any other type of contract involving foreign participation and control (see Article 44), ownership (see Article 45), or the establishment of foreign companies (Article 81) are not permitted.⁷⁴³ It is worth briefly considering, however, the implications of this lack of resource ownership and control rights for IOCs for the overall efficacy of the IPC model.

This thesis has established the different phases of petroleum projects and that the length of the exploration and production phases especially can present significant challenges to the planning and implementation of operational activities. A weakness identified in the Buy Back Contact model was that it included a fixed cap on obligations and that the IOC was required

⁷³⁹ Ibid.

⁷⁴⁰ Constitution of Iran (n 217) 18.

⁷⁴¹ Shahri (n 438) 121.

⁷⁴² Ibid.

⁷⁴³ Giuseppe Sammarco and Simone Tagliapietra, 'The Political and Commercial Barriers to the Development of Iran's Gas Market', in *Iran after the (Potential) Nuclear Deal: What's Next for the Country's Natural Gas Market*? Fondazione Eni Enrico Mattei (FEEM) 14.

to implement a MDP while simultaneously trying to add flexibility to how they transitioned through the phases (i.e. exploration, development, production, and site abandonment).⁷⁴⁴ On the one hand, the IPC is a arguably better option than the Buy Back Contract model because it more clearly defines the phases and stages in which the foreign company conducts its operations. The main activities involved in the life cycle of the petroleum product include upstream exploration and production process, along with midstream and downstream activities carried out via prospecting, drilling, developing, refining, marketing, sales, and the like. Because the JVA in an IPC can help to define more accurately the IOCs involvement, greater clarity can be achieved around the extent to which it participates in exploration and production, or just during the exploration or development phases.⁷⁴⁵ On the other hand, given that IPCs are long-term contracts of up to 15 to 20 years, their long-term duration arguably has the potential to undermine its efficacy when considered in the context of Article 81. International oil companies are obliged in the terms and conditions of the IPC to hand over management of the petroleum mining project to domestic companies as soon as the latter has acquired the necessary capacity and capabilities to manage the projects.⁷⁴⁶ While the domestic companies on the list as potential partners may be public, private, or semi-private, the extent to which private companies especially "function independently from government agencies or political interests" remains an area of concern for IOC.⁷⁴⁷ In addition, although IPCs permit booking reserves by IOCs in some instances this does not equate to the transferring of ownership of the petroleum field to the IOC.748

These contractual conditions in the context of the long-term duration of the contract may undermine the efficacy of the IPC and potentially constrains the operations of IOCs who would prefer to maintain control over the project for as long as possible.⁷⁴⁹ In turn, a number of reasons can be suggested for drawing this conclusion. Domestic companies in Iran are generally unable to contribute meaningful financing to petroleum projects.⁷⁵⁰ This forces IOCs to contribute a significant portion of the investment for potentially a longer duration

⁷⁴⁴ Kuhn and Jannatifar (n 571) 15; Me (n 501) 22.

⁷⁴⁵ Ebrahimi and Shahmoradi (n 679) 6.

⁷⁴⁶ Chow (n 22) 5.

⁷⁴⁷ Ibid.

⁷⁴⁸ Bi-Weekly News & Analysis, No. 1, International Law Office of Dr. Behrooz Akhlaghi & Associates < <u>http://intllaw.net/wp-content/uploads/2014/06/Iran-Petroleum-Contract-IPC-1st-Bi-Weekly-N-A-April-07-2014.pdf</u> > 6.

⁷⁴⁹ Ibid.

⁷⁵⁰ Chow (n 22) 4.

than in previous contract forms. In addition, the long-term duration of the IPC exposes the IOC "to substantial risk for an extended period", from the initial outlay of capital to get the exploration project running, to the ongoing appraisal and development costs.⁷⁵¹ These are costs which the IOC naturally aims to recoup from earnings.⁷⁵² As a result, it may be the case that IOCs could look to withhold technologies or the transfer of know-how to local companies to defer for as long as possible the development of their capacity and capabilities to assume control of the project. This arguably has the effect to undermine the efficacy of the IPC in terms of its capacity to deliver the best outcomes for Iran's energy sector in a timely and most cost-effective way.

6.1.2 Establishing stable and predictable contractual relationships: The role of stabilisation clauses

Another important consideration in relation to the efficacy of the IPC to deliver optimal outcomes for Iran's energy sector is the is the extent to which all parties can benefit from its flexibility provision. A s previously established (see section 5.7.4) the IPC By-laws provide the Iranian government with more control over the operations through additional powers to the NIOC to specify KPIs around the commercialisation of the field.⁷⁵³ These KPIs increase the capabilities of the NIOC to identify and address operational issues and to promote a more collaborative relationship with the IOC to achieve strategic objectives.⁷⁵⁴ The NIOC or its affiliated companies may also benefit from greater visibility of the operational outcomes. Furthermore, it is reasonable to suggest that the inclusion of clauses in the IPC for extensions of time due to unpredicted factors and force majeure provides for a contract mechanism to support the management of apparent financial and operational risks due to delay, disruption, and requests for time extensions. Yet, the reality of the implementation context of IPCs is that ownership of the petroleum remains in the hands of the state, IOCs contribute the required capital and expertise for petroleum exploration and exploitation, and state-owned oil companies have only a minor role.⁷⁵⁵ As a result, IOCs and the Iranian Government must reach agreement on the processes for the development of the petroleum resources; that is,

⁷⁵¹ Margarita Coale, 'Stabilization Clauses in International Petroleum Transactions' (2020) 30(2) *Denver Journal of International Law & Policy* 218.

⁷⁵² Ibid.

⁷⁵³ Soleimani and Tavakolian (n 681) 3.

⁷⁵⁴ Ebrahimi and Shahmoradi (n 679) 6.

⁷⁵⁵ Coale (n 756) 218.

they must try to establish stable and predictable contractual relationships.⁷⁵⁶ One way for IOCs to try to minimise their investment risks is via contract provisions, with stabilisation clauses specifically aiming to address political risk.⁷⁵⁷ Such political risks of concern to IOCs include decisions by the host State to expropriate or nationalise the IOC's operation via legislation directly. Alternatively, more indirect control mechanisms may be enforced such as changes in labour laws, accounting rules leading to increased taxes, or the mandating of domestic service and supply contracts and the like which potentially constrain the IOC's capacity to make a profit.⁷⁵⁸

Stabilisation clauses have the specific objective to "secure the [contract] agreement against future government action or changes in law".⁷⁵⁹ In this way, they specify a contractual commitment by the host State to not alter the contract terms, by legislation, regulation or otherwise, without the consent of the other party or parties to the contract.⁷⁶⁰ However, the extent to which such stabilisation clauses permit the Iranian government to benefit from the flexibility embedded into the IPC depends to some degree on how the clauses are treated at arbitration. Stabilisation clauses are "a contractual and domestic law matter" and as such there is inherent uncertainty as to the extent the IOC can successfully rely on the clause should it seek investment treaty arbitration and recourse under international law.⁷⁶¹ As Coale points out, however, stabilisation clauses "are taken seriously by arbitrators" who rely less on the precision of the language used in the clauses and more on the "limitation on sovereignty expressed by such a clause in light of all the circumstances surrounding the transaction".⁷⁶² The flexibility produced through the stabilisation clauses is generally towards IOCs being able to adapt financially to changing pressures and demands and to taxation triggers (i.e. after investment recovery).⁷⁶³ Arguably, there is a general benefit to the Iranian Government from

⁷⁵⁶ Ibid.

⁷⁵⁷ Coale (n 756) 220.

⁷⁵⁸ Ibid.

⁷⁵⁹ Christopher T. Curtis, 'The Legal Security of Economic Development Agreements' (1988) 29(1) Harvard International Law Journal 346.

⁷⁶⁰ Michelle Flores, 'A Practical Approach to Allocating Environmental Liability and Stabilizing Foreign Investment in the Energy Sectors of Developing Countries, (2001) 141(1) *Colorado Journal of International Environmental Law and Policy* 159-161.

⁷⁶¹ Andrey V Kuznetsov, 'The Limits of Contractual Stabilization Clauses for Protecting International Oil and Gas Investments Examined Through the Prism of the Sakhalin-2 Psa: Mandatory Law, The Umbrella Clause, and the Fair and Equitable Treatment Standard' (2015) 22(2) *Willamette Journal of International Law and Dispute Resolution* 268.

⁷⁶² Coale (n 756) 236-237.

⁷⁶³ Eli Lauterpacht, 'Issues of Compensation and Nationality in the Taking of Energy Investments' (1990) 8(1-4) *Journal of Energy and Natural Resources* 243-244.

the inclusion of stabilisations clauses in that the flexibility they imply for the IOC provides the IOC with a greater confidence in the contract and hence commitment to the petroleum project. That is, "stabilisation clauses enhance certainty and predictability which are key ingredients for the success of long-term investment projects".⁷⁶⁴ Regarding petroleum exploitation specifically, given that they are capital intensive, and it takes a long time for the IOC to recoup its investment, the prospect of changes to the laws of the host country can give a degree of economic uncertainty and unpredictability over the project. Stabilisation clauses which constrain the host country's legislative prerogative to change laws unilaterally help to return a degree of certainty and predictability to the project for the IOC which can, in turn, benefit the host country through better project outcomes.⁷⁶⁵

In terms of perceived or potential 'limitations on sovereignty', the Iranian Government may look to ensure two key terms and conditions are in place to also benefit from the flexibility afforded the IOC. First, the economic terms and conditions of the IPC should be structured in such a way as to permit the Iranian Government to start to receive some of the agreed revenue early and which do not grant the IOC unrestricted cost recovery at the expense of Iran's petroleum resources.⁷⁶⁶ Second, there is a benefit to the Iranian Government of having local companies as joint partners with the IOC in the project from its outset. This provides the Government with a voice in the project and helps to ensure that any stabilisation provisions are specific in terms of the types of laws and measures they cover. This gives the Iranian Government the opportunity to 'test' the proposed stabilisation provisions during negotiation rather than at a point when significant project costs have accrued.⁷⁶⁷

Hence, by making explicit each party's obligations regarding the completion of the project, mechanisms can be put in place around the requirements for providing timely notice of any changes in time or financial provisions when things do not go as expected.⁷⁶⁸ Such consideration of the possible causes of disruptions or delays to operations and the potential for time extensions when drafting the contract may help to avoid contract disputes. That is, during contract negotiations the terms and conditions for how to define, interpret, and monitor delays and disruptions in accordance with the contract provisions can be established,

 ⁷⁶⁴ Deloitte, 'Stabilisation Clauses in International Petroleum Contracts Illusion or safeguard?' (2014)
 <u>https://www2.deloitte.com/content/dam/Deloitte/ug/Documents/tax/tax_StabilisationClauses_2014.pdf</u>> 9.
 ⁷⁶⁵ Ibid.

⁷⁰⁵ Ibid.

⁷⁶⁶ Kuznetsov (n 766) 268.

⁷⁶⁷ Ibid.

⁷⁶⁸ Ibid.

along with other key aspects including the 'extension of time' entitlements afforded the IOC, penalties, and rescheduling commitments.⁷⁶⁹

There are, however, several issues for the Iranian parliament to consider. It has been demonstrated historically that participation of the State in mining operations reduces the attractiveness of the contract option to foreign entity private investors. Additionally, the involvement of the State increases opportunities for patronage and corruption. This is typically when a domestic entity is imposed upon an IOC aiming to enter into JVAs with Iran.⁷⁷⁰

6.1.3 Meeting legal, economic, and social objectives

This thesis also reported a number of weaknesses in the IPC model in regard to its capacity to achieve the desired legal, economic and social objectives. Although the Iranian government has legally ratified the tendering process for petroleum projects under the IPC model, the analysis revealed that risks remain for successful IOC investors. Indeed, despite the introduction of IPCs as a new contractual and fiscal framework, oil contracts in Iran remain both complicated and remain a point of intense political debate over the extent to which foreign oil companies should have access to the nation's energy resources.⁷⁷¹ To attract investors to inject investment into the Iranian energy sector, these outcomes must nonetheless be facilitated by transparency in institutional processes and the ability to forecast new petroleum contracts.⁷⁷² Moreover, for Iran to provide itself with the best opportunity to have the IPC address its oil industry and economic needs it is vital for its parliament to administer comprehensive oversight of State-owned entities. This is needed to ensure that they demonstrate the efficient management of the petroleum resources and support the development goals of the nation.⁷⁷³ The IPC model permits the insertion of provisions to address these concerns. For example, IPC By-law Article 2(7) stipulates that the Contractor must fund and perform the petroleum operations on behalf of the NIOC ...

In addition, a key finding to emerge from the comparative analysis of the Buy Back Contract and IPC models is the potentially disruptive role that domestic political pressure, historical

⁷⁶⁹ Ghorbani (n 455) 4.

⁷⁷⁰ Ebrahimi and Shahmoradi (n 679) 5.

⁷⁷¹ Jalilvand and Ebert Foundation (n 499) 8.

⁷⁷² Omid Shokri Kalehsar and Azime Telli, 'The Future of Iran-Russia Energy Relations Post-Sanctions' (2017) 24(3) *Middle East Policy*, 164.

⁷⁷³ The Natural Resource Governance Institute, 'State Participation in Oil, Gas and Mining' (2015), <<u>http://www.resourcegovernance.org/sites/default/files/nrgi_state_owned_Briefing_Eng_20150811.pdf</u>> (accessed 22 January 2020).

grievances held by IOCs, and geo-political tensions (manifest as sanctions against Iran by the US) may impact the negotiations for the contracts. 'Risks' and 'profit potential' are the main factors guiding the decision making by IOCs as to whether or not to pursue resource mining projects in Iran. As such, the recent re-establishment of sanctions on Iran by the Trump administration presents as a major obstacle to understanding the true extent to which IPCs strike the best balance in meeting the needs of both Iran and foreign investor companies to optimally develop the nation's petroleum sector.

6.2 Impact of the Buy Back Contract Model on Iran's Participation in the International Petroleum Trade

The introduction of the IPC is significant for its attempt to address the inadequacies of the Buy Back Contract model to attract international investment in the energy sector.⁷⁷⁴ In particular, the legal framework of the IPC defines the interests and commitments of the contracting parties are more explicit and transparent in expressing the objective to protect and increase the interests of Iran, while also meeting the interests of IOCs investing in the sector.⁷⁷⁵ Indeed, the Buy Back Contract Model demonstrated the need for flexibility within the mechanisms of long-term petroleum contracts, even beyond clauses that permit renegotiation, in order to limit their rigidity.⁷⁷⁶ The Iranian government has initiated the IPC as a form of production sharing agreement to promote more flexibility and adaptability in their relationship with IOCs.⁷⁷⁷ That is, IPCs make it easier to adapt the production balance to match the primary petroleum interest of each party as well as to accommodate the local conditions. In this way, greater scope for movement within the conditions of the contract is achieved by adjusting the production share of the IOC. The Buy Back Contracts, as a traditional form of a concession license agreements, did not include the provisions to make such adjustments and thus was not an attractive option for IOCs. As discusses above in Section 2.3.2, the capacity for adjustment was manufactured in the Buy Back Contract via a progressive royalty arrangement that relied on profitability indicators to guide adjustments. In contrast, the provisions in PSAs, including the recent IPC model, to provide greater scope for

⁷⁷⁴ Ibid 165.

⁷⁷⁵ Ibid.

⁷⁷⁶ Bright Erakpoweri Okogu, 'The Middle East and North Africa in a Changing Oil Market' (International Monetary Fund, 2003) 40.

⁷⁷⁷ Ebrahimi and Shahmoradi (n 679) 13.

adjustments to changing conditions is achieved via non-linear arrangements to share profitoil.⁷⁷⁸

6.2.1 Buy Back Contract model's limited fiscal system

The limitations around the fiscal system controlling the Buy Back Contract model also impacted the Iran's participation in the international petroleum trade. The progressive tax schemes embedded into the IPC provide the means to achieve more stable and flexible conditions required for the fair sharing of oil profits between the IOC and the host government. Significant profits can be achieved from petroleum projects and the Buy Back Contract model demonstrated the need for there to be fiscal systems in place that can ensure the profits are divided fairly.⁷⁷⁹ It may be argued that the IPC model demonstrates the Iranian governments awareness of the importance of meeting these demands as its design includes a variable fiscal regime whereby the government return is adjusted according to profitability levels. As such, the sliding scale mechanism and profit-oil splits in the IPC provide flexibility to account for the volatility of the global price of crude oil.

Therefore, the Buy Back Contract was found to inadequately meet the demands of IOCs for international contract forms and global practices that permit some flexibility, including the scope for renegotiation, in how outcomes are achieved, and profits divided up. Any contract reform initiative to address these types of demands from foreign investors is arguably beneficial to Iran's energy sector because it can include provisions to revise the terms of the contract in response to changing economic and market conditions not predicted when agreeing to the contract. Moreover, relying on a model with greater flexibility than the Buy Back Contract (i.e. the IPC) to facilitate its participation in the international petroleum trade it arguably benefits the energy sector in Iran because it provides IOCs with stronger investment security. By being able to renegotiate contractual terms the IOC will feel a greater sense of certainty that it can adapt to changing conditions as well as incentivised to increase production capacity.⁷⁸⁰

The 'advantage' of the IPC compared to the Buy Back Contract model is also apparent in the inclusion of By-laws⁷⁸¹ that allow for a review and possible adjustment to the Development Plan. As described in Section 5.3, the development plan in an IPC functions as a type of

⁷⁷⁸ Aghion and Quesada (n 737).

⁷⁷⁹ Ebrahimi and Shahmoradi (n 679) 13.

⁷⁸⁰ Hadiza TijjaniMato, 'The Role of Stability and Renegotiation in Transnational Petroleum Agreements' (2012) 5(1) *Journal of Politics and Law* 33.

⁷⁸¹ IPC by-laws art 1(14).

predictive plan for the lifecycle of the petroleum operations (e.g. underlying assumptions about the project, strategies to achieve desired outcomes, presumed main activities and the like).

As such, it forms the basis of both the prepared submission on the field development and production activities and any adjustments to be undertaken during the lifecycle of the operations. The capacity to make such adjustments thus reduces the rigidness of the contract model, with the focus during initial contract negotiations on establishing the principles of the contract arrangement.⁷⁸² Previous Buy Back Contract models demonstrated the complications to operations and the potential for conflicts between the contract parties when there are unanticipated changes in field development costs. By offering greater fiscal flexibility via fee per barrel of production,⁷⁸³ the IPC is more accommodating of both parties' needs.

6.2.2 Technology and knowledge transfer

This thesis has discussed at length the Buy Back Contract model and the perceived inefficiencies it engendered across the energy resources sector in Iran due to its lack of flexibility and appropriate mechanisms to incentivise IOCs towards increased production. In turn, to provide greater potential for the development of the sector than what was being provided by the Buy Back Contract model, several articles are contained in the IPC model to improve the efficiency of operations. The requirement in IPC By-law Article 4(2) for potential IOCs to maximise their use of qualified and experienced Iranian workers to conduct the petroleum operations provides the platform for the development of local content in accordance with Iranian law.

In terms of the transfer of technology and know-how into Iran, the Buy Back Contract model afforded the NIOC with a relatively ineffective mechanism for delivering cost-effective and targeted training and development to Iranian employees working on the oil mining industry In contrast, the provision in the IPC that the NIOC has final approval as to the type and level of training for Iranian employees provides it with a more effective mechanism than Buy Back Contracts for the know-how and technology transfer.⁷⁸⁴ As a result, the government in Iran can attempt to coordinate the upgrading of Iranian employee knowledge and skills to perform the responsibilities and tasks needed for successful petroleum operations. IPC By-law Article 4(3) on joint R&D mechanisms further illustrates this point. The IOC is required to

⁷⁸² Ebrahimi and Shahmoradi (n 679) 13.

 $^{^{783}}$ IPC by-laws art 1(20) and art 6(2).

⁷⁸⁴ Ebrahimi and Shahmoradi (n 679) 6.

incorporate R&D proposals into the annual Work Program and Budget of the project. As such, it is reasonable to conclude that this provides the Iranian government with an additional lever to pull to promote the modernization of the energy sector.

6.3 The Extent to which the Iran Petroleum Contract Addresses the Current and Potential Needs of the Oil Resources Industry in Iran (RQ3)

A major concern of Iran as a host nation is with regard to the ownership and sovereignty of its natural resources. These two legal requirements are considered in close relationship with one another and host governments typically aim to achieve most of the advantages from these two factors.⁷⁸⁵ A key component within the IPC structure to best address the current and potential needs of the oil resources industry in Iran is the JVA structure. The JVAs under which the IOCs operate include joint operating companies, a Joint Management Committee and related contractual structures. One positive for consideration is the by-law ratified by the President stipulating the basic rules and principles of governance for the IPC.⁷⁸⁶ Iran has the bargaining power in the JVA structure because it has ownership of and sovereignty over the oil reserves and can thus use them in international markets. This is primarily because Contract models based on state participation in upstream and downstream petroleum sectors create a continuum of State control over the oil reserves.⁷⁸⁷ Pressure from the upstream network is an important consideration in oil production. Significant investment is made over the life-cycle of an upstream network to preserving network flow. As a result, an upstream oil system is often managed by one operator only to ensure that there is coordination in how the fields are managed. Such a coordinated approach to the operations and management of the fields ensures that the entire network maintains a smooth flow. To achieve this outcome the upstream network must be centrally managed.⁷⁸⁸ Iran's reliance on IOCs working in joint venture with a domestic operators will ensure the national interests are met in terms of access to adequate resources for domestic use, as well as support the growth of the private oil industry sector.⁷⁸⁹ Current trends favouring vertical and horizontal integration illustrate the integrated management of joint operation arrangements within supply chains. JVAs have persisted as an operational option within the oil industry in

⁷⁸⁵ Ibid 5.

⁷⁸⁶ Ibid 3.

⁷⁸⁷ Ebrahimi and Shahmoradi (n 679) 3.

⁷⁸⁸ Ibid 3.

⁷⁸⁹ Mohammad Maleki, M. Eslami, and M. Rezaei, 'Principles of Arbitration in Oil Contracts' (2018) 5(14) *Revista Publicando* 779, 780.

Iran for one main reason. The significant investment costs tied to upstream exploration, development and production projects means it is difficult for one entity alone to develop and finance the entire project. JVAs are entered into as a way to lower the financial burden and to share the risks.⁷⁹⁰ As a result, IPCs help to ensure that there are adequate oil resources for domestic use – arguably the first priority of the Iranian government – as well as provide access to the international market.

Furthermore, IPCs best address the current and potential needs of the oil resources industry in Iran because the level of State participation provided through the JVA structure means the host nation has a fiscal tool at its disposal and a more robust platform from which to promote national development goals.⁷⁹¹ With the NIOC as a party to the IPC and the State representative awarding rights to third parties, there is more direct national involvement in the management and utilisation of the national oil resources.

The affordance of the IPC for greater State participation in key extractive projects via the involvement of State-owned entities is a more positive outcome for Iran than that afforded by the Buy Back Contract. With constructive and considered participation, Iran can generate significant financial returns, encourage capacity building, and develop its capabilities around the monitoring of its oil sector.⁷⁹²

As has been established throughout this thesis, the main objective the Iranian government in issuing IPCs are to gain access to project financing and to develop its energy sectors. IPCs provide the government with a new set of fiscal tools including taxes and levies to achieve a satisfactory financial return to achieve its objectives for the sector. Moreover, stipulations in the IPC around the participation of the State endorse local content arrangements thus may help to meet the sector needs in Iran. The IPC By-laws for instance (e.g. Article 4[1]) increases the ability of Iranian companies to participate in oil development and production projects. Furthermore, they gain the experience of operating in regional and international markets and can learn the skills and approaches needed to manage competently the oil fields. In sum, the IPC addresses the two primary (legal) concerns of the Iranian government; namely, ownership and sovereignty over the nation's natural resources, by way of a joint arrangement structure (i.e. JVA of JOCs).⁷⁹³ Primary consideration is given in IPC By Laws

⁷⁹⁰ Ebrahimi and Shahmoradi (n 679) 3.

⁷⁹¹ Marta Peirano and Johannes Grenzfurthner, *Oil Contracts: How to Read and Understand a Petroleum Contract* (Times Up Press, 2012) 24.

⁷⁹² Maleki et al (n 795) 781.

⁷⁹³ Bindemann (n 190) 73.

to ensuring Iran's petroleum (and other natural) resources remain under the ownership and sovereignty of the government rather than a private entity as stipulated in the Constitution. The participation of the state is facilitated in IPCs using JOA, JOC, JMC mechanisms and rotational management procedures.⁷⁹⁴ The IPC aims to attract IOCs to participate in the development of Iran's energy sector through the exploration and production of petroleum products. Lessons learned from the Buy Back Contract era however have highlighted that international companies in possession of the resources and expertise desired by Iran to exploit its energy resources, have strong negotiation positions when determining the terms of the contract. As such, the IPC model includes greater flexibility in its mechanisms to manage the fiscal, technical, production and technology transfer objectives. In turn, the different project phases embedded into the IPC provides the contract with a stronger platform from which to manage the project operations and all of its moving parts.

⁷⁹⁴ Ibid.

7 **BIBLIOGRAPHY**

7.1 Books

Afshar, Haleh, Iran: A Revolution in Turmoil (Springer, 1985)

Alavi, Hamed, History of Oil Industry in Iran (California Institute of Asian Studies, 1978)

Alramahi, Mohammad, Oil and Gas Law in the UK (Bloomsbury, 2012)

Atkin, M, Russia and Iran, 1780-1828 (University of Minnesota Press, 1980)

Barkhuizen, Gary and Rod Ellis, *Analysing Learner Language* (Oxford University Press, 2005)

Benton, Ted, *Philosophical Foundations of the Three Sociologies (RLE Social Theory)* (Routledge, 2014)

Beredjick, Nicky, *Petroleum Investment Policies in Developing Countries* (Springer Science & Business Media, 1989)

Blinn, Keith W, et al, International Petroleum Exploration and Exploitation Agreements: Legal, Economic and Policy Aspects (Barrows Co. Inc., 2nd ed., 2009)

Burns, Robert E, Introduction to Research Methods (Longman, 4th ed, 2000)

Cohen, Louis, Lawrence Manion and Keith Morrison, *Research Methods in Education* (Routledge, 7th ed, 2007)

Corbin, Juliet and Anselm Strauss, 'Strategies for Qualitative Data Analysis' in *Basics of Qualitative Research. Techniques and Procedures for Developing Grounded Theory* (Sage, 4th ed, 2014)

Creswell, John W, Qualitative Inquiry and Research Design: Choosing Among Five Approaches (Sage, 2012)

Curzon, George N, Persia and Persian Question (London, 1892)

Eden, Richard J, Michael Vivian Posner, Richard Bending, Edmund Crouch and Joe Stanislaw, *Energy Economics: Growth, Resources and Policies* (CUP Archive, 1981)

Elm, Mostafa, *Oil, Power, and Principle: Iran's Oil Nationalization and Its Aftermath*, (Syracuse University Press, 1994)

Elwell-Sutton, Lawrence Paul, Persian Oil: A Study in Power Politics (Lawrence & Wishard Ltd, 1955)

Farshadgohar, Nasar, A Survey on Iran's Oil Agreements (Seyri Dar Gharardadhaye Naftiye Iran) (Tehran, 1962)

Fesharaki, Fereidun, Development of Iranian Oil Industry (New York, 1976) 13.

Guilak, Hooshang, Fire Beneath the Ashes: The United States and Iran: a Historic Perspective 1829–1947 (Xlibris Corporation, 2011)

Haery, Mohsen S, *The Legal Implication of the Dispute Over the Iranian Nationalization of the Oil Industry* (University of California, Berkeley, 1955)

Jennings, Anthony, Oil and Gas Exploration Contracts (Sweet & Maxwell Publish, 2002)

Johnston, Daniel, International Petroleum Fiscal Systems and Production Sharing Contracts (Pennwell, 1994)

Kambin, Parviz, *A History of the Iranian Plateau: Rise and Fall of an Empire* (iUniverse, 2011)

Kardel, Mahmoud F, The Development of Iran's Upstream Oil and Gas Industry: The Potential Role of New Concession Contracts (Routledge, 2020)

Katouzian, Homa, *Musaddiq and the Struggle for Power in Iran* (I.B. Tauris & Co Ltd., 1990)

Kuhn, Maximilian, Enabling the Iranian Gas Export Options: The Destiny of Iranian Energy Relations in a Tripolar Struggle over Energy Security and Geopolitics (Springer Science & Business Media, 2014)

Lim, Chin Leng, Jean Ho and Martins Paparinskis, *International Investment Law and Arbitration: Commentary, Awards and other Materials* (Cambridge University Press, 2018).

Mahdavi, Amir H, *History of the Iranian Foreign Relations (Tarikhe Ravabete Kharejiye Iran)* (Tehran, 1985)

Mahdavi, Paasha, 'Oil, Monarchy, Revolution and Theocracy: A Study on the National Iranian Oil Company', in David G. Victor, David R. Hults and Mark C. Thurber (eds) *Oil and Governance: State-Owned Enterprises and the World Energy Supply* (Cambridge University Press, 2012)

Marcel, Valerie, *Oil Titans: National Oil Companies in the Middle East* (Brookings Institution Press, 2006)

McMurry, John E, Organic Chemistry with Biological Applications (Cengage Learning, 2014)

Mecini Paolo and Ezio Mesini, *The Petroleum Upstream Industry: Hydrocarbons Exploration and Production* (Encyclopaedia of Life Support System, 2011)

Mobaser, Dariush, et al, Autopsy of the Oil Contract (Buy-Back) (2000)

Mohebi Mohsen, Legal Issues Facing the Oil and Gas Sectors in the Light of International Arbitration (Publication of Knowledge, 1967)

Mousavi, Seyed Hassan, A Legal Analysis of Service Contract in Oil and Gas Sector in Iran focusing on IPC (GanjeDanesh, 2017)

Muhiberger, Steve, History of Islamic Civilization (Nipissing University, 1999)

Pahuja, Sundhya, Rethinking Iran and International Law: The Anglo-Iranian Oil Company Case Revisited. In James Crawford, Abdul Koroma, Said Mahmoudi, Alain Pellet (eds) *The International Legal Order: Current Needs and Possible Responses: Essays in Honour of Djamchid Momtaz* (Brill, 2017)

Peirano, Marta and Johannes Grenzfurthner, *Oil Contracts: How to Read and Understand a Petroleum Contract* (Times Up Press, 2012)

Rabiei, Faranak, Contracts Law (Behnami, 2002)

Radon, Jenik, 'The ABCs of Petroleum Contracts: License-Concession Agreements, Joint Ventures, and Production-Sharing Agreements' in Svetlana Tsalik and Anya Schiffrin, *A Reporter's Guide to Energy and Development of Open Society Institute* (Open Society Institute, 2005)

Guy Redden, *Questioning Performance Measurement: Metrics, Organizations and Power* (Sage, 2019)

Rees, Judith, The International Oil Industry (London, 1987)

Rivlin, Paul, 'Leverage of Economic Sanctions: The case of US Sanctions Against Iran, 1979–2016' in Wigell, Mikael, Sören Scholvin and Mika Aaltola (eds) *Geo-Economics and Power Politics in the 21st Century* (Routledge, 2018)

Silverman, David, Interpreting Qualitative Data: Methods for Analysing Talk, Text and Interaction (Sage Publications, 1993)

Skrine, Catherine, The World War in Iran (Constable, 1962)

Smith, Benjamin, *Hard Times in the Lands of Plenty. Oil Politics in Iran and Indonesia*. Cornell University Press, 2007) 23

Smousavi, Shaghayegh, 'Iran' in Schwartz, David L (ed) *The Energy Regulation and Markets Review - Edition 8* (The Law Reviews, 2019)

Speight, James G, Handbook of Offshore Oil and Gas Operations (Elsevier, 2014)

Stevens, Paul, *Prospects for Iran's Oil and Gas Sector* (Chatham House for the Royal Institute of International Affairs, 2015)

Svalova, Valentina, Risk Assessment (BoD - Books on Demand, 2018)

Tabari, Nima M, Lex Petrolea and International Investment Law: Law and Practice in the Persian Gulf (Routledge, 2016)

Ule, Christian and Alexander Brexendrorff, Investing in the Oil & Gas Industry (Mena, 2005)

Yinka Omorogbe, *The Oil ana Gas Industry: Exploration and Production Contracts* (Malthouse Press, 1997)

Zamani, Ali, *Challenges in the Development of Iran's Oil Industry* (Desert Publications, 1960)

Zoghi, Iraj, Economic and Political Issues of Iran's Oil (Masael Eghtesadi va Siyasi Naft Iran) (Pazhang, 1962)

7.2 Journal Articles

Abdelrehim, Neveen, Josephine Maltby and Steven Toms, 'Corporate Social Responsibility and Corporate Control: The Anglo-Iranian Oil Company, 1933–1951' (2011) 12(4) *Enterprise & Society* 824

Abdelrehim, Neveen and Steven Toms, 'The obsolescing bargain model and oil: the Anglo-Iranian Oil Company 1933–1951' (2017) 59(4) *Business History* 554.

Ahmadi, Amir, 'The Impact of Economic Sanctions and the JCPOA on Energy Sector of Iran' (2018) 13(5) *Global Trade and Customs Journal* 198

Alavinasab, Seyed Mohammad, 'Determinants of Foreign Direct Investment in Iran' (2013)
3(2) International Journal of Academic Research in Business and Social Sciences 258

Alimohammadlou, Moslem and Abbas Bonyani, 'Iran's Energy Policy after the Nuclear Deal for Cooperation with Foreign Oil and Gas Companies' (2019) 12(2) *International Journal of Procurement Management* 199

Aliyev, A V, 'Post-Sanction Iran and its "Game Changer" Effect in Energy Market' (2018)49 *The Scientific and Pedagogical News of Odlar Yurdu University* 124

Animah, Isaac and Mahmood Shafiee. 'Condition Assessment, Remaining Useful Life Prediction and Life Extension Decision Making for Offshore Oil and Gas Assets' (2018) 53(1) *Journal of Loss Prevention in the Process Industries* 20

Ardalan, Farzam, Nejad Ali Almasi and Mansour Atasheneh, 'Effects of Contractor and
Employer's Obligations in Buy Back Contracts: Case Study of Oil Exporting Country' (2017)
5(2) Entrepreneurship and Sustainability Issues, Entrepreneurship and Sustainability Center
345

Aryanpur, Vahid, Mohammad Saeid Atabaki, Mousa Marzband, Pierluigi Siano and Kiarash Ghayoumi, 'An Overview of Energy Planning in Iran and Transition Pathways Towards Sustainable Electricity Supply Sector' (2019) 112 *Renewable and Sustainable Energy Reviews* 58

Bagheri, Seyed Kamran and Alberto Di Minin, 'The Changing Competitive Landscape of the Global Upstream Petroleum Industry' (2015) 8(1) *The Journal of World Energy Law & Business* 1

Bernardini, Piero, 'Stabilization and Adaptation in Oil and Gas Investments' (2008) 1(1) *Journal of World Energy Law & Business* 98

Bowen, Glenn A, 'Document Analysis as a Qualitative Research Method' (2009) 9(2) *Qualitative Research Journal* 27

Brew, Gregory, 'In Search of "Equitability": Sir John Cadman, Rezā Shah and the Cancellation of the D'Arcy Concession, 1928–33' (2017) 50(1) *Iranian Studies* 125

Brexendorff, Alexander, Christian Ule and Maximilian Kuhn, 'The Iranian Buy-Back Approach' (2009) 7(1) *Oil, Gas & Energy Law Journal (OGEL)* 20

Bunter, Michael A G, 'The Iranian Buy Back Agreement' (2009) 3(1) *Oil, Gas & Energy Law Intelligence (OGEL)* 5

Childs, Thomas C, 'Update on Lex Petrolea: The Continuing Development of Customary Law Relating to International Oil and Gas Exploration and Production' (2011) 4(3) *Journal of World Energy Law and Business* 214

Coale, Margarita. 'Stabilization Clauses in International Petroleum Transactions' (2020) 30(2) *Denver Journal of International Law & Policy* 218

Cronberg, Tarja, 'No EU, No Iran Deal: The EU's Choice Between Multilateralism and the Transatlantic link' (2017) 24(3-4) *The Nonproliferation Review* 244

Cronin, Stephanie, 'The Politics of Debt: The Anglo-Persian Oil Company and the Bakhtiyari Khans' (2004) 40(4) *Middle Eastern Studies* 1

Curtis, Christopher T, 'The Legal Security of Economic Development Agreements' (1988) 29(1) *Harvard International Law Journal* 317.

Derakhshanalavijeh, Roya and Jose Manuel Cardoso Teixeira, 'Cost Overrun in Construction Projects in Developing Countries, Gas-Oil Industry of Iran as a Case Study' (2017) 23(1) *Journal of Civil Engineering and Management* 125

Dudlák, Tamás, 'After the Sanctions: Policy Challenges in Transition to a New Political Economy of the Iranian Oil and Gas Sectors' (2018) 121 *Energy Policy* 464

Ebrahimi, Seyed Nasrollah and Esmat Shahmoradi, 'A Comparative Analysis of Contractual Considerations in Oil and Gas Contracts with an Emphasis on Iranian Petroleum Contracts' (2017) 15(3) *Oil, Gas & Energy Law Intelligence* 1

Elwell-Sutton, Laurence Paul, 'The Iranian Press, 1941–1947' (1968) 6(1) Iran 65

Emami Meibodi, Ali, Atefeh Taklif, Hamidreza Arbab and Hassan Bovairi Monji, 'Investigating the Effects of Contractual Factors and Arrangements on the Optimum Level of Production in Oil and Gas Projects: Evidence from the South Pars Phases 17 & 18' (2020) 24(1) *Iranian Economic Review* 181

Feng, Zhuo, Shui-Bo Zhang and Ying Gao, 'On Oil Investment and Production: A Comparison of Production Sharing Contracts and Buyback Contracts' (2014) 42 *Energy Economics* 395

Flores, Michelle, 'A Practical Approach to Allocating Environmental Liability and Stabilizing Foreign Investment in the Energy Sectors of Developing Countries' (2001) 141(1) *Colorado Journal of International Environmental Law and Policy* 159 Ghandi, Abbas and C-Y Cynthia Lin, 'Oil and Gas Service Contracts Around the World: A Review' (2014) 3 *Energy Strategy Reviews* 63

Ghandi, Abbas and Cynthia Lin Lawell, 'On the Rate of Return and Risk Factors to International Oil Companies in Iran's Buy-Back Service Contracts' (2017) 103(1) *Energy Policy* 16

Ghorbani, Elaheh, 'Exploration, Development and Production of Iran's Fields and Reservoirs through the Iran Petroleum Contract (IPC)' (2020) *Journal of Energy & Natural Resources Law* 1

Hadavimoghaddam, Fakhime and Masoud Mostajeran Gortani, 'Iran Petroleum Contract (IPC)-A New Generation of Oil Contracts' (2019) 13(4) *Actual Problems in Economics & Law* 1522

Haddadi, Mahdi, 'The Nature of Iranian Petroleum Contracts in Upstream Section' (2014) 44(1) *International Letters of Social and Humanistic Sciences* 45

Heidari, Ahmad, 'The Governing Law of International Oil Contracts In Iran Legislations' (2015) 36(4) *Fen Bilimleri Dergisi (CFD)* 1756

Hutchinson, Terry and Nigel Duncan, 'Defining and Describing What We Do: Doctrinal Legal Research' (2012) 17(1) *Deakin Law Review* 83

Kalehsar, Omid S and Azime Telli, 'The Future of Iran-Russia Energy Relations Post-Sanctions' 24(3) (2017) *Middle East Policy* 163

Kamal, Rami A, 'Overview of the Petroleum Industry in the Middle East (1869–1950)' (2005) 24(8) *The Leading Edge* 818

Khah, Maryam S and Ali Amiri, 'Petroleum Contracts in Iran' (2014) 3(3) *European Online Journal of Natural and Social Sciences* 375.

Khan, Kameel, 'The Transfer of Technology and Petroleum Development in Developing Countries: With Special Reference to Trinidad and Tobago' (1986) 4(1) *Journal of Energy & Natural Resources Law* 10

Kuznetsov, Andrey V, 'The Limits of Contractual Stabilization Clauses for Protecting International Oil and Gas Investments Examined Through the Prism of the Sakhalin-2 Psa: Mandatory Law, The Umbrella Clause, And The Fair And Equitable Treatment Standard' (2015) 22(2) *Willamette Journal of International Law and Dispute Resolution* 268 Kuhn, Maximillian and Mohammadjavad Jannatifar, 'Foreign Direct Investment Mechanisms and Review of Iran's Buy-Back Contracts: How Far has Iran Gone and How Far may it Go?' (2012) 5(3) *Journal of World Energy Law & Business* 207

Lauterpacht, Eli, 'Issues of Compensation and Nationality in the Taking of Energy Investments' (1990) 8(1-4) *Journal of Energy & Natural Resources Law* 241

Leader, Sheldon, 'Human Rights, Risks, and New Strategies for Global Investment' (2006) 9(3) *Journal of International Economic Law* 657

Li, Chengyuan, Mingjun Jiang, Haiming Ge, Zhen Li and Dongkun Luo, 'An Operational Risk Analysis of Iran Buyback Contract and its Policy Implication' (2017) 16 *Energy Strategy Reviews* 43

Mafi, Homayoun, 'Iran's Concession Agreements and the Role of the National Iranian Oil Company: Economic Development and Sovereign Immunity' (2008) 48(1) *Natural Resources Journal* 407

Majd, Mohammad G, 'The Oil Boom and Agricultural Development: A Reconsideration of Agricultural Policy in Iran' (1989) 15(1) *The Journal of Energy and Development* 125

Maleki, Mohammad, M. Eslami and M. Rezaei, 'Principles of Arbitration in Oil Contracts' (2018) 5(14) *Revista Publicando* 779

Masoud Derakhshan and Atefeh Taklif, 'The Transfer and Development of Technology in Iranian Upstream Oil Sector: Considerations on the Concepts, Requirements, Challenges and Remedies' (2015) 4(14) *Iranian Energy Economics* 33

Mazlumrahni, Ali Reza and Mohsen Esmaeli, 'Legal Analysis and Economic Impact of Iran's Petroleum New Contracts (IPC)' (2018) 3(2) Örgütsel Davranış Araştırmaları Dergisi 3

Me, M, 'The Iranian Buyback Model and Its Efficiency in the International Petroleum Market-A Legal View' (2009) 7(1) *Oil, Gas and Law Intelligence* 2

Melamid, Alexander, 'Petroleum Product Distribution and the Evolution of Economic Regions in Iran' (1975) 65(4) *Geographical Review* 510

Milenkovic-Kerkovic, Tamara, 'Drafting an International Buy-Back Agreement' (2004) 2 *Economic and Organization* 167
Mohammadi, Teymour, Farshad Momeni, Abbas Kazemi Najaf Abadi and Shirkou Bahadori, 'Effect of Petroleum Contracts on Iranian Oil Production Trend' (2016) 12(50) *Quarterly Energy Economics Review* 25

Mostafa, Mostafavi S and Amani Masoud, 'The Nature of Petroleum International Contracts' (2008) 3 *Islamic Research Journal* 1

Nejad, Roohollah K H, Davood Manzoor and Masoud Amani, 'Economic Evaluation of Fiscal Regime of Buy-Back Contracts in Comparison with Production Sharing Contracts (Case Study: Azadegan Oil Field)' (2018) 22(2) *Iranian Economic Review* 579

Reza Moghaddam, Mohammad, 'Analysis of Buy-back Contracts and Adduction Optimization Model of Contract for Iran (Farsi)' (2008) 76 *Economy Research Journal* 161

Rivlin, Paul, 'Iran's Energy Vulnerability' (2006) 10(4) *The Middle East Review of International Affairs* 103

Sadeghi, Shahdani M and Mahdi Pakzat Seyed, 'Oil Futures Contracts, Obligation to Selling or to Daily Settlement?' (2018) 6(1) *Journal of Asset Management and Finance* 181

Sahebonar, Hamed, Ali Taheri Fard and Fazel M. Farimani, 'Economic Analysis of New Iranian Petroleum Contract (IPC): The Case Study of the Caspian Sea Fields' 1

Shafiee, Mahmood, Isaac Animah, Babakalli Alkali and David Baglee, 'Decision Support Methods and Applications in the Upstream Oil and Gas Sector' (2019) 173(1) *Journal of Petroleum Science and Engineering* 1173

Shahri, Nima Nasrollahi, 'The Petroleum Legal Framework of Iran: History, Trends and the Way Forward' (2010) 8(1) *China & Eurasia Forum Quarterly* 111

Shiravi, Abdolhossein and Seyed Nasrollah Ebrahimi, 'Exploration and Development of Iran's Oilfields through Buyback' (2006) 30(3) *Natural Resources Forum* 199

Stern, Roger, 'The Iranian Petroleum Crisis and United States National Security' (2007) 104(1) *Proceedings of the National Academy of Sciences* 377

Sumbal, Muhammad Saleem, Eric Tsui, Eric See-to, and Andrew Barendrecht, 'Knowledge Retention and Aging Workforce in the Oil and Gas Industry: A Multi Perspective Study' (2017) 21(1) *Journal of Knowledge Management* 180 Sweis, Rateb, <u>Moarefi, Alireza</u>, <u>Mahmood Hosseini Amiri</u>, <u>Moarefi, Soad</u> and <u>Saleh, Rawan</u>, 'Causes of Delay in Iranian Oil and Gas Projects: A Root Cause Analysis' (2019) 13(3) *International Journal of Energy Sector Management* 630

Tavakkoli Mohammadi, Mohammad, Mansour Ahmadnejad and Ali Eshaghzade, 'Identification of Factors Affecting the Development of Accounting and Financial Management Procedures for Joint Operating Agreements (JOA) in Iran's Oil and Gas Industry' (2018) 2(1) *Petroleum Business Review* 23

Tavakkoli, Mohammadi Mohammad and Ali Eshaghzadeh, 'Developing a Framework for Accounting and Financial Management Procedures of Joint Operating Agreements (JOAs) in Iran's Oil and Gas Industry' (2018) 3(10) *International Journal of Finance and Managerial Accounting* 69

TijjaniMato, Hadiza, 'The Role of Stability and Renegotiation in Transnational Petroleum Agreements' (2012) 5(1) *Journal of Politics and Law* 33

van Groenendaal, Willem JH and Mohammad Mazraati, 'A Critical Review of Iran's Buyback Contracts' (2006) 34(18) *Energy Policy* 3709

Yeganehshakib, Reza, 'Iran's New Generation of Oil and Gas Contracts: Historical Mistrust and the Need for Foreign Investment' (2016) 4(2) *Journal of Political Risk* 1

Zhang, Zhijie, Huiqing Lan, and Wanli Xing, 'Global Trade Pattern of Crude Oil and Petroleum Products: Analysis Based on Complex Network' (2018) 153(2) *IOP Conference Series: Earth and Environmental Science* 1.

7.3 Iran

Annual Budget Law (1953) Annual Budget Law (1954) Annual Budget Law (1957) Annual Budget Law (1958) Annual Budget Law (1959) Annual Budget Law (1966) Constitution of Iran <<u>https://www.wipo.int/edocs/lexdocs/laws/en/ir/ir001en.pdf</u>> (accessed 14 July 2019)

Executive Regulations Promotion and Protection of Foreign Investments Act (1962) Law Amendment Act (1941)

Law of the First Five-Year Economic, Social and Cultural Development Plan (1990-1995)

Law of the Second Five-Year Economic, Social and Cultural Development Plan (1995-2000)

Law of the Third Five-Year Economic, Social and Cultural Development Plan (2000-2005)

Law of the Fourth Five-Year Economic, Social and Cultural Development Plan (2005-2010)

Law of the Sixth Five-Year Economic, Social and Cultural Development Plan (2010-2015)

Organization for Investment Economic and Technical Assistance of Iran, 'Laws and Implementing Regulations Concerning Promotion and Protection of Foreign Investment' OIETAI <<u>http://www.satba.gov.ir/suna_content/media/image/2017/02/5236_orig.pdf</u>> (accessed 29 July 2019).

53b04e9652ab&categoryid=9d32c839-2930-4ee6-9321-782d4ac9484a> (accessed 20 March 2020).

7.4 Government Publications (outside of Iran)

National Defense Authorization Act for Fiscal Year 2012, Sec. 1245, <https://www.congress.gov/112/plaws/publ81/PLAW-112publ81.pdf> (accessed 12 March 2020)

National Security Presidential Memorandum/NSPM-11, <<u>https://fas.org/irp/offdocs/nspm/nspm-11.pdf</u>> (accessed 20 March 2020).

7.5 Industry Publications

Addison, Joanna, 'The New Iranian Petroleum Contract Government Approval' (Herbert Smith Freehills, 2016). <<u>https://www.herbertsmithfreehills.com/latest-thinking/the-new-iranian-petroleum-contract-%E2%80%93-government-approval</u>> (accessed 17 February 2020)

Alem, Yasmin, 'Duality by Design: The Iranian Electoral System' (IFES, 2011).

Bijan Khajehpour. 'The Future of the Petroleum Sector in Iran' *Global Transitions* (Legatum Institute, 2013).

Bindemann, Kirsten, 'Production-Sharing Agreements: An Economic Analysis' (The Oxford Institute for Energy Studies, 1999) 1 < https://www.oxfordenergy.org/wpcms/wpcontent/uploads/2010/11/WPM25-ProductionSharingAgreementsAnEconomicAnalysis-KBindemann-1999.pdf> (accessed 17 February 2020).

Creed, Adrian and Dr Amir Kordvani, 'Iran's New Integrated Petroleum Contracts' (MENA Legal Update, Clyde and Co., 2014).

Darvish, Shervin, 'Legal Aspects of Iran's New Oil Contracts' (Natural Gas Europe, 2016) <<u>https://www.linkedin.com/pulse/legal-aspects-irans-new-oil-contracts-shervin-darvish</u>> (accessed 17 February 2020).

Dentons, 'New Model Iranian Petroleum Contract' (2016).

Devine, Richard and Ida Mokhtassi, 'Iran's Council of Ministers ratifies IPC Terms' (Clyde & Co., 2016) 1.

Ebrahmi, Seyed N and Abdul Shiroui Khouozani, 'The Contractual Form of Iran's Buy Back Contracts in Comparison with Production Sharing and Service Contracts' (Society of Petroleum Engineers Inc., 2003).

Energy Information Administration, 'Iran: Country Analysis Brief August 2006' <<u>https://www.yumpu.com/en/document/view/42151382/iran-country-analysis-brief-doe-january-2006-pars-times</u>> (accessed 4 July 2019).

International Monetary Fund, Article IV Consultation, <u>Islamic Republic of Iran</u> (IMF Country Report No. 17/62, February 2017) 4.

Iran Business News, 'Oil Contracts Restructuring Committee' (2015) <<u>https://www.iran-bn.com/tag/oil-contracts-restructuring-committee/</u>> (accessed 30 June 2019).

Iran Group, 'The New Iranian Petroleum Contract – Government Approval' (Herbert Smith Freehills, 2016) <<u>https://www.herbertsmithfreehills.com/latest-thinking/the-new-iranian-petroleum-contract-%E2%80%93-government-approval</u>> (accessed 22 June 2020).

Jalilvand, David R, 'Iran's Gas Exports: Can Past Failure become Future Success?' (The Oxford Institute for Energy Studies, 2013).

Jalilvand, David R and Friedrich Ebert Foundation, 'Iranian Energy: A Comeback with Hurdles' (The Oxford Institute for Energy Studies, 2017).

Munro, Diane, 'Past Imperfect, Future Tense: Iran's Oil Industry Post-Sanctions' (The Arab Gulf States Institute in Washington, 2016).

Natural Resource Governance Institute, 'State Participation in Oil, Gas and Mining' (2015) <http://www.resourcegovernance.org/sites/default/files/nrgi_state_owned_Briefing_Eng_201 50811.pdf> (accessed 6 March 2020).

Okogu, Bright Erakpoweri, 'The Middle East and North Africa in a Changing Oil Market'. (International Monetary Fund, 2003) 40.

Organisation for Economic Cooperation and Development, 'Foreign Direct Investment for Development, Maximizing Benefits, Maximizing Costs' (OECD, 2002) 3.

Organisation of the Petroleum Exporting Countries, 'Annual Statistical Bulletin 2019' <<u>https://www.opec.org/opec_web/en/data_graphs/330.htm</u>> (accessed 9 November 2019)

Priyanka Prakash, 'Iran Opens Up: New Goals, New Deals' (Centre for Air Power Studies, 2017) 1.

R Global, 'Petroleum Contract Models' < https://irglobal.com/article/petroleum-contractmodels-0f76> (accessed 12 December 2019).

Rosneft, 'Rosneft and National Iranian Oil Company Sign Oil and Gas Strategic Cooperation Agreement' (2018) <https://www.rosneft.com/press/releases/item/188381/> (accessed 12 December 2019).

Sabetghadam, Morteza, 'Energy and Sustainable Development in Iran' (Helio International, Sustainable Energy Watch, 2005/2006).

Sammarco, Giuseppe and Simone Tagliapietra, 'The Political and Commercial Barriers to the Development of Iran's Gas Market', in *Iran after the (Potential) Nuclear Deal: What's Next for the Country's Natural Gas Market*? Fondazione Eni Enrico Mattei (FEEM).

Seener, Barak, 'Commercial Risks Entering the Iranian Market: Why Sanctions Make Investment in the Islamic Republic of Iran a High-Risk Proposition' (Jerusalem Center for Public Affairs, 2018) 1. Tavana Energy, 'Iran Petroleum Contract' (Tehran University, 2016) <<u>http://tavana-</u> energy.com/iran-petroleum-contract/> (accessed 17 February 2020).

United Nations Economic Commission for Europe, 'International Buy-Back Contracts' (2010) <<u>http://ierc.bia-</u>

bg.com/uploads/vtcontract/files/vtcontract__00743d3417d145769033e1b5c06e4113.pdf> (accessed 9 December 2019).

United States Energy Information Administration, 'Background Reference: Iran' (2019) <https://www.eia.gov/beta/international/analysis_includes/countries_long/Iran/pdf/iran_bkgd. pdf> (accessed 7 November 2019).

United States Energy Information Administration, 'Monthly Iran Crude Oil and Lease Condensate Exports 2018' (2018) https://www.flickr.com/photos/eiagov/43699512980 (accessed 21 June 2019).

World Bank, 'Ease of Doing Business Rankings'

<http://www.doingbusiness.org/en/rankings?region=middle-east-and-north-africa> (accessed 12 April 2020).

World Energy Council, World Energy Council World Energy Issues Monitor 2017 (2017) <www.worldenergy.org> (accessed 12 April 2021).

Yong, William, 'NIOC and the State–Commercialization, Contestation and Consolidation in the Islamic Republic of Iran' (Oxford Institute for Energy Studies, 2013) 15.

7.6 Online News Publications / Newspapers

Batmanghelidj, Esfandyar, 'Emerging Privitization in Iran's Energy Sector Deserves a Second Look' (2017) *Bourse & Bazaar*

<<u>https://www.bourseandbazaar.com/articles/tag/petropars</u>> (accessed 15 February 2020).

BBC News, 'Six Charts That Show How Hard US Sanctions Have Hit Iran' (2019) <<u>https://www.bbc.com/news/world-middle-east-48119109</u>> (access 12 January 2020).

Bi-Weekly News & Analysis, No. 1, International Law Office of Dr. Behrooz Akhlaghi & Associates http://intllaw.net/wp-content/uploads/2014/06/Iran-Petroleum-Contract-IPC-1st-Bi-Weekly-N-A-April-07-2014.pdf (accessed 17 April 2021).

Dayarayan, 'Iran at a Glance' <http://www.parstimes.com/business/iran_at_a_glance.pdf≥ (accessed 9 December 2019).

Deloitte. Stabilisation Clauses in International Petroleum Contracts Illusion or safeguard? (2014)

<https://www2.deloitte.com/content/dam/Deloitte/ug/Documents/tax/tax_StabilisationClause s 2014.pdf>.

El Sayed, Aline 'Contractors' Right to Clearer Provisions for Time Extension' *Al Misnad Law* (2015) < http://almisnad-law.com/contractors-right-clearer-provisions-time-extensionconstruction-agreements/> (accessed 1 April 2020).

Emirates News Agency, 'Abu Dhabi-Iran joint venture in Shell deal'' (January, 2002) <<u>http://wam.ae/en/details/1395227046863</u>> (access 9 August 2019 2020).

Faucon, Benoit, 'Amid Tensions, Iran's Crude Buyers Jump Ship', *The Wall Street Journal*, (27 May, 2019) https://www.wsj.com/articles/amid-tensions-irans-crude-buyers-jump-ship-11558964215 (accessed 21 June 2019).

Godwin, Peter et al, 'The Prevention Principle, Time at Large and Extension of Time Clauses' (Herbert Smith Freehills LLP, 2009)

<<u>http://documents.lexology.com/4001b4ac8d7a-4d10-b880-ac10e8f54d2a.pdf</u>> (accessed 12 April 2020).

Hosseini, Mehdi, 'National Iranian Oil Company Deputy Managing Director, Speaks Up (Part One)', *Payvand* (May 2005) <www.payvand.com/news/05/may/1146.html> (accessed 21 June 2019).

Iran Petroleum, 'Who Are Iran's New Oil Managers?' (2019) *Shana Petro Energy Information Network* <<u>https://en.shana.ir/news/286983/Who-Are-Iran-s-New-Oil-Managers</u>> (accessed 12 March 2020).

Jalilvand, David Ramin, 'What's Fueling Opposition to Iran's New Oil Contracts?' (2016b) *al-Monitor* < <u>http://www.al-monitor.com/pulse/originals/2016/05/iran-petroleum-contract-</u> <u>framework-delayed.html#ixzz4Zz768ZpT</u> > (accessed 5 February 2020).

Khajehpour Bijan, 'Will Iran Attract International Oil Firms in Post-Sanctions Era?' (2016a) *al-Monitor* <https://www.al-monitor.com/pulse/ar/originals/2016/11/iran-petroleum-contractipc-total-cnpc-azadegan.amp.html?skipWem=1 > (accessed 5 February 2020). Karami, Arash, 'Iran Signs First Energy Contract Since Nuclear Deal' (2017) *al-Monitor* < <u>http://www.al-monitor.com/pulse/originals/2017/07/iran-total-france-south-pars-gas-dealcriticism.html</u> > (accessed 5 February 2020). Kochkodin, Brandon and Grant Smith, 'Iran Ships Most of Its Oil to Asia, Turkey and Europe', *Bloomberg* https://www.bloomberg.com/news/articles/2018-05-08/iran-ships-most-of-its-oil-to-asia-turkey-and-europe-chart (accessed 30 June 2019).

Reuters, 'US is Reportedly Granting 8 Countries Iran Sanctions Waivers', *Reuters* https://www.cnbc.com/2018/11/02/us-is-reportedly-granting-8-countries-iran-sanctions-waivers.html> (accessed 19 June 2019).

Shalal, Andrea, 'Pompeo Warns Russia, China Against Ignoring Move to Reimpose U.N. Sanctions on Iran' *Reuters* (August 2020) <https://www.reuters.com/article/us-usa-iran-unsanctions/pompeo-warns-russia-china-against-ignoring-move-to-reimpose-un-sanctions-oniran-idUSKCN25F2S4> (accessed 7 September 2020).

7.7 Working Papers

Brumberg, Daniel and Ariel I. Ahram, 'The National Iranian Oil Company in Iranian Politics' (Paper Prepared in Conjunction with an Energy Study Sponsored by Japan Petroleum Energy Center and the James A. Baker III Institute for Public Policy, Rice University 2007).

Chandler, Albert T, 'Thailand Petroleum Concessions' (Research Discussion Paper Chandler & Thong-ek Law Offices, 14 December 2015) 2.

Chow, Edward, Cyrus Ashayeri and Andrew J Stanley, 'The Future of Iran's Oil and Gas Industry' (USAEE Working Paper No. 18-364, 2018) 5.

Hamilton, James D, 'Historical Oil Shocks' (Working Paper No. w16790. National Bureau of Economic Research, 2011) 1.

Parris, Richard and Louis Skyner, 'Key Comparisons of the New Iran Petroleum Contract and Buy-Back' (Briefing Note, Clifford Chance, 2017) 1.

https://www.cliffordchance.com/content/dam/cliffordchance/briefings/2017/01/key-comparisons-of-the-new-iran-petroleum-contract-and-buyback.pdf (accessed 1 December 2019).

Tordo, Silvana, 'Fiscal Systems for Hydrocarbons: Design Issues' (World Bank Working Paper, no. 123, 2007).

7.8 Conferences

Aghion, Philippe and Lucía Quesada, 'Petroleum Contracts: What does Contract Theory Tell Us?' (International Workshop on Microeconomics Applied to the Energy Industry, 15 December 2011).

Paliashvili, Irina, 'The Concept of Production Sharing', Outlines of the Presentation of Dr. Irina Paliashvili, the President of the Russian-Ukrainian Legal Group, at the Seminar on the Legislation on Production Sharing Agreements, 14 September 1998 <http://www.rulg.com/documents/The_Concept_of_Production_Sharing.htm> (accessed 7 January 2020).

Soleimani, M and Newsha Tavakolian, 'Comparative Financial Analysis of IPC, the New Iran's Oil Fields Development and Production Enhancement Framework IPC vs. Buyback Contracts' (SPE Symposium: Production Enhancement and Cost Optimisation, Society of Petroleum Engineers, 2017).

7.9 Interviews

Afarideh, Hossein Dr, 'Iran Ready to Give Production Sharing Agreements' (Menas Associates' Iran Energy Focus, February 2003)

<<u>http://www.payvand.com/news/03/mar/1026.html</u>> (access 12 February 2020).

7.10 Dictionaries

Black's Law Dictionary (4th ed, 1951) 89.

Encyclopædia Britannica (online) 2021 < https://www.britannica.com>.

Macquarie Dictionary (3rd ed, 1997) 241.